Diss. ETH No. 25628

## Perturbation of eigenvalues and scattering resonances with applications in imaging

A thesis submitted to attain the degree of DOCTOR OF SCIENCES of ETH ZÜRICH (Dr. sc. ETH Zürich)

presented by

#### ALEXANDER DABROWSKI

Laurea Magistrale in Matematica – Università degli Studi di Trieste born June 16, 1991 citizen of Italy and Poland

accepted on the recommendation of

Prof. Dr. Habib Ammari, ETH Zürich, examiner Prof. Dr. Nilima Nigam, Simon Fraser University, co-examiner

2019

## Contents

A	Abstract									
Introduction										
1	Fun	damental notation and results	11							
<b>2</b>	Hig	Higher-order terms in a small volume expansion								
	2.1	Preliminaries	18							
		2.1.1 Integral formulation	20							
	2.2	Computations for explicit formulae	23							
		2.2.1 Zero-order term	24							
		2.2.2 First-order term	26							
		2.2.3 On higher-order terms	27							
	Results for special cases	28								
3	Small domain perturbations which split the spectrum									
	3.1	Preliminaries	32							
		3.1.1 Stability of eigenvalues of the Laplacian	34							
	3.2	Hadamard's formula and boundary properties of eigenfunctions	35							
	3.3	Splitting by a perturbation of the boundary of a hole	38							
	3.4	Splitting by a perturbation of the boundary	42							
4	Extending asymptotic formulae from simple to repeated eigenval-									
	ues		45							
	4.1	Preliminaries	46							
		4.1.1 Stability and simplicity of the spectrum of second-order el-								
		liptic operators	47							
	4.2	Variational asymptotic formula	49							

		4.2.1	Asymptotic formulae involving a bilinear function $\ldots$	51						
	4.3	Applications to eigenvalues of the Laplacian								
		4.3.1	Perturbation by a grounded inclusion	52						
		4.3.2	Perturbation by a conductivity inclusion	54						
		4.3.3	Perturbation by boundary deformation	58						
		4.3.4	Summary of results	60						
<b>5</b>	Rec	onstru	ction of small perturbations from eigenvalues' shifts	63						
	5.1	Grounded inclusion								
	5.2	2 Conductivity inhomogeneity								
	5.3	Bound	lary deformation	72						
6	6 Perturbation of scattering resonances - TM case									
	6.1	One d	imensional case	76						
	6.2	Multi-	dimensional case	80						
	6.3	Pertur	bation of whispering-gallery modes by an external particle	84						
	6.4	Nume	rical illustrations	86						
7	ion of scattering resonances - TE case	89								
	7.1	.1 Scattering resonances of a dielectric cavity								
	7.2	Shift o	of resonances by internal particles	91						
	7.3	Shift o	of resonances by external particles	93						
	7.4	Shift o	of resonances by plasmonic particles	94						
	7.5	Asym	ptotic analysis near exceptional points	94						
8	Scattering of highly refractive particles 9									
	8.1	Deriva	tion of asymptotic estimates	98						
	8.2	Hybrid	dization of subwavelength resonances for a dimer $\ldots \ldots \ldots$	101						
Bi	Bibliography 103									

### Abstract

#### English

This thesis contributes to the study of eigenvalue and scattering resonance perturbations caused by domain variations. We derive asymptotic formulae with layer potential techniques, and use them to image small particles.

We present several new results. We compute higher-order terms in the asymptotic expansions of eigenvalues, study their simplicity, and derive a variational characterization in the case of multiplicity higher than one. We report the results of different numerical experiments which validate and illustrate our findings, and propose computational techniques to reconstruct small domain variations from eigenvalue perturbations. We also study the asymptotic behavior of scattering resonances, first formally, via the method of matched expansions, and then rigorously, using the spectral properties of the Newtonian potential. We focus in particular on the analysis of plasmonic and highly refractive nanoparticles. Our formulae can be applied to the analysis and design of resonant structures.

#### Italiano

Questa tesi contribuisce allo studio di perturbazioni di autovalori e risonanze di scattering causate da variazioni di dominio. Deriviamo formule asintotiche con i potenziali di superficie, e le utilizziamo per ricostruire la forma di piccole particelle.

Presentiamo diversi risultati nuovi. Calcoliamo termini di grado superiore negli sviluppi asintotici degli autovalori, studiamo la loro semplicità, e deriviamo una caratterizzazione variazionale nel caso di molteplicità superiore a uno. Riferiamo i risultati di diversi esperimenti numerici che validano e illustrano le nostre conclusioni, e proponiamo tecniche computazionali per ricostruire piccole variazioni di domini da perturbazioni di autovalori. Studiamo anche il comportamento asintotico di risonanze di scattering, prima formalmente, attraverso il metodo delle espansioni abbinate, e poi rigorosamente, utilizzando le proprietà spettrali del potenziale Newtoniano. In particolare ci focalizziamo sull'analisi di nano-particelle plasmoniche e altamente rifrangenti. Le nostre formule si possono applicare all'analisi e alla progettazione di strutture risonanti.

## Introduction

In this thesis, we study the asymptotic behavior of eigenvalues and scattering resonances with respect to different types of domain perturbations. Our purpose is to develop some procedures to quickly estimate quantities of interest, and provide asymptotic error bounds.

The dependence of the spectrum of a differential operator on its domain of definition has been a thoroughly studied topic since the seminal work of Hadamard in the early 20th century. Probably even a larger literature has been devoted to the development of a perturbation theory for eigenvalues of linear operators, first from the quantum physics community, and then from the point of view of mathematical analysis. The foundational results of this area can be found in the works of Rellich and Kato from the mid 20th century. However, there is much interest in this line of research still today. For instance, the shift of eigenvalues caused by singular domain perturbations is a problem recently examined, with techniques from layer potential theory, by Ammari and collaborators. One of the goals of this thesis is an extension of these results. In fact, in Chapter 2, we show how the generalized argument principle of Gohberg and Sigal can be used to derive new higher-order terms in the asymptotic expansion of Neumann eigenvalues of the Laplacian.

A related issue, which has also been examined extensively in the literature, concerns the behavior of repeated eigenvalues. However, we believe that the study of repeated eigenvalues, for some particular operators, has not been thoroughly pursued, mainly due to the results of Uhlenbeck and Micheletti in the 60s and 70s, which state that, generically, eigenvalues are simple. We report these results in Chapter 3, where we also provide some generalizations to non-smooth domains and Neumann boundary conditions. One of the highlights of this thesis is a new variational characterization for multiple eigenvalues of self-adjoint, elliptic differential operators, presented in Chapter 4. We also discuss in Chapter 5 its applications to domain variations, which are of interest in applications.

A natural extension to the setting examined so far is to consider scattering resonances on unbounded domains. This is another area of research, rooted in the work on scattering theory by Lax and Phillips in the '60s, which is still active today, with recent results from Popov, Vodev, Zworski, and collaborators. Scattering resonances are of great importance in physics and technology, and new results in this setting are of great interest in applications, especially due to recent advancements in nano-optics and micro-biology. The mathematical models in these areas, however, often differ among communities and use fundamentally diverse assumptions and terminology. In this thesis, we provide a mathematical framework to analyze perturbations of scattering resonances, focusing our attention on scalar waves. We show how a modification of the scatterers influences the resonances, and derive new asymptotic expansions, first formally in Chapter 6 with the method of matching asymptotic expansions, and then rigorously in Chapter 7, using the spectral properties of the Newtonian potential and pole-pencil decompositions of the involved operator-valued functionals. We focus our attention on analyzing some situations which are currently of great interest in nano-opitcs: whispering gallery resonators, plasmonic nanoparticles, and finally, in Chapter 8, highly refractive particles. In all of these cases, we derive new asymptotic formulae, which can be used for the analysis and design of nano-structures with given resonance requirements.

We move on to present more in detail the topics of each chapter.

Structure of the thesis. The thesis is divided into eight chapters.

In Chapter 1, we fix the notation which will be used through the rest of the thesis and recall some useful results from the literature, mainly regarding Bessel functions, fundamental solutions, layer potentials and Gohberg-Sigal theory. We also describe the tools used to numerically validate the theoretical results.

In Chapter 2, we derive some terms in the small-volume asymptotic expansion of the shift of Neumann eigenvalues of the Laplacian caused by a grounded inclusion of area  $\varepsilon^2$ . We present a new higher-order explicit formula to compute them from the capacity, the eigenvalues and the eigenfunctions of the unperturbed domain, and the size and the position of the inclusion. The key step in the derivation is the filtering of the spectral decomposition of the Neumann function with the residue theorem. As a consequence of the formula, when a bifurcation of a double eigenvalue occurs (as for example in the case of a generic inclusion inside a disk) one eigenvalue decays like  $O(1/\log \varepsilon)$ , the other like  $O(\varepsilon^2)$ . The results of this chapter have been published in [38].

In Chapter 3, we construct an arbitrarily small and localized perturbation of any Lipschitz domain, which splits the Dirichlet, Neumann, or Robin repeated eigenvalues of the Laplacian into simple ones. We showcase two different approaches. The first one consists in the excision of a hole inside the domain and the perturbation of its boundary, and is based on a Hadamard's formula and sharp spectral stability estimates. The second one consists in the deformation of the boundary of the domain itself, and requires further properties of the bilinear form of the variational problem. The results of this chapter are the subject of [39].

In Chapter 4, we derive a new variational characterization for the shift of eigenvalues caused by a general type of perturbation for self-adjoint elliptic differential operators. This result allows the direct extension of asymptotic formulae from simple eigenvalues to repeated ones. It generalizes the results from Chapter 2 to higher multiplicity eigenvalues. Some interesting examples for the Laplacian are presented theoretically and numerically for the following domain perturbations: excision of a small hole, local change of conductivity, small boundary deformation. The results of this chapter are the subject of [40].

In Chapter 5, we present some techniques to reconstruct features of domain perturbations from the shift of eigenvalues they cause in a closed resonator. For this purpose, we use asymptotic formulae, under the assumption of smallness of the perturbations considered. The main appeal of the methods introduced is that they provide a very quick estimate of particles' features of any shape and in any dimension. We provide also many practical numerical examples, considering different types of perturbations. The results of this chapter are the subject of [44].

In Chapter 6, we derive formally a small-volume expansion for the scattering resonances of an open cavity perturbed by a small particle. We derive an asymptotic expression for the induced shift of the scattering frequencies, without neglecting the radiation effect. We consider also the case of plasmonic particles and show that they cause a stronger enhancement in the frequency shift. The derived formula can be used to image small particles located near the boundary of an open resonator which admits whispering-gallery modes. Numerical examples of interest for applications are also presented. The results of this chapter are the subject of [5]. In Chapter 7, we consider the transverse electric polarization case, and derive a small-volume formula for the shifts in the scattering resonances of a radiating dielectric cavity perturbed by small particles. We show that again a strong enhancement in the frequency shift appears in the case of plasmonic particles. We also consider exceptional scattering resonances and characterize their asymptotic behavior for a specific form of the Green's function. Our approach relies on pole-pencil decompositions of volume integral operators. The results of this chapter are the subject of [6].

In Chapter 8, we analyze the subwavelength resonances of dielectric particles with high refractive indices. We show that for an arbitrary shaped particle, these subwavelength resonances can be expressed in terms of the eigenvalues of the Newtonian potential associated with its shape. We derive an asymptotic estimate for the enhancement of the scattered field at the resonant frequencies. We also characterize the hybridization of the subwavelength resonances of a dimer of highly refractive dielectric nanoparticles. The results of this chapter are the subject of [7].

## Chapter 1

## Fundamental notation and results

In this chapter, we fix the notation and recall some results from the literature which will be extensively used through the rest of the thesis.

**Reference space.** We will mostly work with *domains*, which we define as subsets of  $\mathbb{R}^d$  which are open, bounded, connected, and have Lipschitz boundaries. We reserve since now the symbol  $\Omega$  to indicate such an arbitrary domain.

**Basic notation.** We collect here some notation and assumptions which will be used often.

- We assume any function and any set we work with to be Lebesgue measurable.
  We indicate with |x| the modulus, the norm, or the Lebesgue measure of x, depending on x being a number, a vector, or a set.
- We indicate as  $\nu$  the outward normal to a boundary  $\partial\Omega$ , as  $\frac{\partial}{\partial\nu}$  or  $\frac{\partial}{\partial\nu}\Big|_+$  the normal derivative from the exterior of  $\Omega$ , and as  $\frac{\partial}{\partial\nu}\Big|_-$  the normal derivative from the interior.
- We indicate as *I* the identity map or, when working on a finite dimensional vector space with a fixed basis, the identity matrix.

Function spaces. The spaces of functions we will encounter most often are:  $C^{k,\alpha}(\Omega)$ , the space of functions with  $\alpha$ -Hölderian kth derivative in  $\Omega$ ;  $C^k(\Omega)$  the space of functions with continuous kth derivative;  $L^p(\Omega)$ , the space of p-integrable functions on  $\Omega$ ;  $H^s(\Omega)$ , the subset of  $L^2(\Omega)$  whose distributional derivatives up to order s are also in  $L^2(\Omega)$ ;  $H^1_0(\Omega)$ , the subset of  $H^1(\Omega)$  with trace zero. **Bessel functions.** Let  $\alpha, z \in \mathbb{C}$ . Let us recall here some fundamental results on the canonical solutions to Bessel differential equation  $z^2 f''(z) + z f'(z) + (z^2 - \alpha^2) f(z) = 0$ . The Bessel function  $J_{\alpha}$  of the first kind and order  $\alpha$  can be defined, when  $\alpha$  is not a negative integer, as

$$J_{\alpha}(z) = \left(\frac{z}{2}\right)^{\alpha} \sum_{k=0}^{\infty} \frac{(-1)^k}{k! \Gamma(k+\alpha+1)} \left(\frac{z}{2}\right)^{2k},\tag{1.1}$$

where  $\Gamma$  indicates the Gamma function, that is  $\Gamma(z) = \int_0^\infty t^{z-1} e^{-t} dt$ . The series in (1.1) converges on the whole  $\mathbb{C}$ . When  $\alpha$  is a negative integer we define  $J_\alpha = -J_{-\alpha}$ .

The Bessel function  $Y_{\alpha}$  of the second kind and order  $\alpha$  can be defined, when  $\alpha$  is not an integer, as

$$Y_{\alpha}(z) = \frac{J_{\alpha}(z)\cos(\pi\alpha) - J_{-\alpha}(z)}{\sin(\pi\alpha)},$$

and, when  $\alpha$  is an integer, as the limit of  $Y_{\beta}$  as  $\beta \to \alpha$ . Both  $J_{\alpha}$  and  $Y_{\alpha}$  are analytic on  $\mathbb{C} \setminus (-\infty, 0]$ . For the particular case of  $\alpha = 0$ , we have that

$$Y_0(t) = \frac{2}{\pi} \sum_{n=0}^{\infty} (-1)^n \left(\frac{t^n}{2^n n!}\right)^2 \log(\eta_n t),$$

with  $\log \eta_n = \text{Euler-Mascheroni constant} - \log 2 + \sum_{k=1}^n (1/k)$ .

The Hankel function  $H^{(1)}_{\alpha}$  of the first kind and order  $\alpha$  can be defined as

$$H_{\alpha}^{(1)}(z) = J_{\alpha}(z) + \mathrm{i}Y_{\alpha}(z).$$

We refer to [2, Chapter 9] for further details.

**Fundamental solutions.** Let L be a linear differential operator on  $\mathbb{R}^d$ . A fundamental solution, or Green function, of L is a distribution  $\Gamma$  such that

$$L\Gamma(\cdot, z) = \delta_z,$$

where  $\delta_z$  is the Dirac mass function concentrated in z and the equality is intended in the distributional sense (see, for instance, [91]).

When L is the Helmholtz operator, that is  $L = \Delta + \omega^2$ , we have an explicit expression for the fundamental solution via the Hankel function. For any  $\omega \in \mathbb{C} \setminus (-\infty, 0], r > 0$ , we define

$$\Gamma_{\omega}(r) = -\frac{i}{4} \left(\frac{\omega}{2\pi r}\right)^{\frac{d}{2}-1} H^{(1)}_{\frac{d}{2}-1}(\omega r), \qquad (1.2)$$

and

$$\Gamma_{0}(r) = \begin{cases} \frac{r}{2} & \text{if } d = 1, \\ \frac{1}{2\pi} \log r & \text{if } d = 2, \\ \frac{1}{d(2-d)|S^{d-1}|r^{d-2}} & \text{if } d \ge 3, \end{cases}$$
(1.3)

where  $S^{d-1}$  is the (d-1)-dimensional unit sphere. We remark that our sign convention is the same as [48] but the opposite of [41].

When dealing with  $x, z \in \mathbb{R}^d$ , we define  $\Gamma_{\omega}(x, z) = \Gamma_{\omega}(|x-z|)$ . As a consequence of the definitions,  $\Gamma_{\omega}$  is a fundamental solution for the operator  $\Delta + \omega^2$  on  $\mathbb{R}^d$  for any  $\omega \in \mathbb{C} \setminus (-\infty, 0)$ , that is

$$(\Delta_x + \omega^2)\Gamma_\omega(|x - z|) = \delta_z(x).$$

We remark that there is no guarantee of uniqueness for fundamental solutions. In fact, whenever working on a bounded domain, for simplicity we will substitute  $\Gamma_{\omega}(r)$ with its real part, that is

$$\frac{1}{4} \left(\frac{\omega}{2\pi r}\right)^{\frac{d}{2}-1} Y_{\frac{d}{2}-1}(\omega r).$$
(1.4)

This simplifies the numerical computations and allows us to restrict the whole analysis to real numbers, since for bounded domains we will consider exclusively  $\omega \in \mathbb{R}$ . On unbounded domains, on the contrary, we must use  $\Gamma_{\omega}(r)$  as defined in (1.2), due to the requirement of a radiation condition, which we now specify.

Sommerfeld radiation condition. A function  $u : \mathbb{R}^d \to \mathbb{C}$ , which is a solution to Helmholtz equation  $(\Delta + \omega^2)u = 0$ , satisfies the Sommerfeld radiation condition if

$$\frac{\partial u}{\partial \nu} - i\omega u \to 0 \quad \text{in } L^2(B_r) \quad \text{as } r \to \infty,$$
 (1.5)

where  $B_r$  is the ball of radius r centered at the origin.

The physical meaning of the condition is to select the outgoing waves among all possible solutions to the Helmholtz problem. We remark that  $\Gamma_{\omega}$ , as defined in (1.2), can be characterized as the unique fundamental solution which satisfies the Sommerfeld radiation when  $\omega \in \mathbb{C} \setminus (-\infty, 0]$ , and when  $\omega = 0$ , as the unique limit of  $\Gamma_{\omega'}$  as  $\omega' \to 0$ . **Layer potentials.** A fundamental tool which we will use extensively is that of boundary layer potentials, which we now introduce.

**Definition 1.1.** The single layer potential, the double layer potential, and the Neumann-Poincaré operator of  $\Omega$  at frequency  $\omega \in \mathbb{C} \setminus (-\infty, 0)$  are defined respectively as the operators  $\mathcal{S}^{\omega}_{\Omega}$ ,  $\mathcal{D}^{\omega}_{\Omega}$ ,  $\mathcal{K}^{\omega}_{\Omega}$  such that for every  $\phi \in L^2(\partial \Omega)$  it holds

$$\mathcal{S}_{\Omega}^{\omega}[\phi](x) = \int_{\partial\Omega} \Gamma_{\omega}(|x-y|)\phi(y) \ d\sigma(y), \text{ for } x \in \mathbb{R}^{d}, \tag{1.6}$$

$$\mathcal{D}_{\Omega}^{\omega}[\phi](x) = \int_{\partial\Omega} \frac{\partial \Gamma_{\omega}(|x-y|)}{\partial \nu(y)} \phi(y) \ d\sigma(y), \text{ for } x \in \mathbb{R}^{d} \setminus \partial\Omega, \tag{1.7}$$

$$\mathcal{K}_{\Omega}^{\omega}[\phi](x) = \int_{\partial\Omega} \frac{\partial \Gamma_{\omega}(|x-y|)}{\partial \nu(y)} \phi(y) \ d\sigma(y), \text{ for } x \in \partial\Omega,$$
(1.8)

where the integral in (1.8) is intended to be in the principal value sense.

In the following propositions, we recall some properties of the layer potentials which we will need later.

**Proposition 1.2.** The L<sup>2</sup>-adjoint of  $\mathcal{K}_{\Omega}^{-\omega}$ , which we indicate as  $(\mathcal{K}_{\Omega}^{\omega})^*$ , satisfies

$$(\mathcal{K}_{\Omega}^{\omega})^{*}[\phi](x) = \int_{\partial\Omega} \frac{\partial\Gamma_{\omega}(|x-y|)}{\partial\nu(x)} \phi(y) \ d\sigma(y), \ \text{for } x \in \partial\Omega$$

Moreover  $\mathcal{K}_{\Omega}^{\omega}$  is bounded from  $L^{2}(\partial\Omega)$  to  $H^{1}(\partial\Omega)$ .

**Proposition 1.3.** The single layer potential,  $S_{\Omega}$ , and the normal derivative of the double layer potential,  $\partial D_{\Omega} / \partial \nu$ , are both continuous across  $\partial \Omega$ . The normal derivative of the single layer potential and the double layer potential are both discontinuous across  $\partial \Omega$  and

$$\frac{\partial \mathcal{S}_{\Omega}^{\omega}}{\partial \nu}\Big|_{\pm} = \pm I/2 + (\mathcal{K}_{\Omega}^{\omega})^*, \qquad (1.9)$$

$$\mathcal{D}_{\Omega}^{\omega}|_{\pm} = \mp I/2 + \mathcal{K}_{\Omega}^{\omega}. \tag{1.10}$$

For an in-depth study of the properties of layer potentials and their extensive applications in the theory of boundary value problems we refer to [35, 97].

**Capacity of a set.** The single layer potential can be used to define the capacity of a set, a useful quantity related to its geometry, as follows. It can be shown that there exists a unique non-zero couple  $(\varphi_{cap}, a) \in L^2(\partial\Omega) \times \mathbb{R}$  which solves

$$\begin{cases} \mathcal{S}_{\Omega}^{0}[\varphi_{\text{cap}}](x) \equiv a \quad \forall x \in \partial \Omega \\ \int_{\partial \Omega} \varphi_{\text{cap}} = 1. \end{cases}$$

Then the *capacity* of  $\partial \Omega$  is defined as

$$\operatorname{cap} \partial \Omega = \begin{cases} e^{2\pi a} & \text{if } d = 2, \\ 1/a & \text{if } d \ge 2. \end{cases}$$

For further details on the capacity see [59, Section 16.4] and [25].

**Polarization tensors.** The *polarization tensor*  $M(k, \Omega)$  of a domain  $\Omega$  with coefficient  $k \in \mathbb{C}$ , is a  $d \times d$  matrix which (i, j)-th entry is

$$\int_{\partial\Omega} \left( \frac{k+1}{2(k-1)} I - (\mathcal{K}^0_{\Omega})^* \right)^{-1} (\nu_j) y_i \ d\sigma(y), \tag{1.11}$$

where  $\nu_j$ ,  $y_i$  indicate respectively the *j*th and *i*th components of  $\nu$  and y.

The polarization tensor encodes different geometric properties of the underlying domain  $\Omega$ . For the discussion of this result and many other properties we refer to [12]. We recall here only the fact that if  $\Omega$  is a two dimensional ellipse with semiaxes a and b, rotated counter-clockwise by a degree  $\alpha$  with respect to the coordinate axes, we can explicitly compute

$$M(k,\Omega) = (k-1)R_{\alpha} \begin{pmatrix} \frac{a+b}{a+kb} & 0\\ 0 & \frac{a+b}{a+kb} \end{pmatrix} R_{\alpha}^{T}, \quad \text{with } R_{\alpha} = \begin{pmatrix} \cos\alpha & -\sin\alpha\\ \sin\alpha & \cos\alpha \end{pmatrix}.$$
(1.12)

**Gohberg-Sigal theory.** Recall that if  $f : \mathbb{C} \to \mathbb{C}$  is meromorphic, then the argument principle states that for any subset  $V \subseteq \mathbb{C}$  with smooth boundary  $\partial V$  on which f has no poles and no zeros, it holds

$$\frac{1}{2\pi i} \int_{\partial V} \frac{f'(z)}{f(z)} dz = N - P,$$

where N and P are respectively the number of zeros and poles of f inside V, counted with their multiplicity. An analogous result can be formulated for infinite dimensional operators. Its formulation however requires the introduction of some technical definitions, which we briefly summarize hereafter.

Let A be a functional from  $\mathbb{C}$  to the space  $\mathcal{L}$  of linear operators between two arbitrary Banach spaces. A point  $z_0 \in \mathbb{C}$  is a *characteristic value* of A, if there exists a function  $\phi(z)$  holomorphic and non-zero at  $z_0$ , and such that  $A(z)\phi(z)$  is holomorphic and zero at  $z_0$ . We say that A is *finitely meromorphic* at a point  $z_0 \in \mathbb{C}$  which is a pole for A, if it can be written as

$$A(z) = \sum_{j \ge -s} (z - z_0)^j A_j,$$

for some natural number s, and if  $A_{-1}, \ldots, A_{-s}$  are finite dimensional operators. It is easy to show that if A is finitely meromorphic at  $z_0$  then  $\int_{\partial V} A(z)dz$  is a finitedimensional operator for any V which contains no other poles than  $z_0$ .

Recall that an operator is called *Fredholm* if its kernel and cokernel are both finite dimensional. With these notions we can formulate a generalization of the argument principle as follows.

**Theorem 1.4.** Let A be an operator-valued functional which is finitely meromorphic and Fredholm in V, continuous and invertible on  $\partial V$ . Then

$$\frac{1}{2\pi i} \operatorname{tr}\left(\int_{\partial V} A^{-1}(z)\partial_z A(z) \ dz\right) = N - P,$$

where N and P are respectively the number of characteristic values and of poles of A(z) in V, counted with their multiplicities.

For the proof of this result and further details we refer to [49, 14].

Tools used for numerical validation. Various numerical experiments have been performed to validate the theoretical formulae obtained in this thesis. Finite element simulations for eigenvalue system have been performed with the pdeeig function from MATLAB. The predefined triangular mesh has been used, varying the number of recursive refinements until convergence of the required significant digits for each experiment.

The multipole expansion method has been implemented in MATLAB following the exposition of [73]. Although this method converges much faster than the finite element approach, it relies on summation formulae of elliptic functions, and is thus limited to problems where the shapes involved are ellipsoids.

Calculations of the terms appearing in the derived asymptotic formulae have been performed in Mathematica. To evaluate numerically the oscillatory integrals involved in the higher-order terms of some asymptotic formulae, the function NIntegrate with adaptive sampling has been used.

## Chapter 2

# Higher-order terms in a small volume expansion

Consider a planar bounded domain  $\Omega$  and let  $\omega^2$  be an eigenvalue of the negative Laplacian on  $\Omega$  with homogeneous Neumann boundary conditions. Suppose a small inclusion  $D = z + \varepsilon B$  (where  $z \in \Omega$ ,  $|B| = |\Omega|$ , and  $\varepsilon$  is small) is inserted inside  $\Omega$ . This may cause the eigenvalue  $\omega_{\varepsilon}^2$  of the perturbed domain  $\Omega \setminus D$  (with Neumann condition on  $\partial\Omega$  and Dirichlet on  $\partial D$ ) to vary in value or in multiplicity with respect to  $\omega^2$ . Asymptotic formulae of the perturbation with respect to the size of the inclusion have been derived in the '80s in [86, 30]. In particular, it has been shown that if  $\omega^2$  is simple and u is the associated L<sup>2</sup>-normalized eigenfunction, then the perturbation is singular and

$$\omega_{\varepsilon}^{2} - \omega^{2} = -\frac{2\pi |u(z)|^{2}}{\log \varepsilon} + o\left(1/\log(\varepsilon)\right).$$
(2.1)

More recently, Gohberg-Sigal theory for meromorphic operators applied to the integral equation formulation of the eigenvalue problem has led to new results (see [15, 22]). In this chapter we elaborate on these results to further improve (2.1), by calculating explicitly the terms up to  $O(\varepsilon^2)$  and by generalizing it to the case of multiplicity 2. As a consequence of our derivation, for perturbed eigenvalues  $\omega_{\varepsilon,1}^2 < \omega_{\varepsilon,2}^2$ splitted from a double eigenvalue  $\omega^2$  of the original domain  $\Omega$ , it holds

$$\omega_{\varepsilon,2} - \omega = -\frac{C_1}{\log(\varepsilon) + C_2} + O(\varepsilon^2)$$
$$\omega_{\varepsilon,1} - \omega = O(\varepsilon^2),$$

where  $C_1$  and  $C_2$  do not depend on  $\varepsilon$  and can be explicitly calculated from the capacity, the eigenvalues, and the value at z of the eigenfunctions of  $\Omega$ .

This chapter is organized as follows. After introducing in Section 2.1 the precise setting of the problem and the notation, in Section 2.1.1 we recall the equivalent formulation of the Laplacian eigenvalues as characteristic values of an appropriate integral operator. An asymptotic expansion of this integral operator can be obtained by expanding in Taylor series the free space fundamental solution. Gohberg-Sigal theory then provides a link between eigenvalues' shifts and the traces of these integral operators through power sum polynomials. In Section 2.2, explicit terms for the small volume expansion of these power sum polynomials are derived by using properties of layer potentials. The key step in this derivation is the filtering of the spectral decomposition of the Neumann function using the residue theorem to obtain geometric-like series which can be summed. A proposal for formal automated computation of higher-order coefficients is given in Section 2.2.3. Finally in Section 2.3 some interesting consequences for special cases and a validation with numerical experiments are provided.

#### 2.1 Preliminaries

The eigenvalue problem. Let  $\Omega$  be an open, bounded and connected subset of  $\mathbb{R}^2$  with C<sup>1</sup>-boundary. It is well known that the eigenvalues of the negative Laplacian on  $\Omega$  with Neumann boundary condition are non-negative, have finite multiplicity and can be arranged in an increasing divergent sequence

$$0 = \omega_0^2 < \omega_1^2 < \omega_2^2 < \dots < \omega_k^2 \to \infty.$$

For each index i, let  $m_i$  be the multiplicity of  $\omega_i^2$ . We choose the associated eigenfunctions  $u_{i,1}, \ldots, u_{i,m_i}$  to be orthonormal in L<sup>2</sup>. Thus, for any i, j, we have

$$\begin{cases} (\Delta + \omega_i^2) u_{i,j} = 0 & \text{ in } \Omega, \\ \frac{\partial u_{i,j}}{\partial \nu} = 0 & \text{ on } \partial \Omega \end{cases}$$

and

$$\int_{\Omega} u_{i,j} u_{k,l} = \begin{cases} 1 & \text{if } i = k \text{ and } j = l, \\ 0 & \text{otherwise.} \end{cases}$$

We will often use the vector notation

$$U_i := (u_{i,1}, \dots, u_{i,m_i}).$$
 (2.2)

Invertibility of the single layer potential. Consider the special case when  $\Omega$  is the unit disk. Writing t for the angle in the usual polar parametrization of the boundary, one can calculate that

$$\mathcal{S}_{\Omega}^{0}[e^{int}](\tau) = \begin{cases} 0 & \text{if } n = 0, \\ -\frac{1}{2n}e^{in\tau} & \text{otherwise.} \end{cases}$$

Thus, we have an explicit expression of  $S^0_{\Omega}$  in the Fourier basis of  $L^2(\partial\Omega)$ . Notice however that the fact that  $S^0_{\Omega}[1] = 0$  causes the non-invertibility of  $S^0_{\Omega}$ . Nonetheless, if we consider

$$\phi \mapsto \mathcal{S}^0_{\Omega}[\phi] + \lambda \int_{\partial \Omega} \phi, \qquad (2.3)$$

we see that, for a constant  $\lambda \neq a$ , this operator is always invertible from  $L^2(\partial \Omega)$  to  $H^1(\partial \Omega)$ . This is still true for a more general domain  $\Omega$  as in our assumptions (see [97, Theorem 4.11] for more details).

**Fundamental solution for a bounded domain.** The Neumann function  $N_{\Omega}^{\omega}$  is defined as the solution of

$$\begin{cases} (\Delta_x + \omega^2) N_{\Omega}^{\omega}(x, z) = \delta_z(x) & \text{ for } x \in \Omega, \\ \frac{\partial N_{\Omega}^{\omega}(x, z)}{\partial \nu(x)} = 0 & \text{ for } x \in \partial\Omega. \end{cases}$$

where  $\omega \in \mathbb{C}$  is not one of  $\omega_i$  and  $z \in \Omega$ . It has the spectral representation

$$N_{\Omega}^{\omega}(x,z) = \sum_{j=1}^{\infty} \frac{U_j(x) \cdot U_j(z)}{\omega^2 - \omega_j^2},$$

where the convergence of the series to  $N_{\Omega}^{\omega}$  in general is only in L<sup>2</sup> (see [37, *expansion theorems*]). Here,  $U_j$  is defined by (2.2).

By integrating  $N_{\Omega}^{\omega}$  against test functions in  $L^2(\partial\Omega)$  and using the properties of layer potentials one can show that

$$(I/2 - \mathcal{K}_{\Omega}^{\omega})^{-1} [\Gamma_{\omega}(\cdot, z)](x) = N_{\Omega}^{\omega}(x, z).$$
(2.4)

We also recall that the Neumann function has a logarithmic singularity. In fact,

$$N_{\Omega}^{\omega}(x,z) = \frac{1}{2\pi} \log|x-z| + R_{\Omega}^{\omega}(x,z) \quad \forall x \neq z,$$
(2.5)

with  $R_{\Omega}^{\omega}$  being continuous on  $\Omega \times \Omega$  (for more details on the last two results, see [14, Section 2.3.5] and the references therein).

The perturbed eigenvalue problem. Let B be a bounded domain with piecewise smooth boundary, with area  $|B| = |\Omega|$ , and centered at the origin in the sense that

$$\int_{\partial B} y_1 \, d\sigma(y_1, y_2) = \int_{\partial B} y_2 \, d\sigma(y_1, y_2) = 0.$$

We fix for the rest of the chapter a point  $z \in \Omega$ , a scaling factor  $0 < \varepsilon \ll 1$  and an index  $\theta \in \mathbb{N}$ . Suppose then that the domain  $\Omega$  is perturbed by inserting a grounded inclusion  $D = z + \varepsilon B$  inside  $\Omega$ . This causes the eigenvalue  $\omega_{\theta}^2$  to split into  $m_{\theta}$ (possibly distinct) eigenvalues  $\omega_{\varepsilon,1}^2 \leq \cdots \leq \omega_{\varepsilon,m_{\theta}}^2$  with associated eigenfunctions  $u_{\varepsilon,1}, \ldots, u_{\varepsilon,m_{\theta}}$ . This means that for  $j = 1, \ldots, m_{\theta}$ ,

$$\begin{cases} (\Delta + \omega_{\varepsilon,j}^2) u_{\varepsilon,j} = 0 & \text{ in } \Omega \setminus D, \\ u_{\varepsilon,j} = 0 & \text{ on } \partial D, \\ \frac{\partial u_{\varepsilon,j}}{\partial \nu} = 0 & \text{ on } \partial \Omega. \end{cases}$$

It has been shown in [89] that under our assumptions  $\omega_{\varepsilon,j}^2 \to \omega_{\theta}^2$  as  $\varepsilon \to 0$ . The problem we will consider is the following:

to find an asymptotic expansion of  $\omega_{\varepsilon,j}^2 - \omega_{\theta}^2$  in terms of  $\varepsilon$ , for all  $j \in \{1, \ldots, m_{\theta}\}$ . With this purpose, we will transform the eigenvalue problem into an equivalent

integral formulation.

Nonstandard notation. In this chapter we will use the following notation:

- we indicate as  $\oint$  the normalized complex path integral  $\frac{1}{2\pi i} \int$ ;
- for clarity, we adopt the symbol  $\diamond$  to indicate the function variable of an operator evaluated at a point, e.g.  $\mathcal{D}_{\Omega}^{\omega}[\diamond](z)$  indicates the function  $L^{2}(\partial\Omega) \ni \varphi \mapsto \mathcal{D}_{\Omega}^{\omega}[\varphi](z) \in \mathbb{R};$
- given a multi-index  $\alpha = (\alpha_1, \dots, \alpha_d) \in \mathbb{N}^d$  we indicate as  $\partial^{\alpha}$  the normalized differential operator

$$\frac{1}{\alpha_1!\dots\alpha_n!}\frac{\partial}{\partial x_1^{\alpha_1}\dots\partial x_d^{\alpha_d}}$$

#### 2.1.1 Integral formulation

Define  $\mathcal{A}_{\varepsilon}(\omega)$  as

$$\mathbb{C} \ni \omega \quad \mapsto \quad \mathcal{A}_{\varepsilon}(\omega) = \begin{pmatrix} I/2 - \mathcal{K}_{\Omega}^{\omega} & -\mathcal{S}_{D}^{\omega} \\ & & \\ \mathcal{D}_{\Omega}^{\omega} & & \mathcal{S}_{D}^{\omega} \end{pmatrix},$$

meaning that for any fixed  $\omega \in \mathbb{C}$ ,  $\mathcal{A}_{\varepsilon}(\omega)$  is the operator which takes  $\phi \in L^2(\partial\Omega), \psi \in L^2(\partial D)$  to

By expanding the fundamental solution in Taylor series in  $\varepsilon$ , one can show that  $\mathcal{A}_{\varepsilon}$  has the same characteristic values as the operator  $\sum_{n=0}^{\infty} \varepsilon^n \mathcal{H}_n$  (the series converges in operator norm), where

$$\mathcal{H}_{0} = \begin{pmatrix} I/2 - \mathcal{K}_{\Omega}^{\omega} & -\Gamma_{\omega}(x, z) \int_{\partial B} \diamond(y) \ d\sigma(y) \\ \mathcal{D}_{\Omega}^{\omega}[\diamond](z) & \tilde{\mathcal{S}}_{B}^{\omega} \end{pmatrix},$$

$$\mathcal{H}_{n} = \begin{pmatrix} 0 & (-1)^{n+1} \sum_{|\alpha|=n} (\partial^{\alpha} \Gamma_{\omega})(x, z) \int_{\partial B} y^{\alpha} \diamond(y) \, d\sigma(y) \\ \\ \sum_{|\alpha|=n} (\partial^{\alpha} \mathcal{D}_{\Omega}^{\omega}[\diamond])(z) x^{\alpha} & \mathcal{X}_{n} \end{pmatrix},$$

with

$$\mathcal{X}_n = \begin{cases} \frac{\omega^n}{2^{n+1}n!\pi} \int_{\partial B} \log(\eta_{\frac{n}{2}} \omega \varepsilon |x-y|) |x-y|^n \diamond(y) \ d\sigma(y) & n \text{ even}, \\ 0 & n \text{ odd}, \end{cases}$$

$$\tilde{\mathcal{S}}_{B}^{\omega} = \frac{1}{2\pi} \int_{\partial B} \log(\eta_{0} \omega \varepsilon |x - y|) \diamond(y) \ d\sigma(y).$$
(2.6)

A study of the properties of  $A_{\varepsilon}$  can be found in [14, Chapter 1 and Section 3.1]). In the next proposition we collect only the properties which will be used in the following discussion. Recall that  $\omega \in \mathbb{C}$  is a characteristic value of  $A_{\varepsilon}$  if the null-space of  $A_{\varepsilon}(\omega)$  contains some non-zero function.

Proposition 2.1. The following results hold:

- (i)  $\omega \mapsto \mathcal{A}_{\varepsilon}(\omega)$  is analytic on  $\mathbb{C} \setminus (-\infty, 0)$  and  $\omega \mapsto \mathcal{A}_{\varepsilon}(\omega)^{-1}$  is meromorphic in  $\mathbb{C}$ ;
- (ii)  $\omega_{\theta}$  is a characteristic value of  $I/2 \mathcal{K}_{\Omega}^{\omega}$  and a simple pole of  $(I/2 \mathcal{K}_{\Omega}^{\omega})^{-1}$ ;
- (iii)  $(\omega_{\varepsilon,j})_{j=1,\dots,m_{\theta}}$  are characteristic values of  $\mathcal{A}_{\varepsilon}$ ;

(iv) There is an open neighbourhood V (which we fix for the rest of the chapter) of  $\omega_{\theta}$  such that  $\omega_{\varepsilon,j} \in V$  for  $j = 1, \ldots, m_{\theta}$ , and no other characteristic values of  $\mathcal{A}_{\varepsilon}$  are in V.

Consider now the power sum polynomials

$$p_l = \sum_{j=1}^{m_{\theta}} (\omega_{\varepsilon,j} - \omega_{\theta})^l.$$

By properties of symmetric polynomials we can express  $\omega_{\varepsilon,1} - \omega_{\theta}, \ldots, \omega_{\varepsilon,m_{\theta}} - \omega_{\theta}$ as roots of a polynomial  $z^{m_{\theta}} + c_1 z^{m_{\theta}-1} + \cdots + c_{m_{\theta}}$ , where the coefficients  $c_k$  are themselves polynomials in  $p_j$ ; in particular, the coefficients  $c_k$  can be recovered from the recurrence relations

$$p_{l+m_{\theta}} + c_1 p_{l+m_{\theta}-1} + \dots + c_{m_{\theta}} p_l$$
 for  $l = 0, \dots, m_{\theta} - 1$ .

Example 2.2. If  $m_{\theta} = 1$ , we have

$$\omega_{\varepsilon,1} - \omega_{\theta} = p_1,$$

while if  $m_{\theta} = 2$ , then

$$\omega_{\varepsilon,2} - \omega_{\theta} = \frac{p_1 + \sqrt{2p_2 - p_1^2}}{2}, \qquad \omega_{\varepsilon,1} - \omega_{\theta} = \frac{p_1 - \sqrt{2p_2 - p_1^2}}{2}.$$

Hence, we have reduced the problem of finding an asymptotic expansion of  $\omega_{\varepsilon,j}^2 - \omega_{\theta}^2$  to finding an asymptotic expansion of  $p_l$ . Before computing the explicit terms in the expansion of  $p_l$  we recall some crucial concepts from Gohberg-Sigal theory.

If A is a finite range operator on a Banach space, its trace tr A can be defined as the trace of A restricted to the finite dimensional space where A is non zero. In the next proposition we recall some properties of the trace which will be extensively used in the subsequent computations.

#### **Proposition 2.3.** The following results hold:

(i) Suppose A<sub>1</sub>, A<sub>2</sub>, A<sub>3</sub>, A<sub>4</sub> are finite dimensional operators on a Banach space.
 Then

$$\operatorname{tr} \begin{pmatrix} A_1 & A_2 \\ A_3 & A_4 \end{pmatrix} = \operatorname{tr} A_1 + \operatorname{tr} A_4.$$

(ii) Suppose B, C are operator valued maps defined on U, a neighborhood of a common singularity  $\nu \in \mathbb{C}$ . If B, C are analytic in  $U \setminus \nu$  and have only finite

dimensional operators in the negative terms of their Laurent expansion in  $\nu$ , then  $\int_{\partial U} B(\omega)C(\omega) d\omega$  is finite dimensional and

$$\operatorname{tr} \oint_{\partial U} B(\omega) C(\omega) \, d\omega = \operatorname{tr} \oint_{\partial U} C(\omega) B(\omega) \, d\omega.$$

(iii) If  $P_{\omega}$  is a projection from  $L^2$  to  $\mathbb{C}$  and  $f_{\omega} \in L^2$ , then

$$\operatorname{tr} \oint_{\partial V} f_{\omega} P_{\omega} \, d\omega = \oint_{\partial V} P_{\omega}[f_{\omega}] \, d\omega$$

An application of the argument principle for operator valued maps (Theorem 1.4) leads to the following crucial representation.

**Theorem 2.4.** We can rewrite  $p_l$  as

$$p_{l} = \operatorname{tr} \oint_{\partial V} (\omega - \omega_{\theta})^{l} \mathcal{H}_{0}(\omega)^{-1} \partial_{\omega} \mathcal{H}_{0}(\omega) d\omega + l \sum_{n=1}^{\infty} \varepsilon^{n} \sum_{j=1}^{n} \frac{(-1)^{j}}{j} \operatorname{tr} \oint_{\partial V} (\omega - \omega_{\theta})^{l-1} \left( \sum_{\substack{n_{1} + \dots + n_{j} = n \\ n_{i} \in \mathbb{Z}^{+}}} \prod_{k=1}^{j} \mathcal{H}_{0}(\omega)^{-1} \mathcal{H}_{n_{k}}(\omega) \right) d\omega.$$

The previous expression can be obtained by following the same algebraic manipulations as those in the proof of [14, Theorem 3.9].

#### 2.2 Computations for explicit formulae

First we highlight the quantities playing a key role in the expansion of  $p_l$  in the following definition.

**Definition 2.5.** Let  $\alpha, \beta$  be multi-indices in  $\mathbb{N}^2$ . The generalized capacity of B of order  $(\alpha, \beta)$ , at frequency  $\omega$  is

$$\mathfrak{c}_{\alpha,\beta}(\omega) = -\int_{\partial B} (\tilde{\mathcal{S}}_B^{\omega})^{-1} [\xi^{\alpha}](y) \, y^{\beta} \, d\sigma(y),$$

where  $\tilde{\mathcal{S}}^{\omega}_{B}$  is defined in (2.6). We also introduce

$$\mathfrak{t}(\omega) = \frac{U_{\theta}(z) \cdot \mathcal{D}_{\Omega}^{\omega}[U_{\theta}](z)}{\omega + \omega_{\theta}}, \qquad \mathfrak{r}(\omega) = \sum_{\substack{j=1\\j\neq\theta}}^{\infty} \frac{U_{j}(z) \cdot \mathcal{D}_{\Omega}^{\omega}[U_{j}](z)}{\omega^{2} - \omega_{j}^{2}}, \qquad (2.7)$$

where  $U_j$  is the vector of eigenfunctions associated to  $\omega_j^2$ , as defined in (2.2).

In the subsequent discussion, we will often indicate the generalized capacity of order  $(\alpha, 0)$  as  $\mathfrak{c}_{\alpha}$  instead of  $\mathfrak{c}_{\alpha,0}$ .

*Remark* 2.6. We collect some useful properties of the quantities introduced in the previous definition:

(i) The generalized capacity of order zero can be rewritten as

$$\mathfrak{c}_0(\omega) = -\int_{\partial B} (\tilde{\mathcal{S}}_B^{\omega})^{-1}[1] = -\frac{2\pi}{\log(\eta_0 \omega \varepsilon \operatorname{cap} \partial B)}.$$

- (ii) It holds  $\mathbf{c}_{\alpha,\beta} = 0$  if  $|\alpha| + |\beta|$  is odd. This is a consequence of the fact that  $\varphi$  is even/odd if and only if  $\mathcal{S}_B^0[\varphi]$  is even/odd (as functions parametrized on  $\partial B$ ).
- (iii) Although the series defining r in (2.7) does not converge absolutely, Weyl's law (which states that ω<sub>j</sub><sup>2</sup> ~ j) and the oscillatory nature of the eigenfunctions of Ω evaluated at z suggest that the series converges conditionally.
- (iv) By exploiting the spectral expansion of the Neumann function (2.5), we have that

$$\mathcal{D}^{\omega}_{\Omega}[N^{\omega}_{\Omega}(\cdot, z)](z) = \frac{\mathfrak{t}(\omega)}{\omega - \omega_{\theta}} + \mathfrak{r}(\omega).$$
(2.8)

In the subsequent calculations, this identity will enable us to rewrite the expansion in Theorem 2.4 in terms of  $\mathfrak{t}(\omega_{\theta})$  and  $\mathfrak{r}(\omega_{\theta})$ .

(v) By Green's identity and the defining property of  $\Gamma_{\omega_{\theta}}$  and  $U_{\theta}$ , we can compute the special value

$$\mathfrak{t}(\omega_{\theta}) = \frac{|U_{\theta}(z)|^2}{2\omega_{\theta}}.$$
(2.9)

#### 2.2.1 Zero-order term

**Lemma 2.7.** The zero-order term in the expansion in powers of  $\varepsilon$  of  $p_l$  is

$$\left(\frac{\mathfrak{t}(\omega_{\theta})}{1/\mathfrak{c}_{0}(\omega_{\theta}) - \mathfrak{r}(\omega_{\theta})}\right)^{l}.$$
(2.10)

*Proof.* By Theorem 2.4, our problem reduces to compute explicitly

$$\operatorname{tr} \oint_{\partial V} (\omega - \omega_{\theta})^{l} \mathcal{H}_{0}(\omega)^{-1} \partial_{\omega} \mathcal{H}_{0}(\omega) \, d\omega$$

To make further computations clearer and more concise, we rename

$$A = I/2 - \mathcal{K}_{\Omega}^{\omega}[\diamond](x), \qquad \Gamma = \Gamma_{\omega}(x, z),$$
$$N = N_{\Omega}^{\omega}(x, z), \qquad D = \mathcal{D}_{\Omega}^{\omega}[\diamond](z), \qquad (2.11)$$
$$\mathfrak{s} = \omega - \omega_{\theta}.$$

The characteristic values of  $\mathcal{H}_0$  are the  $\omega \in \mathbb{C}$  for which there exist  $\phi \in L^2(\Omega), \psi \in L^2(B)$ , at least one of them non-zero, such that

$$\begin{cases} A\phi - \Gamma \int_{\partial B} \psi = 0, \\ D\phi + \tilde{\mathcal{S}}_{B}^{\omega} \psi = 0. \end{cases}$$

By recalling from (2.3) that  $\tilde{\mathcal{S}}_B^{\omega}$  is invertible, applying  $(\tilde{\mathcal{S}}_B^{\omega})^{-1}$  and integrating the second equation of the system, we obtain

$$\int_{\partial B} \psi = \mathfrak{c}_0 D \phi$$

Substituting this back into the first equation, we have that the characteristic values of the system correspond to the characteristic values of the operator

$$H = A - \mathfrak{c}_0 \Gamma D.$$

Therefore, the coefficient we are looking for will be given by

$$E = \operatorname{tr} \oint_{\partial V} \mathfrak{s}^{l} H^{-1} H' \, d\omega,$$

where ' denotes differentiation with respect to  $\omega$ . A straightforward calculation shows that

$$H^{-1} = (I - \mathfrak{c}_0 ND)^{-1} A^{-1} = \sum_{m=0}^{\infty} (\mathfrak{c}_0 ND)^m A^{-1},$$

(where the *m* exponent indicates *m* times the composition of  $c_0 ND$ ) and

$$H' = A' - \Gamma' \mathfrak{c}_0 D - \Gamma(\mathfrak{c}_0 D)'.$$

Then

$$H^{-1}H' = A^{-1}A' + \sum_{m=1}^{\infty} (\mathfrak{c}_0 ND)^m A^{-1}A' - (\mathfrak{c}_0 ND)^{m-1}A^{-1}\Gamma'\mathfrak{c}_0 D - (\mathfrak{c}_0 ND)^{m-1}N(\mathfrak{c}_0 D)'.$$

Since A is analytic in V,  $A^{-1}$  has a simple pole at  $\omega_{\theta}$ , and we are only interested in the case  $l \geq 1$ . Hence, we have

$$\oint_{\partial V} \mathfrak{s}^l A^{-1} A' = 0$$

Then, by Item iii of Proposition 2.3,

$$E = \sum_{m=1}^{\infty} \operatorname{tr} \oint_{\partial V} \mathfrak{s}^{l} \left( (\mathfrak{c}_{0} N D)^{m} A^{-1} A' - N^{m-1} (\mathfrak{c}_{0} D)^{m} A^{-1} \Gamma' - N^{m} (\mathfrak{c}_{0} D)^{m-1} (\mathfrak{c}_{0} D)' \right)$$
  
$$= \sum_{m=1}^{\infty} \oint_{\partial V} \mathfrak{s}^{l} \left( \mathfrak{c}_{0}^{m} (DN)^{m-1} D A^{-1} A' A^{-1} \Gamma - \mathfrak{c}_{0}^{m} (DN)^{m-1} D A^{-1} \Gamma' - (\mathfrak{c}_{0} DN)^{m-1} (\mathfrak{c}_{0} D)' N \right).$$

Since  $(A^{-1})' = -A^{-1}A'A^{-1}$ , by applying multiple times the chain rule, we arrive at

$$E = -\sum_{m=1}^{\infty} \frac{1}{m} \oint_{\partial V} \mathfrak{s}^{l} \left( (\mathfrak{c}_{0} DN)^{m} \right)^{\prime}.$$

Then, by an integration by parts followed by a binomial expansion of  $(DN)^m$ , we have that

$$E = \sum_{m=1}^{\infty} \frac{l}{m} \oint_{\partial V} \mathfrak{s}^{l-1} (\mathfrak{c}_0 DN)^m$$
  
= 
$$\sum_{m=1}^{\infty} \frac{l}{m} \oint_{\partial V} \mathfrak{s}^{l-1} \mathfrak{c}_0^m \left(\frac{\mathfrak{t}}{\mathfrak{s}} + \mathfrak{r}\right)^m$$
  
= 
$$\sum_{m=1}^{\infty} \frac{l}{m} \sum_{k=0}^m \binom{m}{k} \oint_{\partial V} \frac{1}{\mathfrak{s}^{k-l+1}} \mathfrak{c}_0^m \mathfrak{t}^k \mathfrak{r}^{m-k}.$$
 (2.12)

Since the only pole in V of the integrand is  $\omega_{\theta}$ , by applying the residue theorem we can cancel each addend of the sum on k except the one corresponding to a pole of order 1, obtaining

$$E = l\mathfrak{t}(\omega_{\theta})^{l}\mathfrak{r}(\omega_{\theta})^{-l}\sum_{m=l}^{\infty}\frac{1}{m}\binom{m}{l}(\mathfrak{c}_{0}(\omega_{\theta})\mathfrak{r}(\omega_{\theta}))^{m}.$$

A final application of the identity

$$\sum_{m=l}^{\infty} \frac{1}{m} \binom{m}{l} x^m = \frac{1}{l} \left( \frac{x}{1-x} \right)^l,$$

leads to (2.10).

г			
L			
L			
	-	_	

#### 2.2.2 First-order term

**Lemma 2.8.** The coefficient of the  $\varepsilon$  term in the expansion of  $p_l$  is zero.

*Proof.* With the notation introduced in (2.11),

$$\mathcal{H}_1 = \sum_{|\alpha|=1} \begin{pmatrix} 0 & (\partial^{\alpha} \Gamma) \int_{\partial B} y^{\alpha} \diamond \\ \\ (\partial^{\alpha} D) x^{\alpha} & 0 \end{pmatrix}.$$

By applying the blockwise inversion formula,

$$\begin{pmatrix} W & X \\ Y & Z \end{pmatrix}^{-1} = \begin{pmatrix} (W - XZ^{-1}Y)^{-1} & -W^{-1}X(Z - YW^{-1}X)^{-1} \\ -Z^{-1}Y(W - XZ^{-1}Y)^{-1} & (Z - YW^{-1}X)^{-1} \end{pmatrix},$$

to calculate  $\mathcal{H}_0^{-1}$ , and rewriting the inverses of sums of operators in a Neumann series, we obtain

$$\mathcal{H}_{0}^{-1} = \begin{pmatrix} (Nc_{0}D)^{m}A^{-1} & N\int_{\partial B}(-(\tilde{\mathcal{S}}_{B}^{\omega})^{-1}[1]DN\int_{\partial B})^{m}(\tilde{\mathcal{S}}_{B}^{\omega})^{-1} \\ -(\tilde{\mathcal{S}}_{B}^{\omega})^{-1}[1]D(Nc_{0}D)^{m}A^{-1} & (-(\tilde{\mathcal{S}}_{B}^{\omega})^{-1}[1]DN\int_{\partial B})^{m}(\tilde{\mathcal{S}}_{B}^{\omega})^{-1} \end{pmatrix}.$$

A straightforward computation leads to

$$\mathcal{H}_0^{-1}\mathcal{H}_1 = \sum_{m=0}^{\infty} \sum_{|\alpha|=1} \begin{pmatrix} \mathfrak{c}_{\alpha}(\dots) & (\dots) \\ \\ (\dots) & -(\tilde{\mathcal{S}}_B^{\omega})^{-1}[1](\dots) \int_{\partial B} y^{\alpha} \diamond \end{pmatrix},$$

where we omit most of the terms by writing (...) instead. They indeed do not count towards our calculations, since from the fact that  $\mathfrak{c}_{\alpha} = 0$  for  $|\alpha| = 1$ , we have that the coefficient of  $\varepsilon$  is

$$\operatorname{tr} \oint_{\partial V} \mathfrak{s}^{l-1} \mathcal{H}_0^{-1} \mathcal{H}_1 = \operatorname{tr} \oint_{\partial V} \mathfrak{s}^{l-1} \left( (\mathcal{H}_0^{-1} \mathcal{H}_1)_{11} + (\mathcal{H}_0^{-1} \mathcal{H}_1)_{22} \right) = \mathfrak{c}_\alpha(\dots) = 0.$$

#### 2.2.3 On higher-order terms

In this subsection we propose a method to calculate explicitly any coefficient of the expansion of  $p_l$ . To the shorthand notation introduced in (2.11) we add a = DN,  $S = \tilde{S}^{\omega}_B, \phi = -S^{-1}[1]$ . Then, we can rewrite

$$\mathcal{H}_{0}^{-1} = \begin{pmatrix} A^{-1} & N \int_{\partial B} S^{-1} \\ \phi D A^{-1} & S^{-1} \end{pmatrix} + \sum_{m=1}^{\infty} \begin{pmatrix} N \mathfrak{c}_{0}^{m} a^{m-1} D A^{-1} & N(\mathfrak{c}_{0} a)^{m} \int_{\partial B} S^{-1} \\ \phi(\mathfrak{c}_{0} a)^{m} D A^{-1} & \phi \mathfrak{c}_{0}^{m-1} a^{m} \int_{\partial B} S^{-1} \end{pmatrix},$$

and

$$\mathcal{H}_n = \sum_{|\alpha|=n} \begin{pmatrix} 0 & (-1)^{n+1} (\partial^{\alpha} \Gamma) \int_{\partial B} y^{\alpha} \diamond (y) \ d\sigma(y) \\ (\partial^{\alpha} D) x^{\alpha} & \mathcal{X}_n \end{pmatrix}$$

An explicit computation leads to

$$-\mathcal{H}_{0}^{-1}\mathcal{H}_{n} = \sum_{|\alpha|=n} \begin{pmatrix} Nc_{\alpha}\partial^{\alpha}D & (-1)^{n}A^{-1}\partial^{\alpha}\Gamma\int_{\partial B}y^{\alpha}\diamond -N\int_{\partial B}S^{-1}\mathcal{X}_{n} \\ c_{\alpha}\partial^{\alpha}D & (-1)^{n}\phi DA^{-1}\partial^{\alpha}\Gamma\int_{\partial B}y^{\alpha}\diamond -S^{-1}\mathcal{X}_{n} \end{pmatrix}$$
$$+\sum_{m=1}^{\infty} \begin{pmatrix} N(\mathfrak{c}_{0}a)^{m}\mathfrak{c}_{\alpha}\partial^{\alpha}D & (-1)^{n}N\mathfrak{c}_{0}^{m}a^{m-1}DA^{-1}(\partial^{\alpha}\Gamma)\int_{\partial B}y^{\alpha}\diamond -N(\mathfrak{c}_{0}a)^{m}\int_{\partial B}S^{-1}\mathcal{X}_{n} \\ \phi\mathfrak{c}_{0}^{m-1}a^{m}\mathfrak{c}_{\alpha}\partial^{\alpha}D & (-1)^{n}\phi(\mathfrak{c}_{0}a)^{m}DA^{-1}(\partial^{\alpha}\Gamma)\int_{\partial B}y^{\alpha}\diamond -\phi\mathfrak{c}_{0}^{m-1}a^{m}\int_{\partial B}S^{-1}\mathcal{X}_{n} \end{pmatrix}$$

Since all elements of the matrix are one-dimensional projection operators on either  $N, \phi$  or 1, the operator  $-\mathcal{H}_0^{-1}\mathcal{H}_n$  has the same characteristic values of the projection operator on 1 given by

$$\sum_{|\alpha|=n} \begin{pmatrix} c_{\alpha}(\partial^{\alpha}D)N & (-1)^{n}\int_{\partial B} y^{\alpha}A^{-1}\partial^{\alpha}\Gamma - \int_{\partial B} S^{-1}\mathcal{X}_{n}N \\ c_{\alpha}(\partial^{\alpha}D)[1] & (-1)^{n}\mathfrak{c}_{0,\alpha}DA^{-1}\partial^{\alpha}\Gamma - S^{-1}\mathcal{X}_{n}[1] \end{pmatrix}$$
  
+
$$\sum_{m=1}^{\infty} \begin{pmatrix} (\mathfrak{c}_{0}a)^{m}\mathfrak{c}_{\alpha}(\partial^{\alpha}D)N & (-1)^{n}\mathfrak{c}_{0}^{m}a^{m-1}DA^{-1}(\partial^{\alpha}\Gamma)\int_{\partial B} y^{\alpha}N - (\mathfrak{c}_{0}a)^{m}\int_{\partial B} S^{-1}\mathcal{X}_{n}N \\ \mathfrak{c}_{0}^{m-1}a^{m}\mathfrak{c}_{\alpha}(\partial^{\alpha}D)\phi & (-1)^{n}(\mathfrak{c}_{0}a)^{m}DA^{-1}(\partial^{\alpha}\Gamma)\mathfrak{c}_{0,\alpha} - \mathfrak{c}_{0}^{m-1}a^{m}\int_{\partial B} S^{-1}\mathcal{X}_{n}\phi \end{pmatrix}.$$

$$(2.13)$$

Suppose that the elements of the matrices in (2.13) can be rewritten explicitly in terms of sums of powers of the singularity  $\mathfrak{s}$ . The same approach used to derive the zero-order term in the previous section could then be applied to compute explicitly

$$\operatorname{tr} \oint_{\partial V} \mathfrak{s}^{l-1} \mathcal{H}_0^{-1} \mathcal{H}_{n_1} \dots \mathcal{H}_0^{-1} \mathcal{H}_{n_j}.$$

Substituting this value back in the expression for  $p_l$  in Theorem 2.4, we could compute, for example with the aid of a computer algebra system, any of the coefficients of the expansion in  $\varepsilon$ .

However, the task of rewriting explicitly the singularity  $\mathfrak{s}$  in (2.13) for  $\alpha \geq 1$  is not trivial, as identities similar to (2.4) and (2.8), but involving the terms  $(\partial^{\alpha} D)N$ ,  $A^{-1}\partial^{\alpha}\Gamma$  and  $S^{-1}\mathcal{X}_nN$  would need to be determined.

#### 2.3 Results for special cases

We collect in this final section some interesting results which follow directly, or with minor algebraic manipulations, from Lemmas 2.7, 2.8, and Example 2.2.

**Proposition 2.9.** Suppose that  $\omega_{\theta}$  is simple.

(i) We have

$$\omega_{\varepsilon,1} - \omega_{\theta} = \frac{\mathfrak{t}(\omega_{\theta})}{1/\mathfrak{c}_0 - \mathfrak{r}(\omega_{\theta})} + O(\varepsilon^2),$$

where, recalling Definition 2.5 and (2.9),

$$1/\mathfrak{c}_0 = -\frac{\log(\eta_0\omega_\theta\varepsilon\operatorname{cap}\partial B)}{2\pi}, \quad \mathfrak{t}(\omega_\theta) = \frac{|U_\theta(z)|^2}{2\omega_\theta}, \quad \mathfrak{r}(\omega_\theta) = \sum_{\substack{j=1\\ j\neq\theta}}^{\infty} \frac{U_j(z)\cdot\mathcal{D}_{\Omega}^{\omega_\theta}[U_j](z)}{\omega_\theta^2 - \omega_j^2};$$

(ii) For  $\varepsilon$  small enough, we can deduce that  $\omega_{\varepsilon,1} \geq \omega_{\theta}$ .

**Proposition 2.10.** If  $\omega_{\theta}$  has double multiplicity then

$$\omega_{\varepsilon,2} - \omega_{\theta} = \frac{\mathfrak{t}(\omega_{\theta})}{1/\mathfrak{c}_0 - \mathfrak{r}(\omega_{\theta})} + O(\varepsilon^2),$$
$$\omega_{\varepsilon,1} - \omega_{\theta} = O(\varepsilon^2).$$

We notice that by considering an expansion in powers of  $1/\log(\varepsilon)$ , we obtain

$$\frac{\mathfrak{t}(\omega_{\theta})}{1/\mathfrak{c}_{0} - \mathfrak{r}(\omega_{\theta})} + O(\varepsilon^{2}) = -\frac{1}{\log \varepsilon} \frac{\pi |U_{\theta}(z)|^{2}}{\omega_{\theta}} + O\left(\frac{1}{\log(\varepsilon)^{2}}\right).$$

In particular, substituting  $2\omega_{\theta} \simeq \omega_{\epsilon} + \omega_{\theta}$  gives back (2.1).

We also notice that if z is on a nodal set of  $U_{\theta}$  (i.e.  $u_{\theta,1}, \ldots, u_{\theta,m_{\theta}}$  are all zero at z) then  $\mathfrak{t}(\omega_{\theta}) = 0$ , and thus in both the cases of a simple or a double eigenvalue, the splitting order will be  $O(\varepsilon^2)$ .

#### Numerical validation for a disk domain and a disk inclusion

Let  $\Omega$  be the unit disk and let  $\omega_{\theta}^2$  be its first non-zero eigenvalue. It is known that  $\omega_{\theta}$  is given by the first root of the derivative of the Bessel function  $J_1$  and has double multiplicity. Suppose that also the rescaled inclusion B is a unit disk.

In Figure 2.1, we compare results obtained through the multipole expansion method with the  $\varepsilon^2$  error given by the formula for  $\omega_{\varepsilon,1} - \omega_{\theta}$ . The multipole expansion is implemented by setting two polar coordinate systems, one centered in the center of  $\Omega$  and one in z, and exploiting Graf's summation formula for Bessel functions to rewrite the eigenvalue problem as a root search for the determinant of the coordinate transformation matrix.

In Figures 2.2 and 2.3, we compare asymptotic formulae for  $\omega_{\varepsilon,2} - \omega_{\theta}$  with results obtained with the multipole expansion method. The asymptotic formula is implemented numerically by truncating at a finite value the series in the definition of  $\mathfrak{r}(\omega_{\theta})$ in (2.7), and approximating the boundary layer integrals in  $\mathfrak{r}(\omega_{\theta})$  by quadrature.

We remark that the improved resolution of the inclusion size and position opens the possibility of the development of accurate inclusion reconstruction algorithms using the asymptotic formulae from Propositions 2.9 and 2.10. Of particular interest, also in applications, would be the development of an efficient reconstruction method for locating and characterizing small inclusions based only on the knowledge of eigenvalues shift. The implementation of such a method and its applications are studied in Chapter 5.



Figure 2.1: A  $\log_2-\log_2$  plot of  $\omega_{\varepsilon,1} - \omega_{\theta}$  as the size of the inclusion varies and the center is fixed (at |z| = .5).



Figure 2.2: A  $\log_2-\log_2$  plot of the shift  $\omega_{\varepsilon,2} - \omega_{\theta}$  as the size of the inclusion varies and its center remains constant (left at |z| = .3, right at |z| = .8).



Figure 2.3: A plot of the shift  $\omega_{\varepsilon,2} - \omega_{\theta}$  as the distance from the origin of the center of the inclusion varies and its size remains constant (left at  $\varepsilon = 10^{-2}$ , right at  $\varepsilon = 10^{-4}$ ).

## Chapter 3

## Arbitrarily small domain perturbations which split the spectrum of the Laplacian

In the seminal works [79] and [96], respectively Micheletti and Uhlenbeck showed that the eigenvalues of the Dirichlet Laplacian are generically simple in the space of smooth manifolds equipped with the  $C^k$ -topology (for subsequent works see also [80], the survey papers [33, Section 4.3], [52, Section 1.3] and references therein). In this chapter, we prove that a localized version of this result holds as follows, even for non-smooth domains.

**Theorem 3.1.** For any Lipschitz domain  $\Omega$  and for any open set U whose intersection with  $\Omega$  is not empty, there exists a domain  $\tilde{\Omega}$  whose symmetric difference with  $\Omega$  is contained in U and whose Dirichlet/Neumann/Robin Laplacian eigenvalues are all simple. Moreover if  $U \cap \partial \Omega$  is not empty, then  $\tilde{\Omega}$  can be taken to be bi-Lipschitz homeomorphic to  $\Omega$ .

More in detail, the structure of the chapter is the following. In Section 3.1, we review some preliminary material, in particular regarding spectral stability. In Section 3.2, we recall a Hadamard's formula and study independence properties of some expressions involving eigenfunctions and their gradients at the boundary. More specifically, Hadamard's formula provides us with a first-order estimate on the shift of an eigenvalue  $\lambda$ , and which depends on the quantity

$$|\nabla u|^2 - hu^2 \tag{3.1}$$

at the boundary of the domain considered, where u is an eigenfunction associated to  $\lambda$  and h is a function which depends on the choice of boundary conditions. By showing that for two orthogonal eigenfunctions, the corresponding values of (3.1) must differ at least at a point, we are able to construct a localized perturbation which splits any non-simple eigenvalue. However, even when small, this perturbation might cause the shift and the overlap of other eigenvalues. This possibility is ruled out in Section 3.3, where uniform bounds for the whole spectrum are adapted to our case from sharp stability estimates from [32]. These bounds allow the construction of a localized perturbation, which consists in a sequence of successive smaller "bumps" at the boundary of a part cut out from  $\Omega$ , which proves the first statement of Theorem 3.1. We prove its second statement in Section 3.4, where we use an additional result from [81] which allows us to deform appropriately directly the boundary of  $\partial\Omega$ , without requiring any topological change of  $\Omega$ .

We remark that under homogeneous Dirichlet boundary conditions, the splitting of an eigenvalue by the application of Holmgren's uniqueness Theorem and Hadamard's formula is a well-known technique (see for instance [57, Lemma 2.5.9]); also in the case of homogeneous Robin boundary conditions with a never-vanishing impedance coefficient the arguments of [81] can be easily adapted for the same purpose (see the proof of Proposition 3.10). Thus, our efforts will be devoted mainly to the analysis of the novel cases of homogeneous Neumann boundary conditions and of *localized* domain perturbations.

#### 3.1 Preliminaries

We collect here some definitions which will be used extensively in the chapter.

(i) We say that  $\lambda$  is an *eigenvalue* of a domain  $\Omega$  with associated *eigenfunction* u if

$$\Delta u + \lambda u = 0 \tag{3.2}$$

in  $\Omega$ , u is not constant zero, and either one of the following homogeneous boundary conditions is satisfied on  $\partial \Omega$ :

$$\begin{cases} u = 0 & \text{(Dirichlet)}, \\ \alpha u + \frac{\partial u}{\partial \nu} = 0 & \text{(Neumann/Robin)}, \end{cases}$$
(3.3)

where  $\alpha$  is a constant ( $\alpha = 0$  for the Neumann condition,  $\alpha > 0$  for the Robin condition), and  $\nu$  indicates the outward unit normal vector.

(ii) We say that  $(\phi_t)_{t \in [0,t_0)}$  is a *deformation* if  $\phi_t$  is a diffeomorphism of  $\mathbb{R}^N$ ,  $\phi_t$ is analytic in t, and  $|\phi_t - I|_{C^2(\mathbb{R}^N)} \leq Ct$  for a certain constant C, for every  $t \in [0, t_0)$ . We say that  $(\phi_t)_t$  is *localized* in U if  $\phi_t$  is the identity outside U for all t.

We actually require (3.2) and (3.3) to be satisfied only in a weak sense, that is:  $\lambda$  is an eigenvalue of  $\Omega$  with associated eigenfunction u, if u is an element of a function space  $V(\Omega)$  and

$$Q_{\Omega}(u,v) = \lambda \int_{\Omega} uv, \quad \text{for every } v \in V(\Omega),$$

where, depending on the choice of boundary conditions, we define

Boundary conditions
$$Q_{\Omega}(u,v)$$
 $V(\Omega)$ Dirichlet $\int_{\Omega} \nabla u \cdot \nabla v$  $H_0^1(\Omega)$ Neumann/Robin $\int_{\Omega} \nabla u \cdot \nabla v + \int_{\partial\Omega} \alpha uv$  $H^1(\Omega)$ 

where  $\mathrm{H}^1$  is the space of square integrable functions with square integrable distributional gradient and  $\mathrm{H}_0^1$  its subspace of trace zero functions. However, from elliptic regularity theory, we know that Laplacian eigenfunctions are analytic inside any open domain; thus (3.2) is satisfied also in the classical sense. Moreover, if  $\Sigma$  is a smooth part of  $\partial\Omega$ , u is also smooth on  $\Sigma$  (see for example [41, Section 6.3] for proofs of these facts).

Recall that from spectral theory, the eigenvalues of  $\Omega$  have finite multiplicity and can be arranged in a non-decreasing sequence which tends to infinity, and which we will denote as

$$0 \leq \lambda_1 \leq \lambda_2 \leq \ldots,$$

where each eigenvalue is repeated as many times as its multiplicity.

For future reference, we record the following uniqueness result.

**Theorem 3.2.** Let u be such that  $\Delta u + \lambda u = 0$  in  $\Omega$ . If u = 0 and  $\partial u / \partial \nu = 0$  on  $\Sigma$ , an open and smooth subset of  $\partial \Omega$ , then u is identically zero in the whole  $\Omega$ .

We briefly outline the classic argument to prove this fact from Holmgren's uniqueness theorem. Let B be an open ball such that  $B \cap \partial \Omega \subseteq \Sigma$ . Extending u to 0 in  $B \setminus \Omega$ , it is easy to check that  $-\Delta u = \lambda u$  in the distributional sense in B. By [60, Theorem 5.3.1], u must be zero also in an open set inside  $\Omega$ . But then u = 0 on the whole  $\Omega$  by analytic continuation.

#### 3.1.1 Stability of eigenvalues of the Laplacian

In this section, we review some results that show that the spectrum of the Laplacian is continuous under domain perturbations, and give some useful quantitative estimates on the eigenvalues' shifts.

We recall a result of analyticity of eigenvalues and eigenfunctions with respect to a perturbation parameter, which is a consequence of the classic Rellich-Nagy Theorem [90, Theorem 1 at p. 33] (see also [33, Section 4.2] and references therein).

**Theorem 3.3.** Let  $(\phi_t)_t$  be a deformation and let  $\lambda$  be an eigenvalue of  $\Omega$  of multiplicity m. Then, there exist  $\lambda_t^1 \leq \cdots \leq \lambda_t^m$  and functions  $u_t^1, \ldots, u_t^m$  such that for  $j = 1, \ldots, m$ ,

- (i) for any t,  $\lambda_t^j$  is an eigenvalue of  $\phi_t(\Omega)$  with associated eigenfunction  $u_t^j$ ;
- (ii) for any t,  $\int_{\phi_t(\Omega)} u_t^i u_t^j$  is 1 if i = j and is 0 for every  $i \neq j$ ;
- (iii)  $\lambda_t^j$  and  $u_t^j$  are analytic in t;
- (iv)  $\lambda_0^j = \lambda$  and  $u_0^j$  is an eigenfunction associated to  $\lambda$ .

Moreover, for any  $\delta > 0$  small enough, there is a T such that for any t < T the only eigenvalues of  $\phi_t(\Omega)$  in  $(\lambda - \delta, \lambda + \delta)$  are  $\lambda_t^1, \ldots, \lambda_t^m$ .

For our purposes, we will also need a finer estimate on the variation of eigenvalues, as expressed in the following lemma.

**Lemma 3.4.** Let  $\phi$  be a diffeomorphism of  $\mathbb{R}^N$ , let  $\lambda_n$  be the n-th eigenvalue of  $\Omega$  and  $\tilde{\lambda}_n$  be the n-th eigenvalue of  $\phi(\Omega)$ . Then there exists a constant C, which depends only on the Lipschitz constant of  $\partial\Omega$ , such that

$$|\tilde{\lambda}_n - \lambda_n| \le C \max\{\tilde{\lambda}_n, \lambda_n\} (|\phi - I|_{\mathcal{C}^1(\mathbb{R}^N)}).$$

The proof of this estimate can be obtained by repeating the same argument from the proof of [32, Lemma 6.1], only substituting appropriately the bilinear form and the function space with the ones defined in (3.4), depending on the boundary conditions considered.

## 3.2 Hadamard's formula and boundary properties of eigenfunctions

In this section, we study some independence properties of Laplacian eigenfunctions and of their gradients at the boundary. We first recall a Hadamard's formula for the variation of eigenvalues under a deformation. The dot superscript will indicate differentiation in t.

**Lemma 3.5.** Let  $(\phi_t)_t$  be a deformation. Let  $\lambda_t, u_t$  be an eigenvalue-eigenfunction pair of  $\phi_t(\Omega)$ , and suppose both are differentiable in t. Then

$$\dot{\lambda}_0 = \int_{\partial\Omega} \left( |\nabla u|^2 - \lambda u^2 + (\partial u/\nu_0) (Hu - 2\partial u/\nu_0) \right) \nu_0 \cdot \dot{e}_0, \tag{3.5}$$

where  $\lambda = \lambda_0$ ,  $u = u_0$ ,  $\nu_t$  indicates the outward unit normal vector,  $e_t$  the identity on  $\phi_t(\partial \Omega)$ , and H the mean curvature of  $\partial \Omega$ .

Hereafter, we briefly prove this fact in the case of Dirichlet or Neumann boundary conditions. The case of Robin conditions requires a finer analysis of the dependence on t of the surfaces  $\phi_t(\partial\Omega)$ , for which we refer to [51, Identity (9)] or [28, Identities (69) and (57)].

Proof. Let  $\Omega_t = \phi_t(\Omega)$  for every t. By the divergence theorem, the distributional gradient of the measure  $\mathbb{1}_{\Omega_t} \mathcal{L}^N$  is given by  $\nu_t \Sigma_t^{N-1}$ , where  $\mathbb{1}_{\Omega_t}$  is the characteristic function of  $\Omega_t$ ,  $\mathcal{L}^N$  is the N-dimensional Lebesgue measure, and  $\Sigma_t^{N-1}$  is the surface measure on  $\partial \Omega_t$ . Therefore, by the chain rule,

$$\frac{d}{dt}(\mathbb{1}_{\Omega_t} \mathcal{L}^N) = \nu_t \cdot \dot{e}_t \Sigma_t^{N-1},$$

so we have the following Leibniz' formula:

$$\frac{d}{dt}\left(\int_{\Omega_t} f_t\right) = \int_{\Omega_t} \dot{f}_t + \int_{\partial\Omega_t} f_t \nu_t \cdot \dot{e}_t.$$
(3.6)

Consider now the identity

$$\lambda_t = -\int_{\Omega_t} u_t \Delta u_t = \int_{\Omega_t} |\nabla u_t|^2.$$
(3.7)

Differentiating in t the first equality in (3.7) and using (3.6), we obtain

$$2\lambda_t \int_{\Omega_t} \dot{u}_t u_t = -\lambda_t \int_{\partial\Omega_t} u_t^2 \nu_t \cdot \dot{e}_t.$$
(3.8)

35

In the case of Neumann boundary conditions, differentiating in t the last term in (3.7), using (3.6), integrating by parts, and substituting (3.8), we have that

$$\dot{\lambda}_t = \int_{\partial\Omega_t} (|\nabla u_t|^2 - \lambda_t u_t^2) \nu_t \cdot \dot{e}_t + 2 \int_{\partial\Omega_t} \dot{u}_t \frac{\partial u_t}{\partial\nu_t},$$

which gives (3.5) since  $\partial u_0 / \partial \nu_0 = 0$  on  $\partial \Omega_0$ . Proceeding in the same way for Dirichlet boundary conditions, only exchanging the roles of the functions in the integration by parts step, we obtain

$$\dot{\lambda}_t = \int_{\partial\Omega_t} (|\nabla u_t|^2 - \lambda_t u_t^2) \nu_t \cdot \dot{e}_t + 2 \int_{\partial\Omega_t} u_t \frac{\partial \dot{u}_t}{\partial\nu_t} + 2\dot{\lambda}_t,$$

which gives (3.5), since  $u_0 = 0$  on  $\partial \Omega_0$ .

We notice that considering

$$h = \begin{cases} 0 & \text{if } u = 0 \text{ on } \partial\Omega, \\ \lambda + \alpha H + 2\alpha^2 & \text{if } \alpha u + \partial u/\nu = 0 \text{ on } \partial\Omega, \end{cases}$$
(3.9)

the integrand in parentheses in (3.5) can be rewritten as  $|\nabla u|^2 - hu^2$ . In the following lemma we study such a quantity, in particular the behavior of its zeros.

**Lemma 3.6.** Let E be a smooth domain and let S be a connected component of its boundary. Let u and v be two orthonormal eigenfunctions of E, associated to the same eigenvalue  $\lambda$ . Let H be the mean curvature of S and let h be as in (3.9). Then, under Dirichlet or Neumann boundary conditions, neither of the quantities

$$|\nabla u|^2 - hu^2, \tag{3.10}$$

$$|\nabla u|^2 - hu^2 - (|\nabla v|^2 - hv^2), \qquad (3.11)$$

can be identically zero on S. The same statement holds in the case of Robin boundary conditions under the additional requirement that the mean curvature satisfies  $-\alpha H < \alpha^2 + \lambda$ .

*Proof.* Suppose that (3.10) is zero on S. If the Dirichlet condition holds, then  $\partial u/\partial \nu = u = 0$  on S. Thus, by Theorem 3.2 we have that u = 0 in E, which is a contradiction.

If the Neumann condition holds, then from the assumption that (3.10) is zero, we have

$$|\nabla_S u|^2 = \lambda u^2, \tag{3.12}$$
where  $\nabla_S$  is the surface gradient on S. By Theorem 3.2, the eigenfunction u cannot be constant zero on S, so there exists a point  $x_0 \in S$  such that  $u(x_0) \neq 0$ . The approach we follow hereafter is inspired by [58, Chapter 6]. Let  $\gamma : (-\infty, +\infty) \to S$ be a solution of the Cauchy problem

$$\begin{cases} \gamma_0 = x_0, \\ \dot{\gamma}_t = (\nabla_S u)(\gamma_t). \end{cases}$$

Existence and uniqueness of such a  $\gamma$  for every t is guaranteed by standard results in the theory of ordinary differential equations on surfaces, since by assumption S is a compact, maximally connected, smooth hypersurface and  $\nabla u$  is smooth on S. Then, by the chain rule and (3.12), we have the differential equation

$$\frac{du(\gamma_t)}{dt} = |\nabla_S u|^2(\gamma_t) = \lambda u(\gamma_t)^2, \qquad (3.13)$$

which is solved, once imposed the initial condition  $u_0 = u(\gamma_0)$ , by

$$u(\gamma_t) = \frac{1}{1/u_0 - \lambda t}.$$

But this would lead to the blow up of u for  $t \to 1/(\lambda u_0)$ , which contradicts the boundedness of u on S.

If the Robin condition holds, with the same reasoning we can obtain (3.13) only with  $\lambda$  substituted by  $h - \alpha^2$ . By the assumption on the mean curvature, we have that  $h - \alpha^2 > 0$ , and therefore  $du(\gamma_t)/dt \ge cu(\gamma_t)^2$  for some positive constant c. Thus u would need to blow up in finite time, which is impossible.

The proof that (3.11) is not constant zero on S employs similar considerations. Suppose that (3.11) is zero on S. In the case the Dirichlet condition holds, this implies that  $(\partial u/\nu)^2 = (\partial v/\nu_0)^2$ . Then Theorem 3.2 implies that  $u = \pm v$  in E. This contradicts the orthogonality assumption of u and v.

In the case the Neumann condition holds, the assumption that (3.11) is zero implies that

$$|\nabla_S u|^2 - |\nabla_S v|^2 = \lambda (u^2 - v^2) \text{ on } S.$$
 (3.14)

By Theorem 3.2 and continuity of u, v, there exists a point  $x_0 \in S$  such that  $u(x_0)$ and  $v(x_0)$  are different and non-zero. By eventually switching them or changing the sign of u, v, we can suppose that  $u(x_0) > v(x_0) > 0$ . Let  $k = u(x_0) + v(x_0) > 0$  and let  $\gamma : (-\infty, \infty) \to S$  be a solution of

$$\begin{cases} \gamma_0 = x_0, \\ \dot{\gamma}_t = (\nabla_S u + \nabla_S v)(\gamma_t) \end{cases}$$

 $\mathbf{37}$ 

Let  $f_t = u(\gamma_t), g_t = v(\gamma_t)$ . Then, by the chain rule, it holds

$$\frac{d(f_t + g_t)}{dt} = |\nabla_S u + \nabla_S v|^2(\gamma_t) \ge 0, \qquad (3.15)$$

so  $f_t + g_t > k$  for every t > 0. Then

$$\frac{d(f_t - g_t)}{dt} = (|\nabla_S u|^2 - |\nabla_S v|^2)(\gamma_t) = \lambda(f_t - g_t)(f_t + g_t) \ge \lambda k(f_t - g_t), \quad (3.16)$$

which, since  $f_0 - g_0 > 0$ , would lead to  $f_t - g_t \to \infty$  as  $t \to \infty$ , which contradicts the boundedness of u and v.

If the Robin condition holds, then by reasoning as before, we obtain the same chain of equalities as in (3.16) only with  $\lambda$  substituted by  $h - \alpha^2$ , that is

$$\frac{d(f_t - g_t)}{dt} = (h - \alpha^2)(f_t - g_t)(f_t + g_t).$$

Since  $f_t + g_t > k \ge 0$  for t > 0 by (3.15), and the mean curvature bound assumption  $-\alpha H < \alpha^2 + \lambda$  implies that  $h - \alpha^2 > 0$ , then there exists a positive constant c such that

$$\frac{d(f_t - g_t)}{dt} \ge c(f_t - g_t),$$

which, since  $f_0 - g_0 > 0$ , would lead again to the unboundedness of  $f_t - g_t = u(\gamma_t) - v(\gamma_t)$  as  $t \to \infty$ , which is impossible.

# 3.3 Splitting by a perturbation of the boundary of a hole

In this section, we cut a circular hole inside  $\Omega$  and deform its boundary to split the spectrum of the perturbed domain. We start by constructing a deformation of  $\Omega$  minus a part which splits a single eigenvalue as follows.

**Proposition 3.7.** Let G be a smooth domain whose closure is contained in  $\Omega$ . Let  $\lambda$  be an eigenvalue of  $\Omega \setminus \overline{G}$  of multiplicity m. In the case of Robin conditions suppose that H, the mean curvature of  $\partial G$ , satisfies  $\alpha H < \alpha^2 + \lambda$ . Then for any open set U containing  $\overline{G}$ , and any choice of different indices i, j among  $1, \ldots, m$ , there exists a deformation  $(\phi_t)_t$  localized in U such that (with the notation of Theorem 3.3 with  $\Omega$  replaced by  $\Omega \setminus \overline{G}$ ) the eigenvalues  $\lambda_t^1, \ldots, \lambda_t^m$  of  $\phi_t(\Omega \setminus \overline{G})$  are such that  $\lambda_t^i \neq \lambda_t^j$  for all  $t \in (0, t_0)$ .

*Proof.* Let h be as in (3.9) and let  $S = \partial G$ . Let  $u_t^1, \ldots, u_t^m$  be the eigenfunctions associated to the eigenvalues  $\lambda_t^1, \ldots, \lambda_t^m$  as in Theorem 3.3. By Lemma 3.6, there exists y on S such that

$$(|\nabla u_0^i|^2 - h(u_0^i)^2)(y) \neq (|\nabla u_0^j|^2 - h(u_0^j)^2)(y).$$
(3.17)

Then, by choosing a deformation  $(\phi_t)_t$  which is the identity outside an appropriately small neighborhood of y, we have

$$\int_{S} (|\nabla u_0^i|^2 - h(u_0^i)^2) \,\nu \cdot \dot{\phi}_0 \neq \int_{S} (|\nabla u_0^j|^2 - h(u_0^j)^2) \,\nu \cdot \dot{\phi}_0. \tag{3.18}$$

Such a deformation can be constructed in many ways (for the sake of completeness, we give an explicit example hereafter). By Lemma 3.5, (3.18) implies that  $\dot{\lambda}_0^i \neq \dot{\lambda}_0^j$ . Since  $\lambda_t^i$  and  $\lambda_t^j$  are both analytic in t, we conclude that there exists  $t_0$  such that  $\lambda_t^i \neq \lambda_t^j$  for  $t \in (0, t_0)$ .

We show how to build a deformation such that (3.18) holds. Let  $\psi$  be a diffeomorphism so that

$$\psi(B_1 \cap \{x_N < 0\}) = V \cap \Omega,$$

where  $B_1$  is the unit ball and V a small neighborhood of y. Let  $\hat{z}$  indicate  $(z_1, \ldots, z_{N-1})$ and let

$$\rho_c(\hat{z}) = \begin{cases} \frac{c^3}{8} \exp\left(\frac{1}{|\hat{z}/c|^2 - 1}\right) & \text{if } |\hat{z}| < c, \\ 0 & \text{otherwise.} \end{cases}$$

Notice that by construction  $|\rho_c|_{C^2(\mathbb{R}^{N-1})} \leq c$  for any  $c \leq 1$ . Let  $\varphi_t(z)$  be the extension of the map  $z \mapsto (\hat{z}, t\rho_c(\hat{z}))$  from  $\{x_N = 0\}$  to a diffeomorphism which is the identity outside  $B_1$  and such that  $|\varphi_t - I|_{C^2(\mathbb{R}^N)} \leq ct$ . Let  $\phi_t = \psi \circ \varphi_t \circ \psi^{-1}$ . Then by taking  $c \leq |\psi|_{C^2} |\psi^{-1}|_{C^2}$ , we have

$$|\phi_t - I|_{C^2} \le t. \tag{3.19}$$

By eventually considering a smaller c, the continuity of u on S together with (3.17) imply that (3.18) holds.

Although Proposition 3.7 shows how to split an eigenvalue, the perturbation chosen might cause two other eigenvalues to overlap, creating a new repeated one. To avoid this potential issue, we need a finer control on the behavior of the whole spectrum. This is what is achieved in the following lemma. **Lemma 3.8.** Let G be a smooth domain whose closure is contained in  $\Omega$ . We indicate as  $\lambda_1 \leq \lambda_2 \leq \ldots$  the eigenvalues of  $\Omega \setminus \overline{G}$ , and as r the first index such that  $\lambda_r$  has multiplicity  $m \geq 2$ . In the case of Robin conditions, suppose that H, the mean curvature of G, satisfies  $\alpha H < \alpha^2 + \lambda$ . Then, for any open set U which contains G and any  $\varepsilon > 0$ , there exists  $(\phi_t)_t$  a deformation localized in U such that, indicating as  $\lambda_1^t \leq \lambda_2^t \leq \ldots$  the eigenvalues of  $\phi_t(\Omega \setminus \overline{G})$ , for all  $t \leq t_0$  we have that

(i) for all  $i \leq r + m + 1$  it holds

$$|\lambda_i^t - \lambda_i| \le \varepsilon d_r,$$

where  $d_r$  is the minimum positive number of  $\{\lambda_{j+1} - \lambda_j : j = 1, ..., r + m\};$ 

(ii) the multiplicity of  $\lambda_r^t$  is strictly smaller than the multiplicity of  $\lambda_r$ ;

(iii) for all i > r + m, it holds  $\lambda_i^t > \lambda_r$ .

Proof. Let  $(\phi_t)_t$  be a deformation as given by Proposition 3.7. Let  $u_1^t, u_2^t, \ldots$  be the eigenfunctions of  $\Omega \setminus \overline{G}$  associated respectively to  $\lambda_1^t, \lambda_2^t, \ldots$ . By Theorem 3.3, we can assume that  $\lambda_i^t, u_i^t$  are analytic in t, that  $\lambda_i^0 = \lambda_i$ , and that  $u_r^0, \ldots, u_{r+m}^0$  is an orthonormal basis for the eigenspace of  $\lambda_r$ . Then, there are two distinct indices i and j among  $\{r, \ldots, r+m\}$ , such that for  $t_0$  small enough,

$$\lambda_i^t \neq \lambda_i^t, \quad \text{for } t \in (0, t_0]. \tag{3.20}$$

By the eigenvalue stability estimate of Lemma 3.4, there is a  $t_0$  small enough such that

$$|\lambda_i^t - \lambda_i| \le \varepsilon d_r, \quad \forall t \le t_0, \forall i \in \{1, \dots, r+m+1\}.$$
(3.21)

Let C, C' indicate two constants which depend only on the dimension N, the Lipschitz constant of  $\partial(\Omega \setminus \overline{G})$  and the area of  $\Omega$ . By Weyl's asymptotic law,  $\lambda_n = Cn^{2/N} + o(n^{2/N})$  for any n. Then, from the uniform estimate of Lemma 3.4, it follows that, for i > r + m,

$$\lambda_i^t - \lambda_r \ge (\lambda_i^t - \lambda_i) + \lambda_i - \lambda_r \ge C'(-Cci^{2/N} + i^{2/N} - r^{2/N}),$$

where c > 0 is a bound on the deformation magnitude which we can make arbitrarily small by taking a smaller  $t_0$ . Therefore, for  $t_0$  and c small enough,

$$\lambda_i^t - \lambda_r > 0, \quad \forall t \le t_0, \forall i > r + m.$$
(3.22)

In conclusion, Items i-ii-iii are consequences of (3.21)-(3.20)-(3.22).

The construction in the previous proof gives us a method to split the first nonsimple eigenvalue without altering the simplicity of smaller eigenvalues. In fact, by taking  $\varepsilon < 1/2$ , from Items i and iii of Lemma 3.8, we have that the eigenvalues  $\lambda_i^t$ perturbed from  $\lambda_i$ :

- lie in disjoint neighborhoods of  $\lambda_i$ , for i < r;
- are not further than  $d_r/2$  from  $\lambda_i$ , for  $r \leq i \leq r + m$ ;
- are larger than  $\lambda_r$ , for i > r + m.

We can iterate this procedure to split the whole spectrum as in the following proof. Some additional care must be taken in the case of Robin boundary conditions.

Proof of the first statement of Theorem 3.1. Let  $G_0$  be a ball of radius R contained in U. We consider the following recursive construction: let  $G_n = \phi_{t_n}^n(G_{n-1})$  where the deformation  $\phi_t^n$  is the one obtained by Lemma 3.8 with  $G, \varepsilon$  substituted respectively by  $G_{n-1}, \varepsilon_n$ . By choosing  $\varepsilon_n$  small enough at each step, we have that  $\partial G_n \to \partial E$  in C<sup>2</sup>-norm, where E is a smooth domain contained in U. In the case of Robin boundary conditions, notice that the mean curvature  $H_0$  of  $G_0$  is -(N-1)/R and satisfies  $\alpha H_0 < \alpha^2 + \lambda$  for any eigenvalue  $\lambda > 0$ , since we assumed  $\alpha \ge 0$ . Moreover, by eventually choosing a smaller  $\varepsilon_n$  for each n, thanks to (3.19) we can assume the mean curvature of  $G_n$  is still negative, so that the assumptions of Lemma 3.8 hold at every step for any boundary condition. Let  $\tilde{\Omega} = \Omega \setminus \overline{E}$ .

Let  $r_n$  be the index of the first non-simple eigenvalue of  $\Omega \setminus \overline{G}_n$ . By Items i and iii of Lemma 3.8 we have that all eigenvalues with index smaller than  $r_n$  are simple for every n. Moreover,  $(r_n)_n$  is a non-decreasing sequence of integers which can not be definitely constant; in fact by Item ii of Lemma 3.8,  $r_{n+j}$  can be equal to  $r_n$  for at most  $j \in \{1, \ldots, r_n\}$ . Therefore,  $r_n \to \infty$  as  $n \to \infty$ , and thus  $\tilde{\Omega}$  can have only simple eigenvalues.

We remark that our previous construction involves cutting out a part of  $\Omega$  which causes  $\tilde{\Omega}$  to be not homeomorphic to  $\Omega$ . However, this can be avoided by deforming appropriately the boundary of  $\Omega$ , instead of the boundary of the hole, as shown in the following section.

# **3.4** Splitting by a perturbation of the boundary

In this section, we show how to deform directly a part of the boundary of  $\Omega$  to split its spectrum.

Let U be an arbitrary open set whose intersection with  $\partial\Omega$  is non-empty. By eventually applying a bi-Lipschitz transformation which modifies  $\Omega$  only in U, we can assume, and will do so, that there is an open set  $V \subseteq U$  such that  $\Sigma = V \cap \partial\Omega$ is contained in a hyperplane.

We recall now a result proved in [81, Lemma 6 and calculations leading to (20)], which relies on the explicit computation of the Fréchet derivative of the bilinear form  $Q_{\psi(\Omega)}$  with respect to  $\psi$  and on properties of Fredholm operators.

**Lemma 3.9.** Let  $(\phi_t)_t$  be a deformation localized in V. Under the same notation of Theorem 3.3, if for every  $t \leq t_0$  we have that  $\lambda_t^1 = \cdots = \lambda_t^m$ , then the matrix with element

$$\int_{\Sigma} (\nabla u_0^k \cdot \nabla u_0^l) \phi_t \cdot \nu - \alpha f u_0^k u_0^l \nu \cdot \nabla \phi_t \nu$$
(3.23)

in position (k, l), where f is a positive continuous function, is a multiple of the identity matrix for every  $t \leq t_0$ .

With the aid of the previous lemma, we can construct a deformation which splits a single eigenvalue as follows.

**Proposition 3.10.** Let  $\lambda$  be an eigenvalue of  $\Omega$  of multiplicity  $m \geq 2$ . Then, with the notation of Theorem 3.3, there are two indices i, j and there exists a deformation  $(\phi_t)_t$  localized in V such that  $\lambda_{\varepsilon}^i \neq \lambda_{\varepsilon}^j$  for a certain  $\varepsilon < t_0$ .

Proof. In the case of Dirichlet conditions, the construction is the same as the one considered in Section 3.3, only with the deformation applied directly to the boundary of  $\partial\Omega$ . In fact, in this case the existence of a point  $y \in \Sigma$  such that (3.17) holds is guaranteed directly by Theorem 3.2. Then the claim in the conclusion can be obtained by constructing a deformation in a small neighborhood of y exactly as in Proposition 3.7.

In the case of Neumann conditions, if there exists a point  $y \in \Sigma$  such that (3.17) holds, we can again build a local deformation of the boundary which fulfils our thesis. If instead

$$|\nabla u_0^k|^2 - \lambda (u_0^k)^2 = |\nabla u_0^l|^2 - \lambda (u_0^l)^2$$
(3.24)

on  $\Sigma$  for every k, l, we show that we can find anyway indices i, j and a deformation  $(\phi_t)_t$  such that  $\lambda_{\varepsilon}^i \neq \lambda_{\varepsilon}^j$  for some  $\varepsilon$ . In fact, if we suppose by contradiction that this is not the case, then by taking  $\nu \cdot \nabla \phi_t \nu = 0$  and by the arbitrariness of  $\phi_t \cdot \nu$  in (3.23), Lemma 3.9 would lead to  $|\nabla u_k|^2 = |\nabla u_l|^2$  on  $\Sigma$  for every k, l. If  $\lambda = 0$  then it has already multiplicity 1. If  $\lambda \neq 0$  then (3.24) would become  $(u_0^k)^2 = (u_0^l)^2$  on  $\Sigma$ , which by Theorem 3.2 would lead to  $u_0^k = \pm u_0^l$  in  $\Omega$ , which is a contradiction.

The case of Robin boundary conditions can be proven from Lemma 3.9 alone. If there was no deformation  $(\phi_t)_t$  localized in V which splits the eigenvalue  $\lambda$ , by choosing  $\phi_t \cdot \nu = 0$  and  $\nu \cdot \nabla \phi_t \nu$  arbitrarily, making use of Lemma 3.9 we would have

$$u_0^k u_0^l = 0$$
 for  $k \neq l$ ,  $(u_0^k)^2 = (u_0^l)^2$  for every  $k, l$ ,

which would lead to  $u_0^k = -\alpha(\partial u_0^k/\partial \nu) = 0$  on  $\Sigma$ , which is prohibited by Theorem 3.2.

Finally we can easily adapt the arguments from Section 3.3 to construct a localized boundary perturbation which splits the spectrum.

Proof of the second statement of Theorem 3.1. Let  $(B_n)_n$  be a sequence of mutually disjoint balls contained in V such that  $B_n \cap \partial \Omega$  is non-empty for every n. By Proposition 3.10 and by the same reasoning as in the proof of Lemma 3.8, we can build a deformation  $\phi_t^1$  localized in  $B_1$  such that for a certain  $t = t_1$ , Items i-ii-iii of Lemma 3.8 hold. Letting  $\Omega_1 = \phi_{t_1}^1(\Omega)$ , we can repeat again the same construction to build a deformation  $\phi_t^2$  localized in  $B_2$  and obtain a new domain  $\Omega_2 = \phi_{t_2}^2(\Omega_1)$ . Iterating this construction, we obtain a sequence of domains  $\Omega_n$  which coincide with  $\Omega$  outside V and whose boundaries converge in C<sup>1</sup>-norm inside V. The limit domain is thus bi-Lipschitz equivalent to  $\Omega$  and, by the same concluding argument of the proof of the first statement of Theorem 3.1, has a simple spectrum.

# Chapter 4

# Extending asymptotic formulae from simple to repeated eigenvalues

Many asymptotic formulae for the shift of eigenvalues of elliptic differential operators caused by small singular perturbations have been obtained in the case of multiplicity one (see for instance [77, Chapter 9], [14] and references therein, [86], [87], [47]). The generalization of such expressions to higher multiplicities often requires nontrivial calculations and is restricted to specific cases (see for example [76], [22], [57, Theorem 2.5.8]). Moreover, such an effort is usually considered unnecessary due to genericity results of simple eigenvalues (see [79], [80], [96], [95]). Nonetheless, higher multiplicities appear in many natural situations; for instance the Laplacian has nonsimple spectrum whenever the domain presents some symmetries. In this chapter, we derive a new tool to study the behaviour and properties of repeated eigenvalues for general types of perturbations. Namely, the main result is a variational characterization (Theorem 4.7) which allows the direct extension of asymptotic formulae which are valid for simple eigenvalues to non-simple ones. We then apply this result to study different domain perturbations of interest in applications.

The chapter is organized as follows. In Section 4.1.1, we recall some results regarding the spectral stability and generic simplicity for second-order elliptic operators. In Section 4.2, we derive the main variational characterization by a technique which involves a double perturbation at asymptotically different speeds. Finally in Section 4.3, we consider some domain perturbations of particular interest in applications: grounded inclusions, conductivity inclusions, and boundary deformations. We show how the general variational asymptotic formula applies in each of these cases and derive new interesting properties for the Dirichlet and Neumann Laplacian eigenvalues. We refer to Section 4.3.4 for a more specific summary of the results obtained for these two cases.

## 4.1 Preliminaries

We briefly recall three types of convergence of domains. We say that  $\Omega_{\varepsilon}$  converges to  $\Omega$  as  $\varepsilon \to 0$  in:

• Hausdorff distance, if  $d_H(\Omega_{\varepsilon}, \Omega) \to 0$ , where

$$d_H(A, B) = \max\{ \sup_{a \in A} \inf_{b \in B} |a - b|, \sup_{b \in B} \inf_{a \in A} |a - b| \};$$

- measure, if  $|\Omega_{\varepsilon} \Delta \Omega| \to 0$ , where  $\Delta$  indicates symmetric difference and  $|\cdot|$  the Lebesgue measure;
- C<sup>k</sup>-topology, if there is a family  $(\phi_{\varepsilon})_{\varepsilon}$  of functions in  $C^k(\mathbb{R}^d, \mathbb{R}^d)$  such that

$$\Omega + \phi_{\varepsilon}(\Omega) = \Omega_{\varepsilon} \quad \text{and} \quad |\phi_{\varepsilon}|_{C^{k}} \xrightarrow{\varepsilon \to 0} 0.$$
(4.1)

We collect hereafter some relationships between these types of convergence which are relevant to us.  $C^k$ -convergence implies Hausdorff convergence, since if (4.1) holds then  $d_H(\Omega_{\varepsilon}, \Omega) \leq |\phi_{\varepsilon}(\Omega)|_{C^0} \xrightarrow{\varepsilon \to 0} 0$ . If the sets considered have Lipschitz and connected boundaries, Hausdorff convergence implies measure convergence, since in this case  $|\Omega_{\varepsilon} \Delta \Omega| \leq d_H(\Omega_{\varepsilon}, \Omega) \min \{|\Omega_{\varepsilon}|, |\Omega|\}$ . Although the converse result fails in general, if the converging sequence's boundaries have uniformly bounded Lipschitz constants, then, by a local patching argument, it can be easily shown that also measure convergence implies Hausdorff convergence. Therefore, although we will state all our results referring to Hausdorff convergence, they can be adapted to other types of convergence whenever the domains under consideration have enough regularity.

# 4.1.1 Stability and simplicity of the spectrum of secondorder elliptic operators

Let  $L(\Omega)$  be a second-order, self-adjoint, elliptic differential operator defined on  $\Omega$ and let *a* be its associated coercive, continuous, and bounded bilinear form. We say that  $\lambda$  is an eigenvalue of  $L(\Omega)$  with associated eigenfunction  $u \in V(\Omega)$  if

$$a(u,v) = \lambda \int_{\Omega} uv, \qquad \forall v \in V$$

where  $V(\Omega) = H_0^1(\Omega)$  in the case of homogeneous Dirichlet boundary conditions, or  $V(\Omega) = H^1(\Omega)/\mathbb{R}$  in the case of homogeneous Neumann boundary conditions. From standard results in spectral theory, we know that the eigenvalues of an elliptic operator defined on a Lipschitz domain have finite multiplicity and can be arranged in a non-decreasing sequence. We also assume that the associated eigenfunctions are orthonormal in  $L^2(\Omega)$ . With these conventions, we can state the following stability result.

**Theorem 4.1.** For every  $\varepsilon > 0$ , let L be an elliptic operator and let  $\rho_{\varepsilon}$  be a smooth function with support contained in  $\Omega$  such that its C<sup>2</sup>-norm is smaller than  $\varepsilon$ . For any  $n \in \mathbb{N}$ , let  $\lambda$  be the n-th eigenvalue of L and  $\lambda_{\varepsilon}$  be the n-th eigenvalue of  $L + \rho_{\varepsilon}$ , and let  $u_{\varepsilon}$  be an associated eigenfunction. Then there exists a constant C such that for every  $\varepsilon$  small enough

$$|\lambda_{\varepsilon} - \lambda| \le C\varepsilon,$$

and there exists u, an eigenfunction of  $\lambda$ , such that

$$|u_{\varepsilon} - u|_{\mathrm{H}^{1}(\Omega)} \leq C\varepsilon.$$

The proof is a consequence of standard results in perturbation theory (see for instance [90, Theorem 1 at p.57]). We are interested not only in stability with respect to smooth perturbations of coefficients but also in singular perturbations of domains. In particular, we will need the following result.

**Theorem 4.2.** Let  $\Omega$  and  $\Omega_{\varepsilon}$  be two Lipschitz domains whose Hausdorff distance one from another is  $\varepsilon$ . Let  $L(\Omega)$  and  $L(\Omega_{\varepsilon})$  be second-order elliptic operators with homogeneous Dirichlet or Neumann boundary conditions imposed on  $\partial\Omega$  and  $\partial\Omega_{\varepsilon}$ . Let M be an upper bound to the Lipschitz constants of  $\partial\Omega$  and  $\partial\Omega_{\varepsilon}$ , and to the coercivity, continuity and boundedness constants of a. For any  $n \in \mathbb{N}$ , let  $\lambda$  be the *n*-th eigenvalue of  $L(\Omega)$  and  $\lambda_{\varepsilon}$  be the *n*-th eigenvalue of  $L(\Omega_{\varepsilon})$ . Then there exists a constant C which depends only on M, n, and the dimension d such that

$$|\lambda_{\varepsilon} - \lambda| \le C\varepsilon. \tag{4.2}$$

A proof of this result can be obtained as a consequence of the theory of transition operators and its applications (see [33] for a survey of this technique). In what follows, we outline another approach which relies on stability results for boundary value problems.

Outline of the proof. We adapt the argument in [69, Section 4.4] to our case. Let E be an arbitrary Lipschitz domain containing  $\Omega_{\varepsilon} \cup \Omega$  and consider  $u_{\varepsilon}$  extended to the whole E. Let u be the orthogonal projection of  $u_{\varepsilon}$  from V(E) onto  $V(\Omega)$ . Let  $\overline{u}$  be the unique solution in  $V(\Omega)$  of

$$a(\overline{u}, v) = \lambda \int uv, \qquad \forall v \in V(\Omega).$$

From [93, Inequality (3.2)], we have that

$$|\overline{u} - u|_{\mathrm{H}^{1}(E)} \leq C |\lambda u|_{\mathrm{L}^{2}(E)} |\lambda u|_{V(E)'} d_{H}(\Omega, \Omega_{\varepsilon}),$$

where C is a constant which depends only on the Lipschitz constant of  $\Omega$ , and the constants involved in the continuity, boundedness and coercivity assumptions on a. Since  $|u|_{V(E)'} \leq |u|_{L^2(E)}$ , and by Weyl's law, there is a constant  $\tilde{C}$  which depends only on the area of  $\Omega$  and the dimension d such that  $\lambda \leq \tilde{C}n^{d/2}$ . Then, it follows that

$$|\overline{u} - u|_{\mathrm{H}^{1}(E)} \leq C\tilde{C}n^{2/d}|u|_{\mathrm{L}^{2}(E)}^{2}d_{H}(\Omega,\Omega_{\varepsilon}).$$

Hence, by (4.31) and Lemma 14 of [69], we obtain estimate (4.2).

The following result concerning stability of eigenfunctions is a particular case of [43, Theorem 1.2 and subsequent discussion].

**Theorem 4.3.** Let  $\Omega_{\varepsilon}$  be a family of Lipschitz domains converging to  $\Omega$  in Hausdorff distance as  $\varepsilon \to 0$ . Suppose that at least one of the following hypotheses holds:

- $\Omega_{\varepsilon} \subseteq \Omega$  for every  $\varepsilon$ ;
- the Lipschitz constants of  $\partial \Omega_{\varepsilon}$  are uniformly bounded.

Let  $\lambda$  and  $\lambda_{\varepsilon}$  be as in the hypothesis of Theorem 4.2 and let  $u_{\varepsilon,1}, \ldots, u_{\varepsilon,m}$  be an orthonormal basis of the eigenspace of  $\lambda_{\varepsilon}$ . Then there exists  $u_1, \ldots, u_m$  an orthonormal basis of  $\lambda$  such that as  $\varepsilon \to 0$  it holds

$$u_{\varepsilon,j} \to u_j \text{ in } \mathrm{H}^1(\Omega), \qquad \forall j \in \{1, \dots, m\}.$$

We move on to the issue of genericity of simple eigenvalues.

**Theorem 4.4.** For every  $\varepsilon > 0$  and every open set  $U \subseteq \Omega$  there exists a smooth function  $\rho$  with support contained in U such that its C<sup>2</sup>-norm is smaller than  $\varepsilon$  and  $L + \rho$  has simple spectrum.

This result is an immediate consequence of [96, Theorem 7]. We remark that in Chapter 3 we proved genericity of simple eigenvalues by domain deformation, instead of coefficient perturbations. The results which follow could indeed be proved using only domain deformations, but the arguments would require some more attention due to the variation of the underlying domain.

### 4.2 Variational asymptotic formula

Recall that the eigenvalues and eigenfunctions of L can be characterized respectively as minima and minimizers of a quadratic functional F. More explicitly, if we indicate as  $\lambda_1 \leq \lambda_2 \leq \ldots$  the eigenvalues of L and  $u_1, u_2, \ldots$  some associated orthonormal eigenfunctions, we have that for  $Fu = \langle Lu, u \rangle_V$  it holds

$$u_n \in \operatorname*{argim}_{u \in U_n} Fu, \qquad \lambda_n = Fu_n,$$

$$(4.3)$$

where  $U_n = \{u \in V : u \perp \{u_1, \ldots, u_{n-1}\}, |u|_V = 1\}$ . However, there is an ambiguity in the choice of eigenfunctions which is particularly relevant to us: if  $\lambda_n = \cdots = \lambda_{n+m}$ , any choice of orthonormal eigenfunctions in the linear space spanned by  $\{u_n, \ldots, u_{n+m}\}$  is still a basis of the eigenspace. Our variational characterization will select the right basis for the problem considered up to a predetermined asymptotic error. More precisely, we make the following assumptions.

Assumption 4.5. We suppose that there is a family of self-adjoint, elliptic, secondorder differential operators  $L_{\varepsilon}$  which converges spectrally to L as  $\varepsilon \to 0$ , that is the *n*-th eigenvalue of  $L_{\varepsilon}$  and any of its associated eigenfunctions converge respectively to the *n*-th eigenvalue of L and to a function in its eigenspace. Moreover, we suppose that there exists an open set on which the differential operators L and  $L_{\varepsilon}$  coincide for every  $\varepsilon$ .

Assumption 4.6. We suppose that we already know an asymptotic expansion for simple eigenvalues. In particular, we assume that if  $\lambda$  is a simple eigenvalue of Lwith associated eigenfunction u, and if  $\lambda_{\varepsilon}$  is an eigenvalue of  $L_{\varepsilon}$  which converges to  $\lambda$  as  $\varepsilon \to 0$ , then

$$\lambda_{\varepsilon} - \lambda = f(\varepsilon, u) + r(\varepsilon),$$

where f is a known function and by  $r(\varepsilon)$  we indicate a function which is of order  $o(f(\varepsilon, u))$  as  $\varepsilon \to 0$ . We also require f to be continuous.

Under these assumptions, we can derive the following result.

**Theorem 4.7** (Variational characterization). Let  $\lambda$  be an eigenvalue of L of multiplicity m, and let  $\lambda_{\varepsilon,1} \leq \cdots \leq \lambda_{\varepsilon,m}$  be the eigenvalues of  $L_{\varepsilon}$  which converge to  $\lambda$  as  $\varepsilon \to 0$ . Then, for any  $j \in \{1, \ldots, m\}$ , it holds

$$\lambda_{\varepsilon,j} - \lambda = f(\varepsilon, v_j) + r(\varepsilon),$$

where

$$v_{j} \in \underset{v \in U_{\lambda,j}}{\operatorname{argim}} f(\varepsilon, v), \qquad U_{\lambda,j} = \left\{ v \in V : Lv = \lambda v, \ v \perp \{v_{1}, \dots, v_{j-1}\}, \ |v|_{V} = 1 \right\}.$$
(4.4)

Proof. Step 1. By Assumption 4.5, there exists an open ball  $B \subseteq \mathbb{R}^d$  on which  $L \equiv L_{\varepsilon}$ . For every  $\delta > 0$ , by Theorem 4.4 there exists  $\rho_{\delta} \in C_c^{\infty}(B)$  such that  $|\rho_{\delta}|_{C^2} < \delta$  and  $L + \rho_{\delta}$  has simple spectrum. Let  $\lambda_{\delta,1} < \cdots < \lambda_{\delta,m}$  be the eigenvalues of  $L + \rho_{\delta}$  which converge to  $\lambda$  as  $\delta \to 0$ , and let  $v_{\delta,1}, \ldots, v_{\delta,m}$  be the associated eigenfunctions. Let  $\lambda_{\varepsilon,\delta,j}$  be the eigenvalues of  $L_{\varepsilon} + \rho_{\delta}$  which converge to  $\lambda_{\delta,j}$  as  $\varepsilon \to 0$ . By standard elliptic estimates (Theorem 4.1), we have that there is a constant C such that  $|\lambda_{\delta,j} - \lambda| \leq C\delta$ ,  $|\lambda_{\varepsilon,\delta,j} - \lambda_{\varepsilon,j}| \leq C\delta$  and  $v_{\delta,j} \to v_j$  as  $\delta \to 0$ , where  $v_j$  is a certain function in the eigenspace of  $\lambda$ . For every  $\varepsilon > 0$ , let us now choose the parameter  $\delta$  as a function of  $\varepsilon$ , small enough so that  $\delta(\varepsilon) = r(\varepsilon)$ . Then, by the continuity of f, we have

$$\lambda_{\varepsilon,j} - \lambda = (\lambda_{\varepsilon,j} - \lambda_{\varepsilon,\delta(\varepsilon),j}) + (\lambda_{\varepsilon,\delta(\varepsilon),j} - \lambda_{\delta(\varepsilon),j}) + (\lambda_{\delta(\varepsilon),j} - \lambda) = f(\varepsilon, v_j) + r(\varepsilon).$$
(4.5)

Step 2. To retrieve the variational characterization of  $v_j$ , recall that we assumed the ordering  $\lambda_{\varepsilon,1} \leq \cdots \leq \lambda_{\varepsilon,m}$  for every  $\varepsilon$ . Therefore, for  $\lambda_{\varepsilon,1}$  to be the smallest eigenvalue,  $v_1$  must be a minimizer of the right hand side of (4.5) among all eigenfunctions of  $\lambda$  of norm 1. Notice that if  $v_{\varepsilon,j} \to v_j$  for every j, then  $v_1, \ldots, v_m$  must be linearly independent (since by assumption the multiplicity of  $\lambda$  is m). Therefore,  $v_2$  must be the minimizer of the right hand side of (4.5) among all eigenfunctions orthonormal to  $v_1$ . Repeating the same reasoning for  $\lambda_{\varepsilon,3}, \ldots, \lambda_{\varepsilon,m}$ , we have that (4.4) must hold for every j.

#### 4.2.1 Asymptotic formulae involving a bilinear function

It is useful to consider more carefully the case where the expression in the asymptotic formula admits a separation of variables as

$$f(\varepsilon, u) = E(\varepsilon)b(u, u), \tag{4.6}$$

with  $b: V \times V \to \mathbb{R}$  being a symmetric bilinear form. This happens for many useful types of domain perturbations (see Section 4.3). The advantage of this case is twofold: the minimizer of (4.4) is unique and can be easily computed as follows. Choosing  $u_1, \ldots, u_m$  an arbitrary orthonormal basis of the eigenspace of  $\lambda$ , condition (4.4) can be rewritten as

$$v_n = E(\varepsilon) \operatorname{argim} \left\{ a \cdot Ba : a \in \mathbb{R}^m, \ |a| = 1, \ a_1 u_1 + \dots + a_m u_m \perp \{v_1, \dots, v_{n-1}\} \right\},$$
  
(4.7)

where B is a symmetric matrix with elements

$$B_i^j = b(u_i, u_j). \tag{4.8}$$

Then, by diagonalizing B, we obtain the following result.

Corollary 4.8. If f can be rewritten as in (4.6), then the minimum and minimizer of (4.4) are respectively the *n*-th eigenvalue of B and  $a_{1,n}u_1 + \cdots + a_{m,n}u_m$ , where  $(a_{1,n}, \ldots, a_{m,n})$  is the normalized eigenfunction of B associated to its *n*-th eigenvalue.

Notice that if the bilinear form b has the further decomposition

$$b(u_i, u_j) = l(u_i)l(u_j) \quad \forall i, j,$$

$$(4.9)$$

where  $l: V \to \mathbb{R}$  is linear, an easy computation shows that the first m-1 eigenvalues of B are zero and the m-th one is  $l(u_1)^2 + \cdots + l(u_m)^2$ . Therefore, we have the following result. Corollary 4.9. If the bilinear form b can be rewritten as in (4.9), then

$$\lambda_{\varepsilon,n} - \lambda = \begin{cases} E(\varepsilon) \sum_{n=1}^{m} l(u_n)^2 + r(\varepsilon) & \text{if } n = m, \\ r(\varepsilon) & \text{if } n < m. \end{cases}$$

Moreover, the eigenfunction in the eigenspace of  $\lambda$  to which the eigenfunction of  $\lambda_{\varepsilon,m}$  converges is given by

$$\frac{\sum_{n=1}^{m} l(u_n)u_n}{\sqrt{\sum_{n=1}^{m} l(u_n)^2}}$$

### 4.3 Applications to eigenvalues of the Laplacian

In this section we consider some domain perturbations of particular interest in applications, and derive explicit asymptotic formulae for repeated eigenvalues of the Laplacian. Different numerical experiments are provided to validate these formulae.

#### 4.3.1 Perturbation by a grounded inclusion

Let *B* be a Lipschitz domain in  $\mathbb{R}^d$  with connected boundary, with volume  $|B| = |\Omega|$ , and centered at the origin in the sense that  $\int_{\partial B} y_1 d\sigma(y_1, y_2) = \int_{\partial B} y_2 d\sigma(y_1, y_2) = 0$ . Fix a point  $z \in \Omega$  and consider a scaling coefficient  $\varepsilon > 0$ . Suppose then that the domain  $\Omega$  is perturbed into  $\Omega_{\varepsilon} = \Omega \setminus D$ , by inserting an inclusion  $D = z + \varepsilon B$  and requiring homogeneous Dirichlet conditions to hold on  $\partial D$ .

Let  $\lambda$  be an eigenvalue of  $\Omega$  with associated eigenfunction u. Then  $\lambda_{\varepsilon}$  is an eigenvalue of  $\Omega_{\varepsilon}$  perturbed from  $\lambda$  with associated eigenfunction  $u_{\varepsilon}$  if  $\lambda_{\varepsilon} \to \lambda$  as  $\varepsilon \to 0$  and

$$\begin{cases} \Delta u + \lambda u = 0 & \text{in } \Omega, \\ u = 0 \text{ or } \partial u / \partial \nu = 0 & \text{on } \partial \Omega, \end{cases} \qquad \begin{cases} \Delta u_{\varepsilon} + \lambda_{\varepsilon} u_{\varepsilon} = 0 & \text{in } \Omega \setminus D, \\ u_{\varepsilon} = 0 & \text{on } \partial D, \\ u_{\varepsilon} = 0 \text{ or } \partial u_{\varepsilon} / \nu = 0 & \text{on } \partial \Omega. \end{cases}$$

In [14, Chapter 3], the leading-order term for the perturbation of a simple eigenvalue is obtained in the case of dimensions 2 and 3. However, these computations can be repeated exactly in the same way for  $d \ge 4$ , and the resulting asymptotic formula can be restated as follows.

**Lemma 4.10.** Given  $\lambda$  a simple eigenvalue of  $\Omega$  with associated eigenfunction u, and  $\lambda_{\varepsilon}$  the eigenvalue of  $\Omega_{\varepsilon}$  perturbed from  $\lambda$ , then

$$\lambda_{\varepsilon} - \lambda = \frac{u(z)^2}{\Gamma_0(\varepsilon)} + o\left(1/\Gamma_0(\varepsilon)\right).$$
(4.10)

We seek to apply the variational characterization of Theorem 4.7 to our case. Assumption 4.5 holds thanks to Theorems 4.2 and 4.3, and the fact the perturbation considered is restricted to a small part of the domain. Although the expression in (4.10) is not continuous under  $L^2$  convergence of u, we can easily rewrite it so that Assumption 4.6 holds. In fact, supposing  $u_n \to u$  in  $L^2$ , we can exploit the regularity properties of solutions to elliptic equations to rewrite

$$u_n(z)^2 + o\left(1/\Gamma_0(\varepsilon)\right) = \int_{B_{r_\varepsilon}} u_n^2 \xrightarrow[n \to \infty]{} \int_{B_{r_\varepsilon}} u^2 = u(z)^2 + o\left(1/\Gamma_0(\varepsilon)\right), \qquad (4.11)$$

where  $B_{r_{\varepsilon}}$  is a small enough ball centered at z. By Corollary 4.9, we immediately obtain the following result.

**Proposition 4.11.** Let  $\lambda$  be an eigenvalue of multiplicity m of the negative Laplacian on  $\Omega$  and  $u_1, \ldots, u_m$  some associated eigenfunctions orthonormal in  $L^2(\Omega)$ . Then, the largest perturbed eigenvalue behaves like

$$\lambda_{\varepsilon,m} - \lambda = \frac{\sum_{j} u_j(z)^2}{\Gamma_0(\varepsilon)} + o\left(1/\Gamma_0(\varepsilon)\right), \qquad (4.12)$$

while all the other eigenvalues behave like

$$\lambda_{\varepsilon,n} - \lambda = o(1/\Gamma_0(\varepsilon)), \quad for \ n < m.$$

*Remark* 4.12. We collect some interesting consequences of Proposition 4.11.

- (i) For  $\varepsilon$  small enough, the largest perturbed eigenvalue  $\lambda_{\varepsilon,m}$  will always be simple as long as at least one of the eigenfunctions  $u_1, \ldots, u_m$  is not zero in z.
- (ii) The eigenfunction associated to  $\lambda_{\varepsilon,m}$  converges to

$$\frac{u_1(z)u_1+\cdots+u_m(z)u_m}{\sqrt{u_1(z)^2+\cdots+u_m(z)^2}}$$

(iii) It can be shown that in two dimensions the higher-order terms in formula(4.10) can be further computed as

$$\lambda_{\varepsilon} - \lambda = \frac{u(z)^2}{\log(\varepsilon) + R(z)} + O(\varepsilon^2), \qquad (4.13)$$

where R is a function of z which does not depend on u (see Proposition 2.9). Therefore, from Corollary 4.9, we have the better approximation

$$\lambda_{\varepsilon,n} - \lambda = \begin{cases} \frac{|U(z)|^2}{\log(\varepsilon) + R(z)} + O(\varepsilon^2) & \text{if } n = m, \\ O(\varepsilon^2) & \text{if } n < m. \end{cases}$$

Example 4.13. Let  $\Omega$  be the unit square  $(0, 1)^2$  and consider the Dirichlet eigenvalue  $\lambda = 50\pi^2$  with associated orthonormal eigenfunctions  $u_1, u_2, u_3$  defined as

$$\begin{cases} u_1(x,y) = 2\sin(\pi x)\sin(7\pi y), \\ u_2(x,y) = 2\sin(7\pi x)\sin(\pi y), \\ u_3(x,y) = 2\sin(5\pi x)\sin(5\pi y). \end{cases}$$

Since for any point z in  $\Omega$  there is at least one eigenfunction which is non-zero at z, the insertion at z of a small grounded inclusion will cause an eigenvalue bifurcation of  $\lambda$ . In particular, one perturbed eigenvalue will shift from  $\lambda$  as  $1/\log(\varepsilon)$  while the other two will shift like  $O(\varepsilon^2)$ . The result of a numerical simulation is presented in Figure 4.1.



Figure 4.1: A  $\log_2-\log_2$  plot of the behaviour of an eigenvalue bifurcation from  $50\pi^2$  as the size coefficient  $\varepsilon$  of the inclusion decreases. The original domain is the unit square  $(0, 1)^2$  and the inclusion is a disk of radius  $\varepsilon$  centered at (1/4, 1/4).

#### 4.3.2 Perturbation by a conductivity inclusion

In this section, we consider a perturbation of  $\Omega$  obtained by the insertion of a small inclusion with a conductivity coefficient different from the background.

Let  $B, z, \varepsilon, D$  be defined as in Section 4.3.1. Suppose that  $-\Delta$  is perturbed into  $-\Delta_{\varepsilon}$  by inserting inside  $\Omega$  a small inclusion D of conductivity k. This causes the

eigenvalue  $\lambda$  to split into m (possibly distinct) eigenvalues  $\lambda_{\varepsilon,1} \leq \cdots \leq \lambda_{\varepsilon,m}$  such that the following system holds:

$$\begin{cases} (\Delta + \lambda_{\varepsilon,n})u_{\varepsilon,n} = 0 & \text{in } \Omega \setminus \overline{D}, \\ (k\Delta + \lambda_{\varepsilon,n})u_{\varepsilon,n} = 0 & \text{in } D, \\ u_{\varepsilon,n} \text{ continuous} & \text{along } \partial D, \\ \frac{\partial u_{\varepsilon,n}}{\partial \nu}\Big|_{+} = k \frac{\partial u_{\varepsilon,n}}{\partial \nu}\Big|_{-} & \text{on } \partial D, \\ u_{\varepsilon,n} = 0 \text{ or } \partial u_{\varepsilon,n}/\partial \nu = 0 & \text{on } \partial \Omega, \end{cases}$$
(4.14)

where n = 1, ..., m and  $u_{\varepsilon,n}$  are eigenfunctions associated to  $\lambda_{\varepsilon,n}$ .

It has been shown in [19] that if  $\lambda$  is a simple eigenvalue with associated eigenfunction u, and  $\lambda_{\varepsilon}$  is a perturbation of  $\lambda$ , then

$$\lambda_{\varepsilon} - \lambda = \varepsilon^d \langle \nabla u(z), \nabla u(z) \rangle_M + O(\varepsilon^{d+1}),$$

where  $\langle x, y \rangle_M = x \cdot M(k, B)y$  for any  $x, y \in \mathbb{R}^d$ , and M(k, B) is a  $d \times d$  matrix known as polarization tensor, which can be defined by

$$(M(k,B))_{i,j} = \int_{\partial B} \left( \frac{k+1}{2(k-1)} I - (\mathcal{K}_B^0)^* \right)^{-1} (\nu_j) y_i \, d\sigma(y).$$

Therefore, in this case Corollary 4.8 specifies to the following result.

**Proposition 4.14.** Let  $\lambda$  be an eigenvalue of multiplicity m of  $-\Delta$  and let  $u_1, \ldots, u_m$ be some associated eigenfunctions orthonormal in  $L^2(\Omega)$ . Let  $\lambda_{\varepsilon,1} \leq \cdots \leq \lambda_{\varepsilon,m}$  be the eigenvalues perturbed from  $\lambda$  which are solutions to (4.14). Then, for every  $n \in \{1, \ldots, m\}$ , the  $O(\varepsilon^{d+1})$  approximation of  $\lambda_{\varepsilon,n} - \lambda$  is given by the n-th eigenvalue of the matrix with element

$$\varepsilon^d \langle \nabla u_i(z), \nabla u_j(z) \rangle_M$$

in position (i, j).

*Remark* 4.15. We consider some extremal cases of Proposition 4.14 that are of particular interest.

(i) If  $\langle \nabla u_i(z), \nabla u_j(z) \rangle_M = 0$  for all  $i \neq j$ , then it is enough to reorder  $u_1, \ldots, u_m$ according to the magnitude of  $\langle \nabla u_1(z), \nabla u_1(z) \rangle_M, \ldots, \langle \nabla u_m(z), \nabla u_m(z) \rangle_M$ , to obtain that for any n it holds

$$\lambda_{\varepsilon,n} - \lambda = \varepsilon^d \langle \nabla u_n(z), \nabla u_n(z) \rangle_M + O(\varepsilon^{d+1}).$$

55

We also remark that if the multiplicity m is larger than the dimension d, only d vectors can be linearly independent, and thus we will have that  $\lambda_{\varepsilon,n} - \lambda = O(\varepsilon^{d+1})$  for  $n \leq m - d$ .

(ii) If  $\nabla u_i(z)$ ,  $\nabla u_j(z)$  are parallel with respect to  $\langle \cdot, \cdot \rangle_M$ , then, by Corollary 4.9, we will have

$$\lambda_{\varepsilon,n} - \lambda = \begin{cases} \varepsilon^d \sum_{n=1}^m \langle \nabla u_n(z), \nabla u_n(z) \rangle_M + O\left(\varepsilon^{d+1}\right) & \text{if } n = m, \\\\ O\left(\varepsilon^{d+1}\right) & \text{if } n < m. \end{cases}$$

Example 4.16. Let  $\Omega$  be the unit square  $(0, 1)^2$  and consider the Neumann eigenvalue  $\lambda = 100\pi^2$  with associated eigenfunctions  $u_1, u_2, u_3, u_4$  defined as

$$\begin{cases} u_1(x,y) = \sqrt{2}\cos(10\pi y), \\ u_2(x,y) = \sqrt{2}\cos(10\pi x), \\ u_3(x,y) = 2\cos(6\pi x)\cos(8\pi y), \\ u_4(x,y) = 2\cos(8\pi x)\cos(6\pi y). \end{cases}$$

Let *B* be the disk of radius  $1/\pi^2$  centered at 0. In this case, it can be explicitly computed that  $M(k, B) = 2(k-1)I/(\pi^2(k+1))$ . We can easily determine, reasoning as in Remark 4.15, whether the first term in the asymptotic expansion of  $\lambda_{\varepsilon,n} - \lambda$  is zero; such behaviour will depend on the choice of *z*. For example:

- for z = (1/2, 1/2), it holds  $\nabla u_n(z) = 0$  for any n, and therefore  $\lambda_{\varepsilon,n} \lambda = O(\varepsilon^3)$ ;
- for z = (1/3, 1/2), ∇u<sub>1</sub>(z) = ∇u<sub>3</sub>(z) = 0 while ∇u<sub>2</sub>(z), ∇u<sub>4</sub>(z) are all parallel and non-zero, thus from Item ii of Remark 4.15 we have λ<sub>ε,n</sub> − λ = O(ε<sup>3</sup>) for n = 1, 2, 3 while λ<sub>ε,4</sub> − λ = Θ(ε<sup>2</sup>);
- for z = (1/7, 1/7) computations of the gradient of the eigenfunctions at z show that  $\lambda_{\varepsilon,n} - \lambda = O(\varepsilon^3)$  for n = 1, 2 and  $\lambda_{\varepsilon,n} - \lambda = \Theta(\varepsilon^2)$  for n = 3, 4;
- by Item i of Remark 4.15 there is no z ∈ Ω such that λ<sub>ε,n</sub> − λ = Θ(ε<sup>2</sup>) for more than two different indices n.

Example 4.17. Let  $\Omega$  be the unit square  $(0, 1)^2$  and consider the Neumann eigenvalue  $\lambda = 4\pi^2$  with associated eigenfunctions  $u_1, u_2$  defined as

$$\begin{cases} u_1(x,y) = \sqrt{2}\cos(2\pi y), \\ u_2(x,y) = \sqrt{2}\cos(2\pi x). \end{cases}$$

Let B be the disk of radius  $1/\pi^2$  centered at 0. Recall that in this case we have  $M(k,B) = 2(k-1)I/(\pi^2(k+1))$ . Although the first term in the asymptotic formula for  $\lambda_{\varepsilon,n} - \lambda$  can be easily computed in this case, here we focus our attention only on the asymptotic order. We can easily determine, reasoning as in Remark 4.15, whether the first term in the asymptotic expansion of  $\lambda_{\varepsilon,n} - \lambda$  is zero; such behaviour will depend on the choice of z. For example:

- for z = (1/2, 1/2), we have that both eigenfunctions  $u_1, u_2$  have zero gradient at 0, and thus both eigenvalues shift from  $\lambda$  as  $O(\varepsilon^3)$ ;
- for z = (1/4, 1/2), one of the eigenfunctions has zero gradient while the other has a non-zero entry, thus one eigenvalue shift behaves like ε<sup>2</sup>, the other like O(ε<sup>3</sup>);
- for z = (1/4, 1/4), the gradients of the two eigenfunctions are orthogonal and non-zero, thus both eigenvalues shift from λ as ε<sup>2</sup>.

Numerical results are presented in Figure 4.2.



Figure 4.2: A  $\log_2-\log_2$  plot of the behaviour of an eigenvalue bifurcation as the size coefficient  $\varepsilon$  of the conductivity inclusion decreases. The original domain is the unit square  $(0, 1)^2$ , the inclusion a disk of conductivity k = 2, centered respectively at (1/2, 1/2) (left graph), at (1/4, 1/2) (center graph), at (1/4, 1/4) (right graph).

#### 4.3.3 Perturbation by boundary deformation

In this section, we consider  $\Omega_{\varepsilon}$  obtained from  $\Omega$  by a boundary deformation. For simplicity, we suppose that  $\Omega$  is globally the epigraph of a Lipschitz function  $\varphi$ , that is  $\Omega = \{x \in \mathbb{R}^d : \varphi(x) \leq 0\}$ . Given  $w \in C^2(\partial\Omega)$ , we also suppose that the boundary perturbation is such that  $\Omega_{\varepsilon} = \{x \in \mathbb{R}^d : \varphi(x) + \varepsilon w(x)\nu(x) \leq 0\}$ . Recall that if  $\lambda$  is a simple eigenvalue of  $\Omega$  with associated eigenfunction u, and  $\lambda_{\varepsilon} \to \lambda$ , then Hadamard's formula reads

$$\frac{\partial \lambda_{\varepsilon}}{\partial \varepsilon}\Big|_{\varepsilon=0} = \int_{\partial \Omega} \left( |\nabla u|^2 - \lambda u^2 - 2(\partial u/\partial \nu)^2 \right) w, \tag{4.15}$$

for Dirichlet or Neumann conditions on  $\partial\Omega$  (for its proof see [28] or [51]). Therefore, if  $\lambda$  has multiplicity m, an application of Theorem 4.7 provides us with the variational formula

$$\lambda_{\varepsilon,n} - \lambda = \varepsilon \operatorname*{argim}_{v \in S_n} \int_{\partial \Omega} \left( |\nabla v|^2 - \lambda v^2 - 2(\partial v/\partial \nu)^2 \right) w + O(\varepsilon^2).$$
(4.16)

Example 4.18. Let  $\Omega = (0,1)^2$  and consider the Dirichlet eigenvalue  $\lambda = 10\pi^2$ with associated orthonormal eigenfunctions  $u_1(x,y) = 2\sin(\pi x)\sin(3\pi y)$ ,  $u_2(x,y) = 2\sin(3\pi x)\sin(\pi y)$ . Suppose that the boundary of  $\Omega$  is perturbed on the upper side of the square  $\Omega$  with  $w(x,y) = \sin(\pi x)$ , that is  $\partial\Omega \cap \{y=1\}$  is deformed into  $\{(x,y): 0 \le x \le 1, y = 1 + \varepsilon \sin(\pi x)\}$ . In this case, the integral in (4.15) is analytically computable as

$$-\int_{\partial\Omega} (a_1 \partial u_1 / \partial \nu + a_2 \partial u_2 / \partial \nu)^2 w = \frac{16\pi}{35} (-105a_1^2 + 14a_1a_2 - 9a_2^2).$$

Thus, by (4.16), we can calculate explicitly

$$u_{\varepsilon,1} \to c_1 u_1 + c_2 u_2, \qquad \lambda_{\varepsilon,1} - \lambda = C_1 \varepsilon + O(\varepsilon^2), u_{\varepsilon,2} \to c_2 u_1 - c_1 u_2, \qquad \lambda_{\varepsilon,2} - \lambda = C_2 \varepsilon + O(\varepsilon^2),$$
(4.17)

where  $c_1 \simeq 0.997, c_2 \simeq -0.0724, C_1 \simeq -152, C_2 \simeq -12.2$ . Numerical results are presented in Figures 4.3 and 4.4.

Notice that in general, minimizing the expression in (4.16) is a computationally expensive task. However, we can still obtain some cheaper, qualitative information if we approximate to a more treatable form the considered domain perturbation. We showcase such an heuristic in the following example, where a local perturbation is "singularized" to obtain an asymptotic formula easier to analyze.



Figure 4.3: Behaviour of an eigenvalue bifurcation as the scaling parameter  $\varepsilon$  of the boundary deformation decreases. The original domain is the unit square  $(0, 1)^2$  and the boundary deformation is given by  $\varepsilon \sin(\pi x)\nu(x, 1)$  on the upper side.



Figure 4.4: Eigenfunctions associated to the eigenvalues perturbed from  $10\pi^2$ . The perturbation consists in a boundary deformation  $\varepsilon \sin(\pi x)\nu(x, 1)$  of the upper side for  $\varepsilon = 1/10$ . Numerical computations obtained with the finite element method of the eigenfunctions are plotted in the left column, their limiting functions as  $\varepsilon \to 0$ in the two-dimensional eigenspace of  $4\pi^2$  given by (4.17) in the right column.

*Example* 4.19. Suppose a small dent is present on the surface  $\partial \Omega$  at the point z, shaped as a cone with circular base of radius  $\delta$  and height  $\varepsilon$ . Let at first  $\lambda$  be a simple Dirichlet eigenvalue with associated eigenfunction u. If we approximate

$$\int_{\partial\Omega} (\partial u/\partial\nu)^2 w \simeq \delta^2 \pi (\partial u(z)/\partial\nu)^2,$$

then for eigenvalues low enough we can estimate

$$\lambda_{\varepsilon} - \lambda \simeq -\varepsilon \delta^2 \pi (\partial u(z) / \partial \nu)^2. \tag{4.18}$$

The right hand side in (4.18) is bilinear in u therefore, if we adopt such an approximation for  $\lambda_{\varepsilon} - \lambda$ , by Corollary 4.9 we have that for any non-simple eigenvalue the largest perturbed eigenvalue will shift like  $O(\varepsilon)$  while all the smaller ones will shift like  $O(\varepsilon^2)$ .

#### 4.3.4 Summary of results

We summarize hereafter the main results obtained for each of the perturbations considered. For this purpose, we recall that  $\Omega$  indicates an arbitrary Lipschitz domain in  $\mathbb{R}^d$ ,  $\lambda$  an eigenvalue of the negative (Dirichlet or Neumann) Laplacian on  $\Omega$ ,  $u_1, \ldots, u_m$  an arbitrary orthonormal basis in  $L^2(\Omega)$  of the eigenspace of  $\lambda$ , and  $\lambda_{\varepsilon,1} \leq \cdots \leq \lambda_{\varepsilon,m}$  the eigenvalues perturbed from  $\lambda$ .

• When a hole D of volume  $\varepsilon^d$  and centered at z is cut out from  $\Omega$  and homogeneous Dirichlet boundary conditions are imposed on  $\partial D$ , we have

$$\lambda_{\varepsilon,m} - \lambda = C\varepsilon^{d-1} \sum_{i=1}^{m} u_i(z)^2 + O(\varepsilon^d),$$
  
$$\lambda_{\varepsilon,n} - \lambda = O(\varepsilon^d) \quad \text{for } n < m,$$

in the case where  $d \ge 3$  (see (4.13) for the case d = 2), where C is a constant which depends only on the dimension d. Therefore, we have that the largest eigenvalue splits at a higher asymptotic order than all the others eigenvalues, as long as one among the quantities  $u_1(z), \ldots, u_m(z)$  is non-zero.

• In the case of a conductivity inclusion, we do not have such an explicit formula, but we can still easily recover a first-order approximation by computing the eigenvalues of a finite matrix. More precisely, if we suppose to change the conductivity coefficient from 1 to k only in D, a small disk of radius  $\varepsilon$  centered at a point z, then for any  $n \in \{1, \ldots, m\}$ ,

$$\lambda_{\varepsilon,n} - \lambda = 2\frac{k-1}{k+1}\varepsilon^d \mu_n + O(\varepsilon^{d+1}),$$

where  $\mu_n$  is the *n*-th eigenvalue of the matrix with element  $\nabla u_i(z) \cdot \nabla u_j(z)$  in position (i, j).

• In the case of a normal boundary deformation of  $\Omega$  with shape  $w \in C^2(\partial \Omega)$ , that is the perturbed domain boundary is given locally by  $\partial \Omega + \varepsilon w \nu$ , to find  $\lambda_{\varepsilon,n} - \lambda$  for any  $n \in \{1, \ldots, m\}$ , one has to find the minimizer  $v_n$  of

$$J(v) = \int_{\partial\Omega} \left( |\nabla v|^2 - \lambda v^2 - 2(\partial v/\partial \nu)^2 \right) w,$$

among all v's of unit L<sup>2</sup>-norm in the eigenspace of  $\lambda$  and perpendicular to  $v_1, \ldots, v_{n-1}$ . Then

$$\lambda_{\varepsilon,n} - \lambda = \varepsilon J(v_n).$$

As a final remark, let us point out that similar formulae can be derived for many other types of domain perturbations or other differential operators. For example, with the same approach of Section 4.3.3, it is immediate to generalize the asymptotic expansion of eigenvalues in the case of shape deformation of conductivity inclusions (see [3, Theorem 2.1]); or, with the same approach of Section 4.3.2, to generalize the asymptotic formulae for eigenvalues of the Lamé operator in the context of linear elasticity (see [4, Theorem 2.1]).

# Chapter 5

# Reconstruction of small perturbations from eigenvalues' shifts

A foundational question in spectral geometry is whether the eigenvalues of a differential operator characterize the shape of the domain on which it is defined. Although in general the answer is negative, in many special cases of interest the answer is affirmative (for instance, a square is uniquely characterized by its first Dirichlet or Neumann Laplacian eigenvalue). In this chapter we try to devise some practical procedures to reconstruct a domain from its eigenvalues under some restricting assumptions. More precisely, we suppose that our unknown domain is given by a small perturbation from a known domain, that the perturbation is completely characterized by some parameters, and that we intend to estimate these parameters from the knowledge of the eigenvalues of both the known and the unknown domain.

More formally, let  $\Omega$  be a domain and let  $(\lambda_j)_{j \in \mathbb{N}}$  be the sequence of eigenvalues of the negative Laplacian on  $\Omega$  with Dirichlet or Neumann boundary conditions. Consider the first N different eigenvalues of  $\Omega$ ,

$$\lambda_1 = \dots = \lambda_{M_1} < \lambda_{M_1+1} = \dots = \lambda_{M_2} < \dots < \lambda_{M_{n-1}+1} = \dots = \lambda_{M_N},$$

where  $M_j = m_1 + \cdots + m_j$  and  $m_j$  is the multiplicity of  $\lambda_j$ . Let  $u_j$  indicate an eigenfunction associated to  $\lambda_j$ , and suppose that  $u_1, \ldots, u_{M_N}$  are orthonormal in  $L^2(\Omega)$ .

Suppose now that  $\Omega$  is perturbed into another domain  $\Omega_{\varepsilon}$ . We write down the

perturbed eigenvalues of  $\Omega_{\varepsilon}$  as

$$\lambda_{\varepsilon,1} \leq \cdots \leq \lambda_{\varepsilon,M_1} < \lambda_{\varepsilon,M_1+1} \leq \cdots \leq \lambda_{\varepsilon,M_2} < \cdots < \lambda_{\varepsilon,M_{n-1}+1} \leq \cdots \leq \lambda_{\varepsilon,M_N},$$

and suppose that the perturbation is small enough so that there is no eigenvalue crossing, i.e.,  $\lambda_{\varepsilon,j} \to \lambda_j$  as  $\varepsilon \to 0$  for  $j = 1, \ldots, M_N$ . In this chapter, we show how to reconstruct some features of  $\Omega_{\varepsilon}$ , via the asymptotic formulae derived in the previous chapters, from the knowledge of  $\Omega$ , of  $\lambda_1, \ldots, \lambda_N$  and of  $\lambda_{\varepsilon,1}, \ldots, \lambda_{\varepsilon,N}$ . We consider three types of domain perturbations: grounded inclusions, conductivity inclusions, and boundary deformations.

# 5.1 Grounded inclusion

In this section we assume that the perturbation is caused by a grounded inclusion, that is  $\Omega_{\varepsilon} = \Omega \setminus \overline{D}$  where  $D = z + \varepsilon B$  and homogeneous Dirichlet boundary conditions are imposed on  $\partial D$ .

Direct problem. Let

$$U_j(z) = \begin{cases} (u_{M_{k-1}+1}(z), \dots, u_{M_k}(z)) & \text{if } j = M_k \text{ for some } k, \\ 0 & \text{otherwise.} \end{cases}$$
(5.1)

From (4.13), we know that

$$\lambda_{\varepsilon,j} - \lambda_j = \mathfrak{c}(B) \frac{|U_j(z)|^2}{\Gamma_0(\varepsilon)} + o(1/\Gamma_0(\varepsilon)), \qquad (5.2)$$

where

$$\mathfrak{c}(B) = \begin{cases} \operatorname{cap}(\partial B) & d \ge 3, \\ 1 & d = 2. \end{cases}$$
(5.3)

In two dimensions, this approximation is sometimes too poor for our purposes due to the slow convergence to 0 of  $1/\log(\varepsilon)$  as  $\varepsilon \to 0$ . In this case, we can use the more accurate, although more computationally expensive, asymptotic expression

$$\lambda_{\varepsilon,j} - \lambda_j = \frac{|U_j(z)|^2}{\Gamma_0(\varepsilon\sqrt{\lambda_j}\operatorname{cap}(\partial B)\eta_0) + \mathfrak{r}(z)} + O(\varepsilon^2),$$
(5.4)

where  $\mathfrak{r}$  is defined in (2.7).

**Inverse problem.** Our inverse problem is to reconstruct  $\varepsilon$ , z and the capacity  $\operatorname{cap}(\partial B)$  from the knowledge of  $\lambda_{\varepsilon,M_k}$  and  $\lambda_{M_k}$  for  $k = 1, \ldots, M_N$ . We will proceed in several steps, assuming the knowledge of some of these quantities to recover the others.

Notice that in the case of an eigenvalue with multiplicity  $m_k > 1$ , only the largest perturbation is of order  $O(1/\Gamma_0(\varepsilon))$  while all the others are of higher order; so only the eigenvalues  $\lambda_{\varepsilon,M_k}$  and  $\lambda_{M_k}$  will be used for the purposes of inclusion reconstruction.

Size reconstruction. Assume at first that  $\mathfrak{c}(B)$  is known. If homogeneous Neumann conditions are imposed on  $\partial\Omega$ , an estimate on the size of the inclusion can be obtained immediately from the first eigenvalue shift; in fact in this case we know that  $\lambda_1 = 0$  is simple and  $u_1^2 \equiv 1/|\Omega|$ . Thus, from (5.2), we immediately have the approximation

$$\varepsilon \simeq \Gamma_0^{-1} (\mathfrak{c}(B)/(|\Omega|\lambda_{\varepsilon,1})).$$
 (5.5)

This estimate is sometimes not accurate enough, especially in two dimensions.

Let us assume that the position z is also known. Then we can obtain an improved approximation by considering multiple eigenvalues and looking for a minimizer of the discrepancy function (obtained from (5.2))

$$\delta \mapsto \sum_{j=1}^{N} \left( \lambda_{\varepsilon,j} - \lambda_j - \mathfrak{c}(B) |U_j(z)|^2 / \Gamma_0(\delta) \right)^2,$$
(5.6)

or, in two dimensions, for a minimizer of

$$\delta \mapsto \sum_{j=1}^{N} \left( \lambda_{\varepsilon,j} - \lambda_j - \frac{|U_j(z)|^2}{\Gamma_0(\delta\sqrt{\lambda_j}\operatorname{cap}(\partial B)\eta_0) + R_j(z)} \right)^2.$$
(5.7)

We remark that both the last two approaches are viable also under homogeneous Dirichlet boundary conditions on  $\partial \Omega$ .

Example 5.1. Consider the case where  $\Omega = B \subseteq \mathbb{R}^2$  is the unit disk and Neumann boundary conditions are imposed on  $\partial\Omega$ . We want to reconstruct the size coefficient  $\varepsilon$  from the knowledge of some eigenvalues' shifts and the position z. A reconstruction based only on the shift of the first eigenvalue through formula (5.5) is presented in Figure 5.1. The values obtained from the asymptotic formulae are confronted with the accurate estimates given by the multipole expansion method. A reconstruction of  $\varepsilon$  via the minimization of (5.6) and (5.7), known the perturbation of the first five eigenvalues, are presented respectively in Figures 5.2 and 5.3. Notice how the reconstruction via the more computationally expensive formula (5.7) is more accurate, especially when the center z of the perturbation is near the boundary of  $\Omega$ , compared to the reconstruction via (5.6) or (5.5).



Figure 5.1: A log-log plot of  $\varepsilon$  as a function of  $\lambda_{\varepsilon,1}$  for different choices of z (from left to right, |z| = .3, .6, .9).



Figure 5.2: A plot of the discrepancy function (5.6) with N = 5 and  $|\varepsilon| = 2^{-6}$  for different choices of z (from left to right |z| = .3, .6, .9).

**Position reconstruction.** Suppose that the size  $\varepsilon$  and the capacity  $\operatorname{cap}(\partial B)$  are known. A position estimate can be obtained by searching for a z which satisfies (5.2). An important warning in this case is that there is no guarantee of uniqueness. In fact, from a single eigenvalue shift one can obtain at most all the points which lie on the same level set of  $|U_j|^2$ . A possible solution to this problem would be to take many eigenvalue shifts, thus restricting significantly the possible values which z can take. More specifically, one hopes that the intersection of the level sets of different eigenfunctions becomes a point if we consider enough eigenfunctions. Although



Figure 5.3: A plot of the discrepancy function (5.7) with N = 5 and  $|\varepsilon| = 2^{-6}$  for different choices of z (from left to right, |z| = .3, .6, .9).

this might be true for some non-symmetric domains, it is false for domains which present some symmetries. However, as is shown in the following examples, we can still recover some useful information even in the case of a symmetric domain.

Example 5.2 (Ball or spherical shell domain). Let  $\Omega \subseteq \mathbb{R}^d$  be a ball or a spherical shell (that is the set difference of two concentric balls of different radii). In this case, the Neumann eigenfunctions of  $\Omega$  can be written by separation of variables as a product of a radial and an angular part. Then, although we can not recover z univocally, we can recover its distance from the origin |z| from the shift of the first non-zero eigenvalue of  $\Omega$ . In fact, the first non-zero Neumann eigenvalue of the ball has associated eigenfunctions  $u_2, \ldots, u_{d+1}$  such that  $|U_2(x)|^2 = u_2(x)^2 + \cdots + u_{d+1}(x)^2 =$ CR(|x|), where C is a constant and R a bijective function. Therefore, from (5.2) we can recover a first-order approximation of |z|. Notice that the dependence on only the distance from the center of the squared eigenfunction actually holds for any eigenfunction, that is, for any j, a constant  $C_j$  and a function  $R_j$  can be found such that  $|U_j(x)|^2 = C_j R_j(|x|)$ . Therefore, the knowledge of more eigenvalues' shifts does not give us any additional information on z. This can be seen also as an immediate consequence of the fact that a ball with a hole at z is isometric to a ball with the same hole at x for any |x| = |z|.

Example 5.3 (Cubic domain). Let  $\Omega = (0, \pi)^d$  be a *d*-dimensional cube. It is well known that the Neumann eigenfunctions and eigenvalues of  $\Omega$  are given respectively by  $\cos(n_1x_1) \dots \cos(n_dx_d)$  and  $n_1^2 + \dots + n_d^2$  for  $n_1, \dots, n_d$  non-negative integers. Therefore, if we know the shift of the first non-zero eigenvalue, that is 1, we can recover from (5.2) the quantity  $f(z) = \cos(z_1)^2 + \dots + \cos(z_d)^2$ . If we also know the shift of the second non-zero eigenvalue, that is 4, we can recover  $g(z) = \cos(2z_1)^2 + \cdots + \cos(2z_d)^2$ . It can be shown that any two level sets of f and g intersect at most at  $d2^d$  points, where each point is the image through a reflection along a symmetry hyperplane of the cube (some of these points might coincide when z lies on a symmetry hyperplane of  $\Omega$ ); see Figure 5.4 for a plot of the level sets of squared eigenfunctions in the two dimensional case. Since we know that such domains are



Figure 5.4: Level sets of squared eigenfunctions.

isometric, using more eigenvalues won't give us any additional information on the position z.

Capacity and shape reconstruction. If the size coefficient  $\varepsilon$  and position z are known, then the capacity  $cap(\partial B)$  can be recovered by inverting (5.2). The capacity can be useful to determine the shape of an inclusion, if we know a priori that it belongs to a certain class of shapes, like for instance in the following example.

Example 5.4. Let  $B \subseteq \mathbb{R}^2$  be an ellipse of semi-axes  $a_1, a_2$  such that  $a_1a_2 = 1$ . Then cap  $\partial B = (a_1 + 1/a_1)/2$ . From the knowledge of  $\varepsilon$  and z, an estimate of  $a_1$  can then be obtained by using (5.2). A numerical computation of  $a_1$  as a function of  $\lambda_{\varepsilon,1} - \lambda_1$ in the case of  $\Omega$  a disk of radius 1 and Dirichlet boundary conditions on  $\partial \Omega$  is given in Figure 5.5.



Figure 5.5: Semi-axis length of an elliptic inclusion as a function of an eigenvalue shift.

Even if in most cases we can not determine univocally the shape of the inclusion from its capacity, we can at least reduce the dimension of the parameter space which describes its shape.

Example 5.5. Let  $B \subseteq \mathbb{R}^3$  be an ellipsoid of semi-axes  $a_1, a_2, a_3$  such that  $4/3\pi a_1 a_2 a_3 =$ 1. Then by [56, p. 156],

$$\frac{1}{\operatorname{cap}\partial B} = 2\pi \int_0^\infty \frac{dx}{\sqrt{a_1^2 + x}\sqrt{a_2^2 + x}\sqrt{(3/(4\pi a_1 a_2))^2 + x}}.$$
(5.8)

Therefore, level curves (see Figure 5.6) of the integral in the previous equation seen as a function of  $a_1, a_2$  will represent the continuous class of ellipsoids with area 1 which cause a given eigenvalue perturbation.



Figure 5.6: level curves of the integral in (5.8)

**General reconstruction.** If the capacity is known, the size and position can be estimated from the knowledge of only two eigenvalues as follows. By considering the ratio of the perturbations, one can estimate the position looking for the level sets of the ratio of squared eigenfunction. Then,  $\varepsilon$  can be found by backsubstituting the value of the eigenfunction estimated at z.

The case where the size, position and capacity are unknown and have to be reconstructed from the eigenvalue perturbation is much more delicate. There may be multiple classes of inclusions with different size, position and shape which cause a certain eigenvalue perturbation. If some a priori information on these features is known, the search in the parameter space can be limited, as has been done in the previously considered cases, but a general approach does not appear to be viable.

# 5.2 Conductivity inhomogeneity

In this section, we suppose that the perturbation is caused by a conductivity inclusion, that is,  $\Omega_{\varepsilon} = \Omega \setminus \overline{D}$  with  $D = z + \varepsilon B$  and  $k \neq 1$  being the conductivity of D.

**Direct problem.** In this case, the perturbed eigenvalues satisfy the system

$$\begin{cases} (\Delta + \lambda_{\varepsilon,n})u_{\varepsilon,n} = 0 & \text{in } \Omega \setminus \overline{D}, \\ (k\Delta + \lambda_{\varepsilon,n})u_{\varepsilon,n} = 0 & \text{in } D, \\ u_{\varepsilon,n} \text{ continuous} & \text{along } \partial D, \\ \frac{\partial u_{\varepsilon,n}}{\partial \nu}\Big|_{+} = k \frac{\partial u_{\varepsilon,n}}{\partial \nu}\Big|_{-} & \text{on } \partial D, \\ u_{\varepsilon,n} = 0 \text{ or } \partial u_{\varepsilon,n}/\partial \nu = 0 & \text{on } \partial \Omega. \end{cases}$$
(5.9)

From Proposition 4.14, we have the asymptotic expansion

$$\lambda_{\varepsilon,j} - \lambda_j = \varepsilon^d \mu_j + O(\varepsilon^{d+1}), \tag{5.10}$$

where  $\mu_j$  is the *j*-th eigenvalue of the matrix

$$\left(\nabla u_p(z) \cdot M(k, B) \nabla u_q(z)\right)_{p=M_{j-1}+1, \dots, M_j}^{q=M_{j-1}+1, \dots, M_j},$$
(5.11)

with M(k, B) being the polarization tensor matrix (see (1.12)). We consider hereafter the reconstruction of different features of the conductivity inclusion. Size reconstruction. Suppose that the position and the polarization tensor of B are known. Then the problem of reconstructing the size coefficient  $\varepsilon$ , as in Section 5.1, can be solved directly by inverting (5.10). In the same way, a finer estimate can be obtained by minimizing a discrepancy functional which takes into account the perturbation of more than one eigenvalue.

**Position reconstruction.** Suppose that the size coefficient  $\varepsilon$  and the polarization tensor of B are known. Then, we can reconstruct the position z by looking at the level sets of  $\nabla U \cdot M \nabla U$ .

Example 5.6. Let  $\Omega = B \subseteq \mathbb{R}^2$  be the unit disk. It is known that the polarization tensor is M = 2(k-1)I/(k+1). Therefore, in this case we need to look for intersections of levels sets of the squared norm of the gradient of the eigenfunctions. Since they are radially symmetric, we can reconstruct the distance of the particle from the center.

Polarization tensor, conductivity and shape reconstructions. We can reconstruct the polarization tensor using formula (5.10) for different frequencies  $\lambda_{\varepsilon,1}, \ldots$ ,  $\lambda_{\varepsilon,M_N}$  and looking for a  $d \times d$  matrix which minimizes the discrepancy.

If we know a priori the shape of the inclusion B, we can directly reconstruct its conductivity, as is done in the following simple example.

Example 5.7. Suppose that  $\Omega = B \subseteq \mathbb{R}^2$  is the unit disk and suppose that we are interested in reconstructing the conductivity k of B. Fix the size coefficient  $\varepsilon = 0.1$  and the center of B in (0.3, 0). A plot of k as a function of the eigenvalue shift is given in Figure 5.7.





### 5.3 Boundary deformation

In this final section, we suppose only the boundary of the domain is deformed, and try to reconstruct some characterizing features.

**Direct problem.** Let  $\phi(t, x) : [0, T] \times \mathbb{R}^d \to \mathbb{R}^d$  be such that  $\phi(0, x) = x$  for any  $x \in \mathbb{R}^d$  and let  $\phi(t, \Omega)$  be diffeomorphic to  $\Omega$  for any t. Let  $\lambda(t)$  be a simple Dirichlet eigenvalue of  $\phi(t, \Omega)$ . Then, from (3.5), we have that

$$\dot{\lambda}(t) = -\int_{\phi(t,\partial\Omega)} (\partial u(t,y)/\partial\nu(t,y))^2 \dot{e}(t,y) \cdot \nu(t,y) \, d\sigma(y), \tag{5.12}$$

where  $u(t, \cdot)$  is the L<sup>2</sup>-normalized eigenfunction of  $\phi(t, \Omega)$  associated to  $\lambda(t)$ ,  $\nu(t, \cdot)$  is the outward normal to the surface  $\phi(t, \partial\Omega)$ ,  $e(t, \cdot)$  is the identity on  $\phi(t, \partial\Omega)$ . By a change of variable,

$$\dot{\lambda}(t) = -\int_{\partial\Omega} (\partial u(t,\phi(t,x))/\partial\eta(t,x))^2 \dot{\phi}(t,x) \cdot \eta(t,x) G(t,x), \qquad (5.13)$$

where G is the absolute value of the determinant of the Gramian of  $\phi$  and  $\eta(t, x) = \nu(t, \phi(t, x))$ .

If w is a smooth function defined on  $\partial \Omega$  such that

$$\phi(t,x) = x + tw(x)\nu(0,x) \qquad \forall x \in \partial\Omega, \tag{5.14}$$

then (5.13) simplifies to

$$\dot{\lambda}(0) = -\int_{\partial\Omega} (\partial u/\partial\eta)^2 w \, d\sigma(x), \qquad (5.15)$$

where  $u, \eta, \dot{\eta}, G$  are assumed to be functions of (0, x), and w to be function of x. We are interested in using this first-order asymptotic formula to reconstruct the magnitude of the boundary deformation.

**Inverse problem.** We consider two simple examples of boundary deformation reconstruction which are of interest in applications.

Example 5.8 (Deformation of a side of a rectangle). Consider the rectangle  $\Omega = [0, 1] \times [-l, 0]$ . Suppose for now that l is irrational, so that all eigenvalues of  $\Omega$  are simple. More precisely, we have that the L<sup>2</sup>-normalized Dirichlet eigenfunctions of  $\Omega$  are given by

$$u_{n,m}(x,y) = \frac{1}{4}\sin(n\pi x)\sin(m\pi y/l),$$
and the associated eigenvalues by  $\lambda_{n,m}(0) = (n\pi)^2 + (m\pi/l)^2$ , for n, m positive integers.

Fix k and let  $w(s) = \sin(k\pi s)$  in (5.14). By explicit computations, we have that (5.15) becomes

$$\dot{\lambda}_{n,m}(0) = -\left(\frac{m\pi}{4l}\right)^2 \int_0^1 \sin^2(n\pi s) \sin(k\pi s) \, ds.$$

Thus,

$$\dot{\lambda}_{n,m}(0) = \begin{cases} 0 & k \text{ even} \\ \\ \frac{(nm)^2 \pi}{4kl^2(k^2 - 4n^2)} & k \text{ odd.} \end{cases}$$

Therefore, if k is odd,

$$\lambda_{n,m}(t) - \lambda_{n,m}(0) = t \frac{(nm)^2 \pi}{4kl^2(k^2 - 4n^2)} + O(t^2), \qquad (5.16)$$

and so we can recover a first-order approximation of t from the knowledge of a single arbitrary eigenvalue shift. To find an approximation of t when k is even, we would need higher-order terms.

Suppose now that we want to determine the shape of the internal cavity inside a hollow cylinder, knowing that it is a small perturbation of a concentric cylinder. For simplicity, we reduce our analysis to a two dimensional section, but the same technique used for this case can be adapted to the higher dimensions.

Example 5.9 (Internal deformation of an annulus). Let  $\Omega$  be an annulus of external radius 2 and internal radius 1. Let  $\Gamma^e$ ,  $\Gamma^i$  indicate respectively the circle of radius 2 and the circle of radius 1. By separation of variables, it can be shown that the Dirichlet eigenfunctions of  $\Omega$  associated to an eigenvalue  $\lambda_{n,k}$ , with  $n \in \mathbb{N}$  and  $k \in \mathbb{Z}^+$ , can be written in polar coordinates as

$$u_{n,k}(r,\mathfrak{t}) = R_{n,k}(r)\cos(n\mathfrak{t}), \qquad (5.17)$$

$$v_{n,k}(r,\mathfrak{t}) = R_{n,k}(r)\sin(n\mathfrak{t}), \qquad (5.18)$$

where

$$R_{n,k}(r) = \alpha_n \left( J_n(\sqrt{\lambda_{n,k}}r) + c_{n,k}Y_n(\sqrt{\lambda_{n,k}}r) \right), \tag{5.19}$$

and the constants  $\lambda_{n,k}$  and  $c_{n,k}$  are determined by the boundary conditions on  $\Gamma^e$ and  $\Gamma^i$ , and  $\alpha_n$  is a normalization constant so that the eigenfunctions have L<sup>2</sup>-norm equal to 1. Let  $\Omega_{\varepsilon}$  be given by a constant speed deformation of the internal circle  $\Gamma^{i}$  with shape  $w \in C^{2}(\Gamma^{i})$ ; that is,  $\partial \Omega_{\varepsilon}$  is given by two disconnected sets  $\Gamma_{\varepsilon}^{e}, \Gamma_{\varepsilon}^{i}$  such that  $\Gamma_{\varepsilon}^{e} = \Gamma^{e}$  and  $\Gamma_{\varepsilon}^{i} = \Gamma^{i} + \varepsilon w(\Gamma^{i})\nu$ . The geometry of the problem allows us to expand w in a Fourier series

$$w(\mathfrak{t}) = \sum_{j=1}^{\infty} b_j \sin(j\mathfrak{t}), \qquad (5.20)$$

with  $b_j \geq 0$ . We can also explicitly compute that the  $O(\varepsilon^2)$  approximation of  $\lambda_{\varepsilon,n,k}^1 - \lambda_{n,k}$  is given by

$$-\varepsilon \min_{A,B:A^2+B^2=1} \int_{\Gamma^i} \left( \partial (Au_{n,k} + Bv_{n,k}) / \partial \nu \right)^2 w$$
(5.21)

$$= \varepsilon R_{n,k}'(1) \min_{\varphi \in [0,2\pi]} \int_0^{2\pi} \frac{1 + \sin(2\varphi)\sin(2n\mathfrak{t}) + \cos(2\varphi)\cos(2n\mathfrak{t})}{2} w(\mathfrak{t}) d\mathfrak{t} \qquad (5.22)$$

$$= \frac{\varepsilon \pi}{2} R'_{n,k}(1) \min_{\varphi \in [0,2\pi]} b_{2n} \sin(2\varphi).$$
(5.23)

Therefore

$$\lambda_{\varepsilon,n,k}^1 - \lambda_{n,k} = -\frac{\varepsilon\pi}{2} R'_{n,k}(1) |b_{2n}| + O(\varepsilon^2), \qquad (5.24)$$

$$\lambda_{\varepsilon,n,k}^2 - \lambda_{n,k} = O(\varepsilon^2). \tag{5.25}$$

Thus, we can recover the 2*n*-th coefficient of the Fourier series of w directly from the shift of the smallest eigenvalue perturbed from  $\lambda_{n,k}$ , for an arbitrary choice of k.

## Chapter 6

# Perturbation of scattering resonances - transverse magnetic polarization case

The influence of a small particle on a cavity mode plays an important role in fields such as optical sensing, cavity quantum electrodynamics, and cavity optomechanics [53, 92, 82]. In this chapter, we consider the transverse magnetic polarization case and provide a formal derivation of the perturbations of scattering resonances of an open cavity due to a small-volume particle without neglecting the radiation effect. Note that the radiation effect has been omitted in the physics literature (see, for instance, [45]). Indeed, the Bethe-Schwinger closed cavity perturbation formula [23, 31 has been widely employed for radiating cavities. The small-volume asymptotic formula in this chapter generalizes to the open cavity case those derived in [14, 16, 78, 23]. It is valid for arbitrary-shaped particles. It shows that the perturbations of the scattering resonances can be expressed in terms of the polarization tensor of the small particle. Two cases are considered: the one-dimensional case and the multidimensional case. Its applicability to the perturbations of whispering-gallery modes by external arbitrary-shaped particles is also discussed. Finally, we characterize the effect that a plasmonic nanoparticle, of arbitrary geometry and which is bound to the surface of the cavity, has on the whispering-gallery modes of the cavity. Since the shift of the scattering frequencies is proportional to the polarization of the plasmonic nanoparticles [8, 17, 18, 21], which blows-up at the plasmonic resonances, the effect of a plasmonic particle on the cavity modes can be significant.

For the analysis of the transverse electric case we refer to Chapter 7. Note that in the one-dimensional case, the scattering resonances are simple while in the multi-dimensional case, they can be degenerate or even exceptional. The analysis of exceptional points is a challenging problem, and the situation is much simpler in the transverse electric case than in the transverse magnetic one. The reader is also referred to [54, 55, 63] for small amplitude sensitivity analyses of the scattering resonances. Numerical computation of resonances has been addressed, for instance, in [50, 64, 70, 71, 85, 100].

The chapter is organized as follows. In Section 6.1, using the method of matched asymptotic expansions, we derive the leading-order term in the shifts of scattering resonances of a one-dimensional open cavity and characterize the effect of radiation. Section 6.2 generalizes the method to the multi-dimensional case. In Section 6.3, we consider the perturbation of whispering-gallery modes by small particles. The formula obtained for the shifting of the frequencies shows a strong enhancement in the frequency shift in the case of plasmonic particles, which allows for their recognition in spite of their small size. The splitting of scattering frequencies of the open cavity of multiplicity greater than one due to small particles is also discussed. In Section 6.4, we present some numerical examples to illustrate the accuracy of the formulas derived in this chapter and their use in the sensing of small particles. The chapter ends with some concluding remarks.

#### 6.1 One dimensional case

We first consider a one dimensional cavity. We let the magnetic permeability  $\mu_{\delta}$  be  $\mu_m$  in  $(a, b) \setminus (-\delta/2, \delta/2)$  and  $\mu_c$  in  $(-\delta/2, \delta/2)$  and the electric permittivity  $\varepsilon_{\delta}$  be  $\varepsilon_m$  in  $(a, b) \setminus (-\delta/2, \delta/2)$  and  $\varepsilon_c$  in  $(-\delta/2, \delta/2)$ . Here,  $0 < \delta < 1/2$  and  $\mu_m, \mu_c, \varepsilon_m$ , and  $\varepsilon_c$  are positive constants.

Let  $\omega_0$  be a scattering resonance of the unperturbed cavity and let  $u_0$  denote the corresponding eigenfunction, that is,

$$\begin{cases} \partial_x \left( (1/\varepsilon_m) \partial_x u_0 \right) + \omega_0^2 \mu_m u_0 = 0 & \text{in } (a, b), \\ (1/\varepsilon_m) \partial_x u_0 + \mathrm{i} \omega_0 u_0 = 0 & \text{at } a, \\ (1/\varepsilon_m) \partial_x u_0 - \mathrm{i} \omega_0 u_0 = 0 & \text{at } b, \\ \int_a^b |u_0|^2 \, dx = 1. \end{cases}$$

We now consider the perturbed problem: for  $\delta$  small, we seek a solution  $u_{\delta}$ , for which  $\omega_{\delta} \to \omega_0$  as  $\delta \to 0$  of the following equation:

$$\begin{cases} \partial_x \left( (1/\varepsilon_\delta) \partial_x u_\delta \right) + \omega_\delta^2 \mu_\delta u_\delta = 0 & \text{in } (a, b), \\ (1/\varepsilon_m) \partial_x u_\delta + \mathrm{i}\omega_\delta u_\delta = 0 & \text{at } a, \\ (1/\varepsilon_m) \partial_x u_\delta - \mathrm{i}\omega_\delta u_\delta = 0 & \text{at } b, \\ \int_a^b |u_\delta|^2 \, dx = 1. \end{cases}$$

$$(6.1)$$

Remark 6.1. The above one-dimensional scattering resonance problems govern scattering resonances of slab-type structures. They are a consequence of Maxwell's equations, under the assumption of time-harmonic solutions. They correspond to the transverse magnetic polarization; see [55]. The scattering resonances  $\omega_0$  and  $\omega_{\delta}$  lie in the lower-half of the complex plane. The eigenfunctions  $u_0$  and  $u_{\delta}$  satisfy the outgoing radiation conditions at a and b and, consequently, grow exponentially at large distances from the cavity. To give a physical interpretation of scattering resonances see, for instance, [50, 55].

#### **Proposition 6.2.** Assuming that

$$\omega_{\delta} = \omega_0 + \delta \omega_1 + \dots$$

we have

$$\omega_1 = \frac{\alpha (\partial_x u_0(0))^2 + \omega_0^2 \varepsilon_m (\mu_c - \mu_m) (u_0(0))^2}{2\omega_0 \mu_m \varepsilon_m + i\varepsilon_m ((u_0(a))^2 + (u_0(b))^2)},$$
(6.2)

where the polarization  $\alpha$  is defined by

$$\alpha = 1 - \frac{\varepsilon_c}{\varepsilon_m}.\tag{6.3}$$

*Proof.* Using the method of matched asymptotic expansions for  $\delta$  small, see [16], we construct asymptotic expansions of  $\omega_{\delta}$  and  $u_{\delta}$ .

To reveal the nature of the perturbations in  $u_{\delta}$ , we introduce the local variable  $\xi = x/\delta$  and set  $e_{\delta}(\xi) = u_{\delta}(x)$ . We expect that  $u_{\delta}(x)$  will differ appreciably from

 $u_0(x)$  for x near 0, but it will differ little from  $u_0(x)$  for x far from 0. Therefore, in the spirit of matched asymptotic expansions, we shall represent  $u_{\delta}$  by two different expansions, an inner expansion for x near 0, and an outer expansion for x far from 0. We write the outer and inner expansions:

$$u_{\delta}(x) = u_0(x) + \delta u_1(x) + \dots$$
 for  $|x| \gg \delta$ ,

and

$$u_{\delta}(x) = e_0(\xi) + \delta e_1(\xi) + \dots$$
 for  $|x| = O(\delta)$ .

The asymptotic expansion of  $\omega_{\delta}$  must begin with  $\omega_0$ , so we write

$$\omega_{\delta} = \omega_0 + \delta \omega_1 + \dots$$

In order to determine the functions  $u_i(x)$  and  $e_i(\xi)$ , we have to equate the inner and the outer expansions in some "overlap" domain within which the stretched variable  $\xi$  is large and x is small. In this domain the matching conditions are:

$$u_0(x) + \delta u_1(x) + \dots \sim e_0(\xi) + \delta e_1(\xi) + \dots$$

Now, if we substitute the inner expansion into (6.1) and formally set to zero the coefficients of  $\delta^{-2}$  and  $\delta^{-1}$ , then we obtain

$$\partial_{\xi}((1/\tilde{\varepsilon})\partial_{\xi}e_i) = 0, \text{ for } i = 0, 1,$$

where the stretched coefficient  $\tilde{\varepsilon}$  is equal to  $\varepsilon_c$  in (-1/2, 1/2) and to  $\varepsilon_m$  in  $(-\infty, -1/2) \cup (1/2, +\infty)$ . From the first matching condition, it follows that  $e_0(\xi) = u_0(0)$  for all  $\xi$ . Similarly, we have

$$e_1(\xi) \sim \xi \partial_x u_0(0) \quad \text{as } |\xi| \to +\infty.$$
 (6.4)

Let  $v^{(1)}(\xi)$  be such that

$$\begin{cases} \partial_{\xi}((1/\tilde{\varepsilon}(\xi))\partial_{\xi}v^{(1)}(\xi)) = 0, \\ v^{(1)}(\xi) \sim \xi \quad \text{as } |\xi| \to +\infty. \end{cases}$$

Although  $v^{(1)}$  can be trivially rewritten as a piecewise affine function, let us take an approach which generalizes easily to the multi-dimensional case. Let  $\gamma = (\varepsilon_m / \varepsilon_c) - 1$ . By how  $v^{(1)}$  is defined, we have

$$v^{(1)}(\xi) = \xi - \gamma \partial_{\xi} v^{(1)}(-1/2)|_{+} \Gamma_{0}(\xi + 1/2) + \gamma \partial_{\xi} v^{(1)}(1/2)|_{-} \Gamma_{0}(\xi - 1/2),$$

where the subscripts + and - indicate respectively the limits at 1/2 from the left and from the right. Moreover,

$$\int_{-1/2}^{1/2} \partial_{\xi}^2 v^{(1)} \, d\xi = 0,$$

yields

$$\partial_{\xi} v^{(1)}(-1/2)|_{+} = \partial_{\xi} v^{(1)}(1/2)|_{-}$$

Hence,

$$v^{(1)}(\xi) = \xi + \gamma \partial_{\xi} v^{(1)}(1/2)|_{-}G(\xi + 1/2) - \gamma \partial_{\xi} v^{(1)}(1/2)|_{-}G(\xi - 1/2).$$

On the other hand,

$$G(\xi \pm 1/2) \sim |\xi| \pm \xi/(2|\xi|) + \dots$$
 as  $|\xi| \to +\infty$ .

Therefore,

$$v^{(1)}(\xi) \sim \xi - ((\varepsilon_m/\varepsilon_c) - 1)\partial_{\xi}v^{(1)}(1/2)|_{-} \xi/|\xi| + \dots$$

From [16, (4.11)], we have that  $e_1(\xi) = (\partial_x u_0)(0)v^{(1)}(\xi)$ . By matching the first-order terms in the inner and outer expansion we obtain  $u_1(x) \sim e_1(\xi) - \xi(\partial_x u_0)(0)$ . Then, the second matching condition (6.4) yields

$$u_1(x) \sim -\gamma(\partial_x u_0(0))(\partial_\xi v^{(1)}(1/2)|_{-})(\xi/|\xi|)$$
 for x near 0.

Assume first that  $\mu_m = \mu_c$ . To find the first correction  $\omega_1$ , we multiply

$$\partial_x((1/\varepsilon_m)\partial_x u_1) + \omega_0^2 \mu_m u_1 = -2\omega_1 \omega_0 \mu_m u_0$$

by  $u_0$  and integrate over  $(a, -\rho/2)$  and  $(\rho/2, b)$  for  $\rho \ge \delta$ . Upon using the radiation conditions, integration by parts, and the asymptotics of  $u_1$  previously derived, we obtain that as  $\rho$  and  $\delta$  go to zero,

$$i\omega_1((u_0(a))^2 + (u_0(b))^2) - \frac{1}{\varepsilon_m}\alpha(\partial_x u_0(0))^2 = -2\omega_1\omega_0\mu_m$$

where the polarization  $\alpha$  is given by

$$\alpha = \gamma \partial_{\xi} v^{(1)}(1/2)|_{-} = 1 - \frac{\varepsilon_c}{\varepsilon_m}.$$
(6.5)

Therefore, we arrive at

$$\omega_1 = \frac{\alpha(\partial_x u_0(0))^2}{2\omega_0 \mu_m \varepsilon_m + \mathrm{i}\varepsilon_m ((u_0(a))^2 + (u_0(b))^2)}.$$
(6.6)

 $\mathbf{79}$ 

The term  $i\varepsilon_m((u_0(a))^2 + (u_0(b))^2)$  accounts for the effect of radiation on the shift of the scattering resonance  $\omega_0$ .

Now, if  $\mu_c \neq \mu_m$ , we need to compute the second-order corrector  $e_2$ . We have

$$\partial_{\xi}((1/\tilde{\varepsilon})\partial_{\xi}e_2) + \omega_0^2\tilde{\mu}e_0 = 0,$$

and

$$e_2(\xi) \sim \xi^2 \partial_x^2 u_0(0)/2 \quad \text{as } |\xi| \to +\infty.$$

here, the stretched coefficient  $\tilde{\mu}$  is equal to  $\mu_c$  in (-1/2, 1/2) and to  $\mu_m$  in  $(-\infty, -1/2) \cup (1/2, +\infty)$ .

From the equation satisfied by  $u_0$ , we obtain

$$\partial_x^2 u_0(0) = -\omega_0^2 \mu_m \varepsilon_m u_0(0).$$

Recall that  $e_0(\xi) = u_0(0)$  and let  $v^{(2)}$  be such that

$$\begin{cases} \partial_{\xi}((1/\tilde{\varepsilon}(\xi))\partial_{\xi}v^{(2)}(\xi)) = (1/(\varepsilon_{m}\mu_{m}))\tilde{\mu}(\xi), \\ v^{(2)}(\xi) \sim \xi^{2}/2 \quad \text{as } |\xi| \to +\infty. \end{cases}$$

It is easy to see that  $\partial_{\xi}((1/\tilde{\varepsilon}(\xi))\partial_{\xi}(v^{(2)}(\xi)-\xi^2/2))$  is  $(1/\varepsilon_m)((\mu_c/\mu_m)-1)$  for  $\xi \in (-1/2, 1/2)$  and is 0 for  $|\xi| > 1/2$ . Therefore,

$$v^{(2)}(\xi) - \xi^2/2 \sim ((\mu_c/\mu_m) - 1)|\xi| \text{ as } |\xi| \to +\infty.$$

Then, by matching the singularities in the inner and outer expansion as in [16, Section 4.1], we obtain

$$u_1(x) \sim \partial_x u_0(0)(\xi - ((\varepsilon_m/\varepsilon_c) - 1)\partial_\xi v^{(1)}(1/2)\xi/|\xi| + \dots) + \partial_x^2 u_0(0)((\mu_c/\mu_m) - 1)|\xi| + \dots$$

and so, proceeding as before,

$$i\omega_1((u_0(a))^2 + (u_0(b))^2) - \frac{1}{\varepsilon_m}\alpha(\partial_x u_0(0))^2 + \frac{1}{\varepsilon_m}\partial_x^2 u_0(0)((\mu_c/\mu_m) - 1)u_0(0)$$
  
=  $-2\omega_1\omega_0\mu_m$ ,

which yields the result.

#### 6.2 Multi-dimensional case

In this section, we generalize (6.2) to the multi-dimensional case. In dimension two, the obtained formula corresponds, as in the one-dimensional case, to an open cavity

with the transverse magnetic polarization [63]. We use the same notation as in Section 6.2.

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^d$  for d = 2, 3, with smooth boundary  $\partial \Omega$ . We assume  $\omega_0$  is a simple scattering resonance of the unperturbed open cavity, that is there exists a unique non trivial solution  $u_0$  to

$$\left\{ \begin{array}{l} \nabla \cdot \left( (1/\varepsilon) \nabla u \right) + \omega_0^2 \mu u = 0 \quad \text{in } \mathbb{R}^d, \\ \int_{\Omega} |u|^2 \, dx = 1, \\ u \text{ satisfies the Sommerfeld radiation condition,} \end{array} \right. \tag{6.7}$$

where  $\mu = 1 + (\mu_m - 1)\chi_{\Omega}$  and  $\varepsilon = 1 + (\varepsilon_m - 1)\chi_{\Omega}$ . Here,  $\chi_{\Omega}$  denotes the characteristic function of the domain  $\Omega$ .

For simplicity, we assume that  $\Omega$  is the ball of radius R centered at the origin, and introduce the capacity operator  $T_{\omega}$ , which is given by [20]

$$T_{\omega}:\phi = \begin{cases} \sum_{m\in\mathbb{Z}} \phi_m e^{im\theta} & \text{if } d = 2, \\ \sum_{m=0}^{+\infty} \sum_{l=-m}^{m} \phi_m^l Y_m^l & \text{if } d = 3, \end{cases} \mapsto \begin{cases} \sum_{m\in\mathbb{Z}} z_m(\omega, R) \phi_m e^{im\theta} & \text{if } d = 2, \\ \sum_{m=0}^{+\infty} z_m(\omega, R) \sum_{l=-m}^{m} \phi_m^l Y_m^l & \text{if } d = 3, \end{cases}$$

where

$$z_m(\omega, R) = \begin{cases} \frac{\omega(H_m^{(1)})'(\omega R)}{H_m^{(1)}(\omega R)} & \text{if } d = 2\\ \frac{\omega(h_m^{(1)})'(\omega R)}{h_m^{(1)}(\omega R)} & \text{if } d = 3. \end{cases}$$

Here,  $\theta$  is the angular variable,  $Y_m^l$  is a spherical harmonic, and  $H_m^{(1)}$  (respectively,  $h_m^{(1)}$ ) is the Hankel function of integer order (respectively, half-integer order). This explicit version of the capacity operator will be used in Section 6.4 to test the validity of our formula. Then, (6.7) is equivalent to

$$\begin{cases} (1/\varepsilon_m)\Delta u_0 + \omega_0^2 \mu u_0 = 0 & \text{in } \Omega, \\ (1/\varepsilon_m)\frac{\partial u_0}{\partial \nu} = T_{\omega_0}[u_0] & \text{on } \partial\Omega, \\ \int_{\Omega} |u_0|^2 = 1, \end{cases}$$
(6.8)

where  $\nu$  denotes the normal to  $\partial\Omega$ . As in the one-dimensional case, the scattering resonances lie in the lower-half of the complex plane and the associated eigenfunctions grow exponentially at large distances from the cavity since they satisfy the outgoing radiation condition. We also remark that from Green's identity we have

$$\int_{\partial\Omega} T_{\omega}[f]g \, d\sigma = \int_{\partial\Omega} fT_{\omega}[g] \, d\sigma \quad \text{for all } f, g \in \mathrm{H}^{1/2}(\partial\Omega), \tag{6.9}$$

for d = 2, 3.

Let  $D \Subset \Omega$  be a small particle of the form  $D = z + \delta B$ , where  $\delta$  is its characteristic size, z its location, and B is a smooth bounded domain containing the origin. Denote respectively by  $\varepsilon_c$  and  $\mu_c$  the electric permittivity and the magnetic permeability of the particle D. The eigenvalue problem is to find  $\omega_{\delta}$  such that there is a non-trivial couple ( $\omega_{\delta}, u_{\delta}$ ) satisfying

$$\begin{cases} (1/\varepsilon_m)\Delta u_{\delta} + \omega_{\delta}^2 \mu_m u_{\delta} = 0 & \text{in } \Omega \setminus \bar{D}, \\ (1/\varepsilon_c)\Delta u_{\delta} + \omega_{\delta}^2 \mu_c u_{\delta} = 0 & \text{in } D, \\ (1/\varepsilon_m)\frac{\partial u_{\delta}}{\partial \nu}\Big|_{+} = (1/\varepsilon_c)\frac{\partial u_{\delta}}{\partial \nu}\Big|_{-} & \text{on } \partial D, \\ (1/\varepsilon_m)\frac{\partial u_{\delta}}{\partial \nu} = T_{\omega_{\delta}}[u_{\delta}] & \text{on } \partial\Omega, \end{cases}$$

$$(6.10)$$

where the subscripts + and - indicate the limits from outside and inside D, respectively.

**Proposition 6.3.** Assuming  $\omega_{\delta} = \omega_0 + \delta^d \omega_1 + \dots$ , we have

$$\omega_1 = \frac{M(\varepsilon_m/\varepsilon_c, B)\nabla u_0(z) \cdot \nabla u_0(z) + \omega_0^2 |B|\varepsilon_m(\mu_c - \mu_m)(u_0(z))^2}{2\omega_0\mu_m\varepsilon_m \int_{\Omega} u_0^2 dx + \int_{\partial\Omega} \partial_\omega T_\omega|_{\omega=\omega_0} [u_0]u_0 d\sigma},$$
(6.11)

where M is the polarization tensor defined in (1.11).

*Proof.* Assume, for now, that  $\mu_c = \mu_m$ . Let  $\lambda_0 = \omega_0^2, \lambda_\delta = \omega_\delta^2$ . We expand

 $\omega_{\delta} = \omega_0 + \delta^d \omega_1 + \dots$  and  $\lambda_{\delta} = \lambda_0 + \delta^d \lambda_1 + \dots$ 

Let the outer expansion of  $u_{\delta}$  be

$$u_{\delta}(y) = u_0(y) + \delta^d u_1(y) + \dots,$$

and the inner one,  $e_{\delta}(\xi) = u_{\delta}((x-z)/\delta)$ , be

$$e_{\delta}(\xi) = e_0(\xi) + \delta e_1(\xi) + \dots$$

Therefore, formally we have

$$T_{\omega_{\delta}} \simeq T_{\omega_0 + \delta^d \omega_1} \simeq T_{\omega_0} + \delta^d \omega_1 \partial_{\omega} T_{\omega}|_{\omega_0} + \dots$$

Moreover, we obtain

$$\begin{cases} ((1/\varepsilon_m)\Delta + \lambda_0\mu_m)u_1(y) = -\lambda_1\mu_m u_0(y) & \text{for } |y-z| \gg O(\delta), \\ (1/\varepsilon_m)\frac{\partial u_1}{\partial \nu} = T_{\omega_0}[u_1] + \omega_1\partial_\omega T_\omega|_{\omega=\omega_0}[u_0] & \text{on } \partial\Omega, \end{cases}$$
(6.12)

and, by setting to zero the coefficients of  $1/\delta^2$  and  $1/\delta$  in  $e_{\delta}$  in (6.10),

$$\begin{cases} \Delta_{\xi} e_j = 0 & \text{in } \mathbb{R}^d \setminus (\partial B), \\ \frac{\partial e_j}{\partial \nu}|_+ = (\varepsilon_m / \varepsilon_c) \frac{\partial e_j}{\partial \nu}|_- & \text{on } \partial B, \end{cases}$$

for j = 0, 1. Imposing the matching conditions

$$u_0(y) + \delta^d u_1(y) + \dots \sim e_0(\xi) + \delta e_1(\xi) + \dots$$
 as  $|\xi| \to +\infty$ ,

and  $y \to z$ , we arrive at  $e_0(\xi) \to u_0(z)$  and  $e_1(\xi) \sim \nabla u_0(z) \cdot \xi$ . So, we have  $e_0(\xi) = u_0(z)$  for every  $\xi$  and  $e_1(\xi) = \nabla u_0(z) \cdot v^{(1)}(\xi)$ , where  $v^{(1)}$  is such that (see [16, (4.11)])

$$\begin{cases} \Delta_{\xi} v^{(1)} = 0 & \text{in } \mathbb{R}^{d} \setminus (\partial B), \\ \frac{\partial v^{(1)}}{\partial \nu}|_{+} = (\varepsilon_{m}/\varepsilon_{c}) \frac{\partial v^{(1)}}{\partial \nu}|_{-} & \text{on } \partial B, \\ v^{(1)}(\xi) \sim \xi & \text{as } |\xi| \to +\infty. \end{cases}$$
(6.13)

Let  $M(\varepsilon_m/\varepsilon_c, B)$  be the polarization tensor associated with the domain B and the contrast  $\varepsilon_m/\varepsilon_c$ . Then, by the same arguments as in [16, Section 4.1], it follows that

$$u_1(y) \sim -M(\varepsilon_m/\varepsilon_c, B)\nabla\Gamma(y-z)\cdot\nabla u_0(z)$$
 as  $y \to z$ . (6.14)

Multiplying (6.12) by  $u_0$  and integrating by parts over  $\Omega \setminus \bar{B}_{\delta}$ , we obtain from (6.9), the identity  $(-1/\varepsilon_m)\Delta u_0 = \lambda_0 \mu_m u_0$ , that

$$-\lambda_{1}\mu_{m}\int_{\Omega\setminus B_{\delta}}(u_{0})^{2} dx = \underbrace{\int_{\partial\Omega} \left(T_{\omega_{0}}[u_{1}]u_{0} - T_{\omega_{0}}[u_{0}]u_{1}\right) d\sigma}_{=0} + \underbrace{\frac{1}{\varepsilon_{m}}\int_{\partial B_{\delta}}(u_{0}\frac{\partial u_{1}}{\partial\nu} - u_{1}\frac{\partial u_{0}}{\partial\nu}) d\sigma}_{=0}.$$

From (6.14), we have

$$\int_{\partial B_{\delta}} \left( u_0 \frac{\partial u_1}{\partial \nu} - u_1 \frac{\partial u_0}{\partial \nu} \right) d\sigma \xrightarrow[\delta \to 0]{} - M(\varepsilon_m / \varepsilon_c, B) \nabla u_0(z) \cdot \nabla u_0(z)$$

Therefore, since  $\lambda_1 = 2\omega_0\omega_1$ ,

$$-\lambda_1 \mu_m \varepsilon_m \int_{\Omega} u_0^2 \, dx - \frac{\lambda_1}{2\omega_0} \int_{\partial\Omega} \partial_\omega T_\omega |_{\omega=\omega_0} [u_0] u_0 \, d\sigma = -M(\varepsilon_m/\varepsilon_c, B) \nabla u_0(z) \cdot \nabla u_0(z),$$

and finally, we arrive at

$$\lambda_1 = \frac{M(\varepsilon_m/\varepsilon_c, B)\nabla u_0(z) \cdot \nabla u_0(z)}{\varepsilon_m \mu_m \int_{\Omega} u_0^2 dx + (1/(2\omega_0)) \int_{\partial\Omega} \partial_\omega T_\omega|_{\omega=\omega_0} [u_0] u_0 d\sigma},$$
(6.15)

or equivalently,

$$\omega_1 = \frac{M(\varepsilon_m/\varepsilon_c, B)\nabla u_0(z) \cdot \nabla u_0(z)}{2\omega_0 \mu_m \varepsilon_m \int_{\Omega} u_0^2 dx + \int_{\partial\Omega} \partial_\omega T_\omega|_{\omega=\omega_0} [u_0] u_0 d\sigma}.$$
(6.16)

If one relaxes the assumption  $\mu_c = \mu_m$ , one can generalize formula (6.16) by computing, as in [16] and in Section 6.1, the second-order corrector  $e_2$ . We then get the desired result.

The expression derived in Proposition 6.3 tells us that in the multi-dimensional case, the effect of radiation on the shift of the scattering resonance  $\omega_0$  is given by  $\int_{\partial\Omega} \partial_{\omega} T_{\omega}|_{\omega=\omega_0} [u_0] u_0 \, d\sigma$ . Note also that (6.16) reduces to (6.6) in the one-dimensional case. In fact, the polarization tensor M reduces to  $\alpha$  defined by (6.5) and the operator  $T_{\omega}$  corresponds to the multiplication by  $-i\omega\varepsilon_m$  at a and  $+i\omega\varepsilon_m$  at b.

# 6.3 Perturbation of whispering-gallery modes by an external particle

Whispering-gallery modes are modes which are confined near the boundary of the cavity. Their existence for spherical resonators can be proved analytically or by a boundary layer approach based on WKB (high frequency) asymptotics [82, 68, 72, 75, 45, 84, 1]. Whispering-gallery modes can be exploited to probe the local surroundings of the resonator [65, 66, 83]. Biosensors based on the shift of whispering-gallery modes in open cavities by small particles have been also described by use of Bethe-Schwinger type formulas, where the effect of radiation is neglected [26, 45, 98, 99]. In this section, we provide a generalization of the formula derived in the previous section and discuss its validity for whispering-gallery modes.

Assume that  $\omega_0$  is a whispering-gallery mode of the open cavity  $\Omega$ . Let  $\Omega_{\rho}$  be a small neighborhood of  $\Omega$ . Suppose that the particle D is in  $\Omega_{\rho} \setminus \overline{\Omega}$ . If the characteristic size  $\delta$  of D is much smaller than  $\rho$ , which is in turn much smaller than  $2\pi/(\sqrt{\varepsilon_m \mu_m} \omega_0)$ , then by the same arguments as those in the previous section, the leading-order term in the shift of the resonant frequency  $\omega_0$  is given by

$$\omega_1 \simeq \frac{M(1/\varepsilon_c, B)\nabla v_0(z) \cdot \nabla v_0(z) + \omega_0^2 |B|(\mu_c - 1)(v_0(z))^2}{2\omega_0 \mu_m \varepsilon_m \int_{\Omega} u_0^2 dx + \int_{\partial \Omega} \partial_\omega T_\omega|_{\omega = \omega_0} [u_0] u_0 d\sigma}.$$

Here, the polarization tensor  $M(\varepsilon_m/\varepsilon_c, B)$  in (6.15) is replaced by  $M(1/\varepsilon_c, B)$  since

 $\varepsilon$  in the medium surrounding the particle is equal to 1 and  $v_0$  is defined in  $\mathbb{R}^d$  by

$$v_0(x) = -\omega_0^2(\mu_m - 1) \int_{\Omega} \Gamma(x - y; \omega_0) u_0(y) \, dy + \left(\frac{1}{\varepsilon_m} - 1\right) \int_{\Omega} \nabla_y \Gamma(x - y; \omega_0) \cdot \nabla u_0(y) \, dy,$$
(6.17)

where  $\Gamma(\cdot; \omega_0)$  is the fundamental solution of  $\Delta + \omega_0^2$ , which satisfies the outgoing radiation condition. We remark that  $v_0 = u_0$  in  $\Omega$ . Moreover, the assumption that  $\omega_0$  is a whispering-gallery mode is needed in order to have the gradient of  $v_0$  at the location of the particle to have a significant magnitude.

Now, assume that the particle D is plasmonic, i.e.,  $\varepsilon_c$  depends on the frequency  $\omega$  and can take negative values. In this case, there is a discrete set of frequencies, called plasmonic resonant frequencies, such that at these frequencies problem (6.13) is nearly singular, and therefore the polarization tensor associated with the particle D blows up at those frequencies, see [8, 18, 21]. Assume that the plasmonic particle is coupled to the cavity, i.e., there is a whispering-gallery cavity mode  $\omega_0$  such that  $\Re \omega_0$  is a plasmonic resonance of the particle.

Then when the particle D is illuminated at the frequency  $\Re \omega_0$ , its effect on the cavity mode  $\omega_0$  is given by

$$\omega_1 \simeq \frac{M((1/\varepsilon_c)(\Re\omega_0), B)\nabla v_0(z) \cdot \nabla v_0(z) + \omega_0^2 |B|(\mu_c - 1)(v_0(z))^2}{2\omega_0\mu_m\varepsilon_m \int_{\Omega} u_0^2 dx + \int_{\partial\Omega} \partial_\omega T_\omega|_{\omega=\omega_0} [u_0]u_0 d\sigma},$$
(6.18)

where  $v_0$  is defined by (6.17). Thus, despite their small size, plasmonic particles can significantly change the cavity modes when their plasmonic resonances are close to the cavity modes.

Finally, let us briefly consider the case when there are multiple linear independent solutions to (6.7). This situation has been studied in Chapter 4 for closed cavities, under the much simpler setting of a self-adjoint operator. Although the extension of the formulae derived in this simpler setting to open resonators would require a finer analysis, we believe that similar expressions still hold. In fact, in the next section we provide numerical evidence to the fact that, if there are two orthogonal solutions  $u_1, u_2$  to (6.7), then the largest and smallest resonance perturbations are indeed proportional respectively to the largest and smallest eigenvalues of the matrix

$$\left(M((1/\varepsilon_c)(\Re\omega_0), B)\nabla u_i(z) \cdot \nabla u_j(z)\right)_{i,j\in\{1,2\}}.$$
(6.19)

#### 6.4 Numerical illustrations

In two dimensions, when the cavity and the small-volume particle are disks we can use the multipole expansion method to efficiently compute the perturbations of the whispering-gallery modes [74]. Our approach is as follows. We first use a projective eigensolver [29] to obtain a coarse estimate of the locations of the resonances of a two disk system. We then focus on the particular resonances in this set that correspond to the whispering-gallery modes of the open cavity and obtain a refined estimate of their locations using the multipole method.

Throughout this section,  $\Omega$  is a disk of radius 1 centered at the origin and  $\omega_0$  is the frequency of a whispering-gallery mode. Let D be a disk of radius  $\delta$  centered at  $(1 + 2\delta, 0)$ . Suppose that  $\varepsilon_m = \varepsilon_c = 1/5$ . The behavior of  $\omega_{\delta,1}, \omega_{\delta,2}$  as  $\delta \to 0$  is plotted in Figure 6.1. The characterization of multiple resonances given in (6.19) matches the first resonance shift, as can be seen in Figure 6.2. On the other hand, we can easily reconstruct  $\delta$  from a single scattering resonance shift.



Figure 6.1: As the size of the small disk  $\delta \to 0$ , the perturbed whispering-gallery modes  $\omega_{\delta,1}$  and  $\omega_{\delta,2}$  converge towards the unperturbed mode  $\omega_0$ .

Next, consider a disk  $D_{\delta}$  of radius  $\delta = 0.1$  centered at (z, 0). A plot of  $|\omega_{\delta,j}^2 - \omega_0^2|$  as z varies between 1.2 and 6 is presented in Figure 6.3. We remark that the again



Figure 6.2: Comparison between the asymptotic formula for the perturbation  $|\omega_{\delta,1}^2 - \omega_0^2|$  of the whispering-gallery mode and the perturbation computed numerically as the size of the small disk  $\delta \to 0$ .

characterization of multiple resonances given in (6.19) matches closely the numerical experiments.



Figure 6.3: Comparison between the asymptotic formula for the perturbation  $|\omega_{\delta,j}^2 - \omega_0^2|$  of the whispering-gallery mode and the perturbation computed numerically as the position of the inclusion (z, 0) varies. The plot on the left corresponds to the perturbed resonance  $\omega_{\delta,1}$  and the plot on the right corresponds to the perturbed resonance  $\omega_{\delta,2}$ .

We highlight now the case of plasmonic particles. In this case we have a strong enhancement in the frequency shift, which allows for the recognition of much smaller particles. Consider a disk D of radius 0.1 centered at (1.2, 0). Suppose  $\varepsilon_m = 1/5$ . A plot of numerical experiments for the shift  $|\omega_{\delta,1}^2 - \omega_0^2|$  as  $1/\varepsilon_c$  varies is presented in Figure 6.4. Notice the high peak in the perturbation as  $\varepsilon_c$  approaches the value -1.

Finally, let us remark that our formulae also allow quick estimates of the polarization tensor of the small particle. Thus the orientation of the perturbing particle can be inferred, which affords the possibility of orientational binding studies in biosensing. We also believe that, based on [23, 13], the derived formulae can be generalized to open electromagnetic and elastic cavities.



Figure 6.4: Resonance perturbation  $|\omega_{\delta,1}^2 - \omega_0^2|$  as a function of  $1/\varepsilon_c$ , here allowed to also take negative values.

## Chapter 7

# Perturbation of scattering resonances - transverse electric polarization case

In this chapter we consider dielectric radiating open cavities [27, 34, 94] and rigorously obtain asymptotic formulae for the shift in the scattering resonances that is caused by a small particle of arbitrary shape. Our formula shows that the perturbation of the scattering resonances can be expressed in terms of the polarization tensor of the small particle. The scattering resonances can be degenerate or even exceptional and the small particle can be plasmonic. Our method is based on polepencil decompositions of the volume integral operator associated with the radiating dielectric cavity problem. The new techniques introduced in this chapter can not be easily extended to the transverse magnetic case considered in Chapter 6, mainly due to the hyper-singular character of the associated volume-integral operator.

The chapter is organized as follows. In Section 7.1, we characterize the scattering resonances of dielectric cavities in terms of the spectrum of a volume integral operator. In Section 7.2, using the method of pole-pencil decompositions (see, for instance, [9, 14]), we derive the leading-order term in the shift of the scattering resonances of an open dielectric cavity due to presence of internal particles. In Section 7.3, using a Lippmann-Schwinger representation formula for the Green's function associated with the open cavity, we generalize the formula obtained in Section 7.2 to the case of external particles. In Section 7.4, we consider the perturbation of an open dielectric cavity by plasmonic nanoparticles. The formula obtained for the perturbation of the frequencies shows a strong enhancement in the frequency shift in the case of plasmonic nanoparticles. In Section 7.5, we examine the case of exceptional points and reformulate the search for scattering resonances as a search for the roots of a polynomial.

#### 7.1 Scattering resonances of a dielectric cavity

Let  $\Omega$  be a domain in  $\mathbb{R}^d$  for d = 2, 3, with smooth boundary  $\partial\Omega$ . Consider the operator  $\Delta + \varepsilon \mu_m \omega^2$  where  $\omega \in \mathbb{C}$  is the frequency,  $\varepsilon \equiv \tau \varepsilon_c + \varepsilon_m$  inside  $\Omega$  and  $\varepsilon \equiv \varepsilon_m$  outside  $\Omega$ . Here,  $\varepsilon_c, \varepsilon_m, \mu_m$  and  $\tau$  are positive constants.

Let  $\Gamma_m$  be the outgoing (i.e., subject to the Sommerfeld radiation condition (1.5)) fundamental solution of  $\Delta + \varepsilon_m \mu_m \omega^2$  in free space, and let G be the outgoing fundamental solution of  $\Delta + \varepsilon \mu_m \omega^2$  in free space. Let

$$K_{\Omega}^{\omega}: u \in \mathcal{L}^{2}(\Omega) \mapsto -\int_{\Omega} u(y)\Gamma_{m}(\cdot, y; \omega)dy \in \mathcal{L}^{2}(\Omega).$$

The operator  $K_{\Omega}^{\omega} : L^2(\Omega) \to L^2(\Omega)$  is of Hilbert-Schmidt type, and thus compact. Therefore, its spectrum is

$$\sigma(K_{\Omega}^{\omega}) = \{0, \lambda_1(\omega), \lambda_2(\omega), \dots, \lambda_j(\omega), \dots\},\$$

where  $|\lambda_j(\omega)| \geq |\lambda_{j+1}(\omega)|$ ,  $\lambda_j(\omega) \to 0$  as  $j \to +\infty$  and  $\{0\} = \sigma(K_{\Omega}^{\omega}) \setminus \sigma_p(K_{\Omega}^{\omega})$ with  $\sigma_p(K_{\Omega}^{\omega})$  being the point spectrum. Note that, due to the uniqueness of the exterior Helmholtz problem, the imaginary part  $\Im \lambda_j(\omega)$  must be non-zero for all jand  $\omega \in \mathbb{R}$ . Let  $H_j$  be the generalized eigenspace associated with  $\lambda_j(\omega)$ . Then, from [24, Lemma 3.5], we have that  $L^2(\Omega)$  is the closure of the union of all generalized eigenspaces  $H_j$  in L<sup>2</sup>-norm.

Let  $\omega_0$  and  $\tau$  be such that there exists a  $j_0$  such that

$$1 - \omega_0^2 \tau \varepsilon_c \mu_m \lambda_{j_0}(\omega_0) = 0. \tag{7.1}$$

We call such an  $\omega_0 \in \mathbb{C}$  (with  $\Im \omega_0 \neq 0$ ) a scattering resonance of the open dielectric cavity  $\Omega$ . Moreover, for  $\tau$  large enough, (7.1) has solutions. Furthermore, assume that for  $\omega$  in a complex neighborhood of  $\omega_0$ , the following assumptions hold:

(i) We have

$$1 - \omega^2 \tau \varepsilon_c \mu_m \lambda_{j_0}(\omega) = R(\omega)(\omega - \omega_0),$$

where  $R(\omega_0) \neq 0$  and  $\omega \mapsto R(\omega)$  is analytic;

(ii) The generalized eigenspace  $H_{j_0}(\omega)$  is of dimension 1.

If both of these assumptions are satisfied, then we say that the scattering resonance  $\omega_0$  is non-exceptional.

Assume that  $\omega_0$  is a non-exceptional scattering resonance. As a consequence of classic expansion theorems for fundamental solutions (see [37, *expansion theorems*]) we have that there exists a complex neighborhood  $V(\omega_0)$  of  $\omega_0$  such that for  $\omega$  in  $V(\omega_0) \setminus \{\omega_0\}$ ,

$$G(x, y; \omega) = \Gamma_m(x, y; \omega) + c_{j_0}(\omega) \frac{e_{j_0}(x; \omega)e_{j_0}(y; \omega)}{\omega - \omega_0}$$
(7.2)

+ some function that is smooth in x, y and analytic in  $\omega$ ,

where  $e_{j_0}$  spans  $H_{j_0}$ ,  $||e_{j_0}||_{L^2(\Omega)} = 1$ , and  $\omega \mapsto e_{j_0}(\cdot, \omega)$  and  $\omega \mapsto c_{j_0}(\omega)$  are analytic in  $V(\omega_0)$ . Note that  $e_{j_0}(y; \omega_0)$  is the restriction to  $\Omega$  of the eigenmode associated with the scattering resonance  $\omega_0$ .

#### 7.2 Shift of resonances by internal particles

Let D be a domain compactly contained in  $\Omega$ , such that  $D = z + \delta B$ , where  $\delta$  is the characteristic size of D, z is its location, and B is a smooth bounded domain containing the origin. We suppose that D has a different magnetic permeability from  $\mu_m$ , and consider the operator

$$abla \cdot rac{1}{\mu} 
abla + arepsilon \omega^2,$$

where  $\mu = \mu_c$  in D and  $\mu = \mu_m$  outside D.

As  $\delta \to 0$ , we seek an  $\omega_{\delta}$  in a neighborhood of  $\omega_0$  such that there exists a non-trivial solution to

$$\left(\nabla \cdot \frac{1}{\mu} \nabla + \varepsilon \omega_{\delta}^2\right) u = 0,$$

subject to the Sommerfeld radiation condition. Integrating by parts and using the radiation condition, it is easy to show that the solution to the above problem admits the following Lippmann-Schwinger representation formula:

$$u(x) = \gamma \int_D \nabla u(y) \cdot \nabla G(x, y; \omega_{\delta}) \, dy \quad \text{for all } x \in \mathbb{R}^d,$$

where  $\gamma = (1/\mu_c) - (1/\mu_m)$ . Consider the volume integral operator

$$T_D^{\omega}: v \mapsto \nabla_x \int_D v(y) \cdot \nabla G(x, y; \omega) \, dy,$$

91

where the integral on the right hand side is intended in the principal value sense. Then the operator  $T_D^{\omega}$ :  $L^2(D)^d \to L^2(D)^d$  is well-defined (see, for instance, [17, Appendix B] and references therein).

We seek  $\omega_{\delta}$  such that there is a non-trivial  $v \in L^2(D)^d$  satisfying

$$\left(I - \gamma T_D^{\omega_\delta}\right)[v] = 0, \tag{7.3}$$

where I denotes the identity operator. Hence, as the characteristic size  $\delta$  of D goes to zero, we seek  $\omega_{\delta}$  in a neighborhood of  $\omega_0$  such that  $1/\gamma$  is an eigenvalue of  $T_D^{\omega_{\delta}}$ .

From the pole-pencil decomposition (7.2) of G, we have

$$\nabla \int_D v \cdot \nabla G = \nabla \int_D v \cdot \nabla \Gamma_m + \frac{c_{j_0}(\omega)}{\omega - \omega_0} \Big( \int_D v \cdot \nabla e_{j_0} \, dy \Big) \nabla e_{j_0}(x;\omega) + R[v],$$

where  $R : L^2(D)^d \to L^2(D)^d$  is an operator with smooth kernel that is analytic in  $\omega \in V(\omega_0)$ . Let

$$N_D^{\omega}: v \mapsto \nabla_x \int_D v(y) \cdot \nabla \Gamma_m(x, y; \omega) dy$$

Then, it follows that

$$(I/\gamma - T_D^{\omega})[v] = (I/\gamma - N_D^{\omega})[v] - \frac{c_{j_0}(\omega)}{\omega - \omega_0}(v, \nabla e_{j_0})\nabla e_{j_0} + R[v],$$

where  $(\cdot, \cdot)$  denotes the L<sup>2</sup> real scalar product on D and I is the identity operator.

From [36, 46], it is known that  $N_D^{\omega}|_W : W \longrightarrow W$  is a compact operator, where W is the space of gradients of harmonic functions in D. Moreover, the spectrum of  $N_D^{\omega}|_W$  is discrete and the associated eigenfunctions form a basis of W.

Let  $L = I/\gamma - N_D^{\omega=0}$ . Then, (7.3) can be rewritten as

$$L[v] - \frac{c_{j_0}(\omega)}{\omega - \omega_0} (v, \nabla e_{j_0}) \nabla e_{j_0} + \widetilde{R}[v] = 0,$$

where  $\widetilde{R} : L^2(D)^d \to L^2(D)^d$  is an operator with smooth kernel that is analytic in  $\omega \in V(\omega_0)$ . Therefore, since for  $\mu_c$  large enough L is invertible,

$$v - \frac{c_{j_0}(\omega)}{\omega - \omega_0} (v, \nabla e_{j_0}) L^{-1}[\nabla e_{j_0}] + L^{-1} \widetilde{R}[v] = 0.$$

So, since

$$||L^{-1}\widetilde{R}||_{\mathcal{L}(\mathcal{L}^2(D)^d,\mathcal{L}^2(D)^d)} = o(1) \quad \text{as } \delta \to 0,$$

(see [17, Lemma 4.2] and [14]), the term  $L^{-1}\widetilde{R}[v]$  can be neglected, and we have the leading-order asymptotic approximation

$$\omega_{\delta} - \omega_0 \sim c_{j_0}(\omega_0) (L^{-1}[\nabla e_{j_0}], \nabla e_{j_0}).$$

Moreover, from [17, Proposition 3.1], it follows that

$$(L^{-1}[\nabla e_{j_0}], \nabla e_{j_0}) \sim \delta^d M(\mu_m/\mu_c, B) \nabla e_{j_0}(z; \omega_0) \cdot \nabla e_{j_0}(z; \omega_0),$$

where M is the polarization tensor given by (1.11).

In conclusion, the following proposition holds.

**Proposition 7.1.** As  $\delta \to 0$ , we have

$$\omega_{\delta} - \omega_0 \sim \delta^d c_{j_0}(\omega_0) M(\mu_m/\mu_c, B) \nabla e_{j_0}(z; \omega_0) \cdot \nabla e_{j_0}(z; \omega_0).$$

$$(7.4)$$

#### 7.3 Shift of resonances by external particles

Now consider the case where the particle is outside  $\Omega$ . Again by integration by parts and the radiation conditions, we obtain the following Lippmann-Schwinger representation formula for G:

$$G(x,z;\omega) = \Gamma_m(x,z;\omega) - \omega^2 \tau \varepsilon_c \mu_m \int_{\Omega} G(z,z';\omega) \Gamma_m(z',x;\omega) \, dz', \qquad (7.5)$$

for  $x, z \in \mathbb{R}^d, x \neq z$ . On the other hand, for x outside  $\Omega$  and z inside  $\Omega$ , using (7.2) we obtain,

$$G(x,z;\omega) = \Gamma_m(x,z;\omega) - \frac{\omega^2 \tau \varepsilon_c \mu_m c_{j_0}(\omega)}{\omega - \omega_0} e_{j_0}(z;\omega) \left( \int_{\Omega} e_{j_0}(z';\omega) \Gamma_m(z',x;\omega) \, dz' \right)$$

+ a function that is smooth in x, z, and analytic in  $\omega$ .

(7.6)

For x, z both outside  $\Omega, x \neq z$ , plugging (7.6) back into (7.5), we arrive at

$$G(x, z; \omega) = \Gamma_m(x, z; \omega) + \frac{c_{j_0}(\omega)}{\omega - \omega_0} g_{j_0}(z; \omega) g_{j_0}(x; \omega)$$

+ a function that is smooth in x, z, and analytic in  $\omega$ ,

where we defined

$$g_{j_0}(\ \cdot\ ;\omega) = \omega^2 \tau \varepsilon_c \mu_m \int_{\Omega} e_{j_0}(z';\omega) \Gamma_m(z',\ \cdot\ ;\omega) \, dz'$$

Note that  $g_{j_0}$ , defined by the above formula for all  $x \in \mathbb{R}^d$ , is the eigenmode associated with the scattering resonance  $\omega_0$ , and it coincides with  $e_{j_0}(x; \omega_0)$  on  $\Omega$ .

Analogously to the calculations in the previous section, we have

$$v - \frac{c_{j_0}(\omega)}{\omega - \omega_0} (v, \nabla g_{j_0}) L^{-1}[\nabla g_{j_0}] + L^{-1}R[v] = 0$$

for some operator R with smooth kernel that is analytic in  $\omega$  in  $V(\omega_0)$ . Therefore, analogously to (7.4), the following asymptotic expansion holds. **Proposition 7.2.** As  $\delta \to 0$ , we have

$$\omega_{\delta} - \omega_0 \sim \delta^d c_{j_0}(\omega_0) M(\mu_m/\mu_c, B) \nabla g_{j_0}(z; \omega_0) \cdot \nabla g_{j_0}(z; \omega_0).$$

$$(7.7)$$

#### 7.4 Shift of resonances by plasmonic particles

Suppose now that D is a plasmonic particle, i.e.,  $\mu_c$  depends on  $\omega$  and for a discrete set of frequencies  $\omega$ , called plasmonic resonances, the polarization tensor is singular, see [8, 18, 21]. In this case, the scattering resonance problem consists in finding  $\omega$ such that there is a non-trivial solution v to

$$L(\omega)[v] - \frac{c_{j_0}(\omega)}{\omega - \omega_0} (v, \nabla e_{j_0}) \nabla e_{j_0} + R[v] = 0,$$

where  $L(\omega) = I/\gamma(\omega) - N_D^{\omega=0}$ . The operator L is not invertible at plasmonic resonances. Using the Drude model for the permeability, we have  $\mu_c(\omega) = \mu_m(1-\omega_p^2/\omega^2)$ , where  $\omega_p$  is the volume plasma frequency. Let  $(\lambda_j, \varphi_j)$  be an eigenvalue-eigenfunction pair of L. Then

$$v - \frac{c_{j_0}(\omega)}{\omega - \omega_0} \frac{(v, \nabla e_{j_0})(\nabla e_{j_0}, \varphi_j)\varphi_j}{\lambda(\omega) - \lambda_j} = 0,$$

and thus,

$$1 - \frac{c_{j_0}(\omega_0)}{\omega - \omega_0} \frac{(\nabla e_{j_0}, \varphi_j)^2}{\lambda(\omega) - \lambda_j} = 0.$$

So, we arrive at the following proposition.

**Proposition 7.3.** As  $\delta \to 0$ , we have

$$(\omega_{\delta} - \omega_0)(\lambda(\omega_{\delta}) - \lambda_j) \sim c_{j_0}(\omega_0)(\nabla e_{j_0}, \varphi_j)^2.$$

Note that if  $\lambda(\omega) - \lambda_j = O(\omega - \omega_0)$  for  $\omega$  close to  $\omega_0$ , then we obtain

$$(\omega_{\delta} - \omega_0)^2 \sim c_{j_0}(\omega_0) (\nabla e_{j_0}(\cdot;\omega_0),\varphi_j)^2,$$

Hence, we have a significant shift in the scattering resonances if the particle D is plasmonic and resonant near or at the frequency  $\omega_0$ . This anomalous effect has been observed in [92].

#### 7.5 Asymptotic analysis near exceptional points

In this section, we examine the asymptotic behavior of an exceptional scattering resonance for a particular form of the Green function. These exceptional resonances are due to the non-Hermitian character of the operator  $T_D^{\omega}$ , see [24, 88]. For simplicity and in view of the Jordan-type decomposition of the operator  $T_D^{\omega}$  established in [24], we assume that, for  $\omega$  near  $\omega_0$ ,  $G(x, y; \omega)$  behaves like

$$G(x, y; \omega) = \Gamma_m(x, y; \omega) + c_1(\omega) \frac{h^{(1)}(x; \omega)h^{(1)}(y; \omega)}{\omega - \omega_0} + c_2(\omega) \frac{h^{(2)}(x; \omega)h^{(2)}(y; \omega)}{(\omega - \omega_0)^2} + R(\omega),$$
(7.8)

for two functions  $h^{(1)}$  and  $h^{(2)}$  in  $L^2(D)$ . In this simple case, we characterize the shift of the scattering resonance  $\omega_0$  due to the small particle D, which is assumed for simplicity to be non-plasmonic.

Following the same arguments as those in the previous sections, we seek a nontrivial v such that

$$L[v] - c_1(\omega) \frac{(v, \nabla h^{(1)})}{\omega - \omega_0} \nabla h^{(1)} - c_2(\omega) \frac{(v, \nabla h^{(2)})}{(\omega - \omega_0)^2} \nabla h^{(2)} = 0,$$

or equivalently,

$$v - c_1(\omega) \frac{(v, \nabla h^{(1)})}{\omega - \omega_0} L^{-1}[\nabla h^{(1)}] - c_2(\omega) \frac{(v, \nabla h^{(2)})}{(\omega - \omega_0)^2} L^{-1}[\nabla h^{(2)}] = 0.$$

By multiplying the above equation by  $\nabla h^{(1)}$  and  $\nabla h^{(2)}$ , respectively, and integrating by parts over D, we obtain the following system of equations:

$$\begin{cases} (v, \nabla h^{(1)}) \left( 1 - c_1(\omega) \frac{(L^{-1}[\nabla h^{(1)}], \nabla h^{(1)})}{\omega - \omega_0} \right) = c_2(\omega)(v, \nabla h^{(2)}) \frac{(L^{-1}[\nabla h^{(2)}], \nabla h^{(1)})}{(\omega - \omega_0)^2}, \\ (v, \nabla h^{(2)}) \left( 1 - c_2(\omega) \frac{(L^{-1}[\nabla h^{(2)}], \nabla h^{(2)})}{(\omega - \omega_0)^2} \right) = c_1(\omega)(v, \nabla h^{(1)}) \frac{(L^{-1}[\nabla h^{(1)}], \nabla h^{(2)})}{\omega - \omega_0}. \end{cases}$$

Therefore, the following result holds.

**Proposition 7.4.** Assume that the decomposition (7.8) holds for  $\omega$  near  $\omega_0$ . Then the perturbed scattering resonance problem (due to the particle D) can be reformulated as a search for  $\omega$  near  $\omega_0$  such that the matrix

$$\begin{pmatrix} 1 - c_1(\omega) \frac{(L^{-1}[\nabla h^{(1)}], \nabla h^{(1)})}{\omega - \omega_0} & -c_2(\omega) \frac{(L^{-1}[\nabla h^{(2)}], \nabla h^{(1)})}{(\omega - \omega_0)^2} \\ c_1(\omega) \frac{(L^{-1}[\nabla h^{(1)}], \nabla h^{(2)})}{\omega - \omega_0} & 1 - c_2(\omega) \frac{(L^{-1}[\nabla h^{(2)}], \nabla h^{(2)})}{(\omega - \omega_0)^2} \end{pmatrix}$$

is singular.

In conclusion, in this chapter we derived the leading-order term in the shift of scattering resonances of a radiating dielectric cavity due to the presence of small particles. We were able also to give a characterization of the shift due to small particles near an exceptional scattering resonance in a specific case. However, developing a general theory near exceptional frequencies remains a challenging open problem.

## Chapter 8

# Scattering of highly refractive particles

Nanoscale optics is usually associated with plasmonic resonant structures made of metals such as gold or silver. Plasmonic resonances of nanoparticles can be treated as an eigenvalue problem for the Neumann-Poincaré operator, see [8, 17, 18, 21]. However, plasmonic structures suffer from the high losses inherent in metals and dissipation due to heating. Recent developments in nanoscale optical physics have led to a new branch of nanophotonics focused on the manipulation of optically induced subwavelength resonances in dielectric nanoparticles with high refractive indices [67, 102, 103]. Resonant high-index dielectric nano-structures form new building blocks which can be used to realize unique functionalities and novel photonic devices [67]. Their study has been established as a new research direction in nanophotonics. Nevertheless, despite strong experimental efforts, mathematical modeling of resonant high-index nanoparticles remains limited.

In this chapter, we consider a dielectric high-index nanoparticle of arbitrary shape and characterize its subwavelength resonances in terms of the eigenvalues of the Newtonian potential associated with its shape. Our formula is closely related to the one established in [78]. Then, we provide an asymptotic formula for the field scattered by a dielectric nanoparticle and estimate the scattering enhancement near its resonant frequencies. We also consider the hybridization phenomenon of a dimer consisting of high refractive index dielectric nanoparticles.

Our results in this chapter provide a new mathematical framework for the analysis of resonant dielectric nanoparticles. They allow quick computations of estimates of resonant frequencies and the design of dielectric nanoparticles that resonate at specified frequencies. They can also be applied in the design of dielectric metamaterials [61, 102, 103].

#### 8.1 Derivation of asymptotic estimates

Let D be a domain contained in  $\mathbb{R}^d$ , for d = 2 or 3, such that  $D = z + \delta B$ , where  $\delta$  is its characteristic size, z its location, and B is a smooth bounded domain containing the origin. Let  $\omega$  denote the frequency, let  $\varepsilon \equiv \tau \varepsilon_c + \varepsilon_m$  inside D and  $\varepsilon \equiv \varepsilon_m$  outside D. Here,  $\varepsilon_c, \varepsilon_m$ , and  $\tau$  are positive constants. Let  $E^{\text{in}}$  be an incident plane wave with frequency  $\omega$ .

Consider the Helmholtz equation

$$\begin{cases} (\Delta + \omega^2 \varepsilon) E = 0 & \text{in } \mathbb{R}^d, \\ E - E^{\text{in}} \text{ satisfies the Sommerfeld radiation condition.} \end{cases}$$

From

$$(\Delta + \omega^2 \varepsilon_m)(E - E^{\rm in}) = -\omega^2 \tau \varepsilon_c E \chi_D$$
 in  $\mathbb{R}^d$ ,

where  $\chi_D$  is the characteristic function of D, it follows that the following Lippman-Schwinger representation formula holds:

$$(E - E^{\rm in})(x) = -\omega^2 \tau \varepsilon_c \int_D E(y) \Gamma_m(x, y; \omega) dy \quad \text{for } x \in \mathbb{R}^d, \tag{8.1}$$

where  $\Gamma_m$  is the outgoing (i.e., subject to the Sommerfeld radiation condition) fundamental solution of  $\Delta + \varepsilon_m \omega^2$  in free space.

Let the volume integral operator  $K_D^{\omega}$  be defined by

$$K_D^{\omega}: E \in L^2(D) \mapsto -\int_D E(y)\Gamma_m(x, y; \omega)dy \in L^2(D).$$

It is well known that, due to the weak singularity of the fundamental solution,  $K_D^{\omega}$  is compact. When  $|\tau \omega^2 \varepsilon_c K_D^{\omega}| < 1$ , then  $I - \tau \omega^2 \varepsilon_c K_D^{\omega}$  is invertible, so (8.1) can be rewritten as

$$E(x) = (I - \tau \omega^2 \varepsilon_c K_D^{\omega})^{-1} [E^{\text{in}}](x) \quad \text{for all } x \in D,$$
(8.2)

where I denotes the identity operator.

Assume that the characteristic size  $\delta$  of the particle D is much smaller than the wavelength  $2\pi/(\omega\sqrt{\varepsilon_m})$ , and let  $\omega \to 0$ . The subwavelength resonance problem is

then to find an  $\omega \in \mathbb{C}$  close to 0 such that  $(I - \tau \omega^2 \varepsilon_c K_D^{\omega})^{-1}$  is singular, see [24]. Such an  $\omega$  would be a subwalength resonance for the high refractive index dielectric particle D.

By expanding in Taylor series the fundamental solution we obtain the following result.

**Lemma 8.1.** Let d = 3. Let  $K_D^{(0)}$  be the Newtonian potential on D, i.e., the operator defined by

$$K_D^{(0)}[E](x) = -\int_D E(y)\Gamma_0(|x-y|) \, dy \quad \text{for } x \in D.$$

The operator  $K_D^{\omega}$  can be rewritten as

$$K_D^{\omega} = \sum_{i=0}^{\infty} \omega^i K_D^{(i)},\tag{8.3}$$

where the series converges in operator norm if  $\omega$  is small enough.

Let  $A_i = \tau \omega^2 \varepsilon_c K_D^{(i)}$ . By expanding in Neumann series, we have

$$\left(I - A_0 - \sum_{i=1}^{\infty} \omega^i A_i\right)^{-1} = \left(I - (I - A_0)^{-1} \sum_{i=1}^{\infty} \omega^i A_i\right)^{-1} (I - A_0)^{-1}$$
$$= \sum_{k=0}^{\infty} \left((I - A_0)^{-1} \sum_{i=1}^{\infty} \omega^i A_i\right)^k (I - A_0)^{-1}$$
$$= (I - A_0)^{-1} + (I - A_0)^{-1} \omega A_1 (I - A_0)^{-1} + O(\omega^3). \quad (8.4)$$

Recall that  $K_D^{(0)} : L^2(D) \to L^2(D)$  is a compact, self-adjoint operator. Let  $\lambda_0$  be an eigenvalue of  $K_D^{(0)}$  associated with the eigenfunction  $\phi_0$ . We remark that the eigenvalues of  $K_D^{(0)}$  are positive. For the analysis of the spectrum of the Newtonian potential, we refer the reader, for instance, to [62].

Let  $\omega_0$  be a frequency at which  $I - A_0$  becomes singular. In particular, let

$$\omega_0 = 1/\sqrt{\tau \varepsilon_c \lambda_0}.\tag{8.5}$$

Note that  $\omega_0$  is small only for  $\tau$  large enough. This shows that subwavelength resonances occur only for particles with high refractive indices.

For  $\omega$  near  $\omega_0$ , we have, by a pole-pencil operator decomposition, that

$$(I - A_0)^{-1}[\psi] = \frac{(\psi, \phi_0)\phi_0}{1 - \tau\omega^2 \varepsilon_c \lambda_0}$$

Therefore considering only the first two terms in the expansion (8.4), we obtain from (8.2) that an approximation of the resonance must satisfy

$$\frac{(E^{\mathrm{in}},\phi_0)\phi_0}{1-\tau\omega^2\varepsilon_c\lambda_0} + \tau\omega^3\varepsilon_c\frac{(K_D^{(1)}[\phi_0],\phi_0)}{(1-\tau\omega^2\varepsilon_c\lambda_0)^2}(E^{\mathrm{in}},\phi_0)\phi_0 = 0,$$

99

where  $(\cdot, \cdot)$  denotes the scalar product on  $L^2(D)$ .

Therefore we have the following approximation for the subwavelength resonance.

**Proposition 8.2.** Let d = 3. Then, the  $O(\omega^4)$ -approximation of the subwavelength resonant frequency  $\omega_{\text{res}}$  of the dielectric particle D satisfies

$$1 - \tau \omega_{\rm res}^2 \varepsilon_c \lambda_0 = -\tau \omega_{\rm res}^3 \varepsilon_c (K_D^{(1)}[\phi_0], \phi_0).$$

Note that, in three dimensions,

$$K_D^{(1)}[\phi] = -i\frac{\sqrt{\varepsilon_m}}{4\pi} \int_D \phi \, dy \quad \text{for all } \phi \in L^2(D).$$

Therefore,  $\omega_{\rm res}$  satisfies

$$1 - \tau \omega_{\rm res}^2 \varepsilon_c \lambda_0 = \frac{i\tau}{4\pi} \omega_{\rm res}^3 \sqrt{\varepsilon_m} \varepsilon_c \Big(\int_D \phi_0 \, dy\Big)^2.$$

Since  $\omega_{\rm res}$  is close to  $\omega_0$ , by approximating  $\omega_{\rm res}^3 \simeq \omega_0^3$ , and since by definition  $\tau \varepsilon_c \lambda_0 = 1/\omega_0^2$ , we obtain

$$1 - \frac{\omega_{\rm res}^2}{\omega_0^2} = \frac{\mathrm{i}\tau}{4\pi} \omega_0^3 \sqrt{\varepsilon_m} \varepsilon_c \Big(\int_D \phi_0 \, dy\Big)^2.$$

**Proposition 8.3.** Let d = 3. Let  $\omega_0$  be defined by (8.5), where  $\lambda_0$  is an eigenvalue of the Newtonian potential  $K_D^{(0)}$ . Assume that  $\omega_0 \ll 1$ . Then, an approximation of the subwavelength resonant frequency of the dielectric particle D can be computed as

$$\omega_{\rm res} = \omega_0 - \frac{\mathrm{i}}{8\pi} \frac{\omega_0^2}{\lambda_0} \sqrt{\varepsilon_m} \Big( \int_D \phi_0 \, dy \Big)^2.$$

By using the Lippman-Schwinger representation formula (8.1), we can also rewrite

$$E(x) - E^{\rm in}(x) \simeq -\omega^2 \tau \varepsilon_c \Gamma_m(x - z; \omega) \frac{(E^{\rm in}, \phi_0)(\int_D \phi_0)}{1 - \tau \omega^2 \varepsilon_c \lambda_0} + \tau \omega^3 \varepsilon_c \frac{(K_D^{(1)}[\phi_0], \phi_0)(E^{\rm in}, \phi_0)(\int_D \phi_0)}{(1 - \tau \omega^2 \varepsilon_c \lambda_0)^2}.$$

By plugging the expression for  $\omega_{res}$  from Proposition 8.2 we obtain the following approximation.

**Proposition 8.4.** For  $\omega$  near the resonant frequency  $\omega_{res}$ , the following monopole approximation of the dielectric nanoparticle D holds:

$$E(x) - E^{\mathrm{in}}(x) \simeq -\frac{\lambda_0 \left(\frac{\omega_{\mathrm{res}}^2}{\omega^2} - 1\right) - \mathrm{i}\frac{\sqrt{\varepsilon_m}}{4\pi} (\int_D \phi_0)^2 \left(\omega - \frac{\omega_{\mathrm{res}}^3}{\omega^2}\right)}{\left(\lambda_0 \left(\frac{\omega_{\mathrm{res}}^2}{\omega^2} - 1\right) - \mathrm{i}\frac{\sqrt{\varepsilon_m}}{4\pi} (\int_D \phi_0)^2 \frac{\omega_{\mathrm{res}}^3}{\omega^2}\right)^2} (E^{\mathrm{in}}, \phi_0)_{\mathrm{L}^2(D)} \Gamma_m(x - z; \omega)$$

$$(8.6)$$

for  $|x-z| \gg 2\pi/(\omega\sqrt{\varepsilon_m})$ .

100

Now, we consider the two-dimensional case. From the asymptotic expansion of the Hankel function  $H_0^{(1)}$  of the first kind of order zero:

$$H_0^{(1)}(s) = \frac{2i}{\pi} \sum_{m=0}^{\infty} (-1)^m \frac{s^{2m}}{2^{2m} (m!)^2} \Big( \log(\gamma s) - \sum_{j=1}^m \frac{1}{j} \Big),$$

where  $2\gamma = \exp(\tilde{\gamma} - i\pi/2)$  with  $\tilde{\gamma}$  being the Euler-Mascheroni constant (see, for instance, [20]), it follows that

$$K_D^{\omega}[E] = -\frac{1}{2\pi} (\log(\omega\gamma)) \int_D E(y) \, dy + K_D^{(0)} + O(\omega^2 \log \omega) \quad \text{as } \omega \to 0, \qquad (8.7)$$

where  $K_D^{(0)}$  is the Newtonian potential in dimension two, that is, the operator defined on  $L^2(D)$  by

$$K_D^{(0)}[E](x) = \int_D E(y)\Gamma(x-y)\,dy \quad \text{for } x \in D,$$

with  $\Gamma$  being the fundamental solution of the Laplacian in  $\mathbb{R}^2$ .

Expanding  $K_D^{\omega}$  as in (8.4) and following the same calculations, we obtain the following characterization of subwavelength resonant frequencies in the two-dimensional case.

**Proposition 8.5.** Let d = 2. Then, the  $o(\omega^4)$ -approximation of the subwavelength resonant frequencies  $\omega_{res}$  of the dielectric particle D satisfies

$$1 - \tau \omega_{\rm res}^2 \varepsilon_c \lambda_0 - \frac{\tau \varepsilon_c}{2\pi} \omega_{\rm res}^2 \log(\gamma \omega_{\rm res}) \Big( \int_D \phi_0 \, dy \Big)^2 = O(\omega^4 \log \omega).$$

# 8.2 Hybridization of subwavelength resonances for a dimer of dielectric nanoparticles

Consider a dimer of two identical particles  $D_1$  and  $D_2$  with the same dielectric parameter as in the above section. Then the field  $E - E^{in}$  scattered by the two particles has the following representation formula:

$$(E - E^{\rm in})(x) = -\omega^2 \tau \varepsilon_c \Big( \int_{D_1} E(y) \Gamma_m(x - y; \omega) dy + \int_{D_2} E(y) \Gamma_m(x - y; \omega) dy \Big).$$
(8.8)

Define the operators  $K_{D_i}^{\omega}$  and  $R_{D_i,D_j}^{\omega}$  for i, j = 1, 2, by

$$K_{D_i}^{\omega}: E|_{D_i} \in \mathcal{L}^2(D_i) \mapsto -\int_{D_i} E(y)\Gamma_m(x-y;\omega)dy\Big|_{D_i} \in \mathcal{L}^2(D_i),$$

and

$$R_{D_i,D_j}^{\omega}: E|_{D_i} \in \mathcal{L}^2(D_i) \mapsto -\int_{D_i} E(y)\Gamma_m(x-y;\omega)dy\Big|_{D_j} \in \mathcal{L}^2(D_j).$$

Then, from (8.8) we obtain the following system of operator equations:

$$\begin{pmatrix} 1 - \tau \omega^2 \varepsilon_c K_{D_1}^{\omega} & -\tau \omega^2 \varepsilon_c R_{D_2, D_1}^{\omega} \\ -\tau \omega^2 \varepsilon_c R_{D_1, D_2}^{\omega} & 1 - \tau \omega^2 \varepsilon_c K_{D_2}^{\omega} \end{pmatrix} \begin{pmatrix} E|_{D_1} \\ E|_{D_2} \end{pmatrix} = \begin{pmatrix} E^{\mathrm{in}}|_{D_1} \\ E^{\mathrm{in}}|_{D_2} \end{pmatrix}$$
(8.9)

The scattering resonance problem is to find  $\omega$  such that the operator in (8.9) is singular. Note that here we have a coupled system of subwavelength resonators. As in [9, 10], the subwavelength resonant frequency  $\omega_{\rm res}$  is hybridized into two subwavelength resonant frequencies  $\omega_{\rm res}^{\pm}$  approximately given by

$$\omega_{\rm res}^{\pm} = \omega_0 \pm \frac{1}{2} \tau \omega_0^3 \varepsilon_c \sqrt{(R_{D_1, D_2}^{\omega_{\rm res}}[\phi_0^{(1)}], \phi_0^{(2)})(R_{D_2, D_1}^{\omega_{\rm res}}[\phi_0^{(2)}], \phi_0^{(1)})}, \tag{8.10}$$

where  $\phi_0^{(i)}$ , for i = 1, 2, is the eigenfunction associated to the eigenvalue  $\lambda_0$  of the Newtonian potential of  $D_i$ . Moreover, in the far-field, the dimer of dielectric particles behaves as the sum of a monopole and a dipole.

In conclusion, in this chapter we provided a mathematical model of resonant high-index nanoparticles. Our results can be used for the analysis, design, and manipulation of resonant dielectric nano-structures and their use as metamaterials. Following [11], we believe that formula (8.6) can be generalized to the time-domain, in order to characterize the temporal response of resonant dielectric nanoparticles and accelerate the computations of the responses of subwavelength dielectric resonators [42, 101].

# Bibliography

- G. Righini, Y. Dumeige, P. Féron, M. Ferrari, G. Nunzi Conti, D. Ristic, and S. Soria. Whispering gallery mode microresonators: Fundamentals and applications. *Rivista Del Nuovo Cimento*, 34:435–488, 2001. doi:10.1393/ ncr/i2011-10067-2.
- M. Abramowitz and I. A. Stegun. Handbook of mathematical functions: with formulas, graphs, and mathematical tables, volume 55. Courier Corporation, 1964.
- [3] H. Ammari, E. Beretta, E. Francini, H. Kang, and M. Lim. Optimization algorithm for reconstructing interface changes of a conductivity inclusion from modal measurements. *Math. Comp.*, 79(271):1757–1777, 2010. doi:10.1090/ S0025-5718-10-02344-6.
- [4] H. Ammari, E. Beretta, E. Francini, H. Kang, and M. Lim. Reconstruction of small interface changes of an inclusion from modal measurements II: the elastic case. J. Math. Pures Appl. (9), 94(3):322–339, 2010. doi:10.1016/j.matpur.2010.02.001.
- [5] H. Ammari, A. Dabrowski, B. Fitzpatrick, and P. Millien. Perturbations of the scattering resonances of an open cavity by small particles. part i: The transverse magnetic polarization case. *preprint*, 2018.
- [6] H. Ammari, A. Dabrowski, B. Fitzpatrick, and P. Millien. Perturbations of the scattering resonances of an open cavity by small particles. part ii: The transverse electric polarization case. *preprint*, 2018.
- [7] H. Ammari, A. Dabrowski, B. Fitzpatrick, P. Millien, and M. Sini. Resonant dielectric nanoparticles with high refractive indices. *preprint*, 2018.

- [8] H. Ammari, Y. Deng, and P. Millien. Surface plasmon resonance of nanoparticles and applications in imaging. Arch. Ration. Mech. Anal., 220(1):109– 153, 2016. URL: https://doi.org/10.1007/s00205-015-0928-0, doi:10. 1007/s00205-015-0928-0.
- [9] H. Ammari, B. Fitzpatrick, H. Kang, M. Ruiz, S. Yu, and H. Zhang. Mathematical and computational methods in photonics and phononics, volume 235 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2018.
- [10] H. Ammari, B. Fitzpatrick, H. Lee, S. Yu, and H. Zhang. Double-negative acoustic metamaterials. ArXiv e-prints, Sept. 2017. arXiv:1709.08177.
- [11] H. Ammari, P. Garapon, L. Guadarrama Bustos, and H. Kang. Transient anomaly imaging by the acoustic radiation force. J. Differential Equations, 249(7):1579-1595, 2010. URL: https://doi.org/10.1016/j.jde.2010.07.
  012, doi:10.1016/j.jde.2010.07.012.
- [12] H. Ammari and H. Kang. Polarization and moment tensors, volume 162 of Applied Mathematical Sciences. Springer, New York, 2007.
- [13] H. Ammari, H. Kang, and H. Lee. Asymptotic expansions for eigenvalues of the Lamé system in the presence of small inclusions. *Comm. Partial Differential Equations*, 32(10-12):1715–1736, 2007. doi:10.1080/03605300600910266.
- [14] H. Ammari, H. Kang, and H. Lee. Layer potential techniques in spectral analysis, volume 153 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2009. URL: http://dx.doi.org/10.1090/ surv/153, doi:10.1090/surv/153.
- [15] H. Ammari, H. Kang, M. Lim, and H. Zribi. Layer potential techniques in spectral analysis. Part I: Complete asymptotic expansions for eigenvalues of the Laplacian in domains with small inclusions. *Trans. Amer. Math. Soc.*, 362(6):2901–2922, 2010. URL: http://dx.doi.org/10.1090/ S0002-9947-10-04695-7, doi:10.1090/S0002-9947-10-04695-7.
- [16] H. Ammari and A. Khelifi. Electromagnetic scattering by small dielectric inhomogeneities. J. Math. Pures Appl. (9), 82(7):749–842, 2003. URL: https://

doi.org/10.1016/S0021-7824(03)00033-3, doi:10.1016/S0021-7824(03) 00033-3.

- [17] H. Ammari and P. Millien. Shape and size dependence of dipolar plasmonic resonance of nanoparticles. to appear in J. Math. Pures Appl. arXiv:1804.
   11092.
- [18] H. Ammari, P. Millien, M. Ruiz, and H. Zhang. Mathematical analysis of plasmonic nanoparticles: the scalar case. Arch. Ration. Mech. Anal., 224(2):597–658, 2017. URL: https://doi.org/10.1007/s00205-017-1084-5, doi:10.1007/s00205-017-1084-5.
- [19] H. Ammari and S. Moskow. Asymptotic expansions for eigenvalues in the presence of small inhomogeneities. *Math. Methods Appl. Sci.*, 26(1):67–75, 2003. doi:10.1002/mma.343.
- [20] H. Ammari and J.-C. Nédélec. Full low-frequency asymptotics for the reduced wave equation. *Appl. Math. Lett.*, 12(1):127–131, 1999. URL: https://doi.org/10.1016/S0893-9659(98)00137-2, doi:10.1016/S0893-9659(98)00137-2.
- [21] H. Ammari, M. Ruiz, S. Yu, and H. Zhang. Mathematical analysis of plasmonic resonances for nanoparticles: the full Maxwell equations. J. Differential Equations, 261(6):3615-3669, 2016. URL: https://doi.org/10.1016/j. jde.2016.05.036, doi:10.1016/j.jde.2016.05.036.
- H. Ammari and F. Triki. Splitting of resonant and scattering frequencies under shape deformation. J. Differential Equations, 202(2):231-255, 2004.
   URL: http://dx.doi.org/10.1016/j.jde.2004.02.017, doi:10.1016/j.jde.2004.02.017.
- [23] H. Ammari and D. Volkov. Asymptotic formulas for perturbations in the eigenfrequencies of the full Maxwell equations due to the presence of imperfections of small diameter. Asymptot. Anal., 30(3-4):331–350, 2002.
- [24] H. Ammari and H. Zhang. Super-resolution in high-contrast media. Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 471(2178), 2015. URL: http:

//rspa.royalsocietypublishing.org/content/471/2178/20140946, arXiv:http://rspa.royalsocietypublishing.org/content/471/2178/ 20140946.full.pdf, doi:10.1098/rspa.2014.0946.

- [25] D. H. Armitage and S. J. Gardiner. Classical potential theory. Springer Monographs in Mathematics. Springer-Verlag London, Ltd., London, 2001. URL: http://dx.doi.org/10.1007/978-1-4471-0233-5, doi:10. 1007/978-1-4471-0233-5.
- [26] S. Arnold, M. Khoshsima, I. Teraoka, S. Holler, and F. Vollmer. Shift of whispering-gallery modes in microspheres by protein adsorption. Opt. Lett., 28(4):272-274, Feb 2003. URL: http://ol.osa.org/abstract.cfm?URI= ol-28-4-272, doi:10.1364/0L.28.000272.
- [27] M. Aspelmeyer, T. J. Kippenberg, and F. Marquardt. Cavity optomechanics. *Rev. Mod. Phys.*, 86:1391-1452, Dec 2014. URL: https://link.aps.org/ doi/10.1103/RevModPhys.86.1391, doi:10.1103/RevModPhys.86.1391.
- [28] C. Bandle and A. Wagner. Second domain variation for problems with Robin boundary conditions. J. Optim. Theory Appl., 167(2):430-463, 2015. doi: 10.1007/s10957-015-0801-1.
- [29] M. Berljafa and S. Güttel. A rational krylov toolbox for matlab, university of manchester, uk. MIMS EPrint 2014.56, 2014. URL: http://eprints.maths. manchester.ac.uk/id/eprint/2209.
- [30] G. Besson. Comportement asymptotique des valeurs propres du laplacien dans un domaine avec un trou. Bull. Soc. Math. France, 113(2):211-230, 1985. URL: http://www.numdam.org/item?id=BSMF\_1985\_\_113\_\_211\_0.
- [31] H. Bethe and J. Schwinger. Perturbation Theory for Cavities. Massachusetts Institute of Technology, Radiation Laboratory, Cambridge, 1943.
- [32] V. I. Burenkov and P. D. Lamberti. Spectral stability of Dirichlet second order uniformly elliptic operators. J. Differential Equations, 244(7):1712–1740, 2008. doi:10.1016/j.jde.2007.12.009.
- [33] V. I. Burenkov, P. D. Lamberti, and M. Lanza de Cristoforis. Spectral stability of nonnegative selfadjoint operators. *Sovrem. Mat. Fundam. Napravl.*, 15:76–

111, 2006. English translation in J. Math. Sci., 149(4):1417–1452, 2008. doi:
10.1007/s10958-008-0074-4.

- [34] H. Cao and J. Wiersig. Dielectric microcavities: model systems for wave chaos and non-Hermitian physics. *Rev. Modern Phys.*, 87(1):61-111, 2015. URL: https://doi.org/10.1103/RevModPhys.87.61, doi:10.1103/RevModPhys. 87.61.
- [35] D. L. Colton and R. Kress. Integral equation methods in scattering theory. Pure and Applied Mathematics (New York). John Wiley & Sons, Inc., New York, 1983.
- [36] M. Costabel, E. Darrigrand, and H. Sakly. The essential spectrum of the volume integral operator in electromagnetic scattering by a homogeneous body. C. R. Math. Acad. Sci. Paris, 350(3-4):193-197, 2012. URL: https://doi.org/10.1016/j.crma.2012.01.017, doi:10.1016/j.crma.2012.01.017.
- [37] R. Courant and D. Hilbert. Methods of mathematical physics. Vol. I. Interscience Publishers, Inc., New York, N.Y., 1953.
- [38] A. Dabrowski. Explicit terms in the small volume expansion of the shift of Neumann Laplacian eigenvalues due to a grounded inclusion in two dimensions. J. Math. Anal. Appl., 456(2):731-744, 2017. URL: https://doi.org/ 10.1016/j.jmaa.2017.07.027.
- [39] A. Dabrowski. A localized boundary deformation which splits the spectrum of the laplacian. *ArXiv e-prints*, June 2017. arXiv:1706.03555.
- [40] A. Dabrowski. On the behaviour of repeated eigenvalues of singularly perturbed elliptic operators. *Submitted*, 2018.
- [41] L. C. Evans. Partial differential equations, volume 19 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, second edition, 2010. doi:10.1090/gsm/019.
- [42] R. Faggiani, A. Losquin, J. Yang, E. Marsell, A. Mikkelsen, and P. Lalanne. Modal analysis of the ultrafast dynamics of optical nanoresonators. ACS Photonics, 4(4):897–904, 2017. URL: https://doi.org/10.1021/

acsphotonics.6b00992, arXiv:https://doi.org/10.1021/acsphotonics. 6b00992, doi:10.1021/acsphotonics.6b00992.

- [43] E. Feleqi. Estimates for the deviation of solutions and eigenfunctions of second-order elliptic Dirichlet boundary value problems under domain perturbation. J. Differential Equations, 260(4):3448-3476, 2016. doi:10.1016/ j.jde.2015.10.038.
- [44] B. Fitzpatrick and A. Dabrowski. Fast reconstruction via asymptotic formulae of small domain perturbations from laplacian eigenvalues shift. *preprint*, 2018.
- [45] M. R. Foreman, J. D. Swaim, and F. Vollmer. Whispering gallery mode sensors. Adv. Opt. Photon., 7(2):168-240, Jun 2015. URL: http://aop.osa. org/abstract.cfm?URI=aop-7-2-168, doi:10.1364/AOP.7.000168.
- [46] M. J. Friedman and J. E. Pasciak. Spectral properties for the magnetization integral operator. *Math. Comp.*, 43(168):447–453, 1984. URL: https://doi. org/10.2307/2008286, doi:10.2307/2008286.
- [47] P. R. Garabedian and M. Schiffer. Convexity of domain functionals. J. Analyse Math., 2:281–368, 1953. doi:10.1007/BF02825640.
- [48] D. Gilbarg and N. S. Trudinger. Elliptic partial differential equations of second order. Classics in Mathematics. Springer-Verlag, Berlin, 2001. Reprint of the 1998 edition.
- [49] I. C. Gohberg and E. I. Sigal. An operator generalization of the logarithmic residue theorem and Rouché's theorem. Mat. Sb. (N.S.), 84(126):607–629, 1971.
- [50] J. Gopalakrishnan, S. Moskow, and F. Santosa. Asymptotic and numerical techniques for resonances of thin photonic structures. SIAM J. Appl. Math., 69(1):37-63, 2008. URL: https://doi.org/10.1137/070701388, doi:10.1137/070701388.
- [51] P. Grinfeld. Hadamard's formula inside and out. J. Optim. Theory Appl., 146(3):654-690, 2010. URL: http://dx.doi.org/10.1007/ s10957-010-9681-6, doi:10.1007/s10957-010-9681-6.
- [52] J. K. Hale. Eigenvalues and perturbed domains. In Ten mathematical essays on approximation in analysis and topology, pages 95–123. Elsevier B. V., Amsterdam, 2005. doi:10.1016/B978-044451861-3/50003-3.
- [53] S. Haroche and J.-M. Raimond. Exploring the quantum. Oxford Graduate Texts. Oxford University Press, Oxford, 2006. Atoms, cavities and photons. URL: https://doi.org/10.1093/acprof:oso/9780198509141.001.0001, doi:10.1093/acprof:oso/9780198509141.001.0001.
- [54] P. Heider. Computation of scattering resonances for dielectric resonators. Comput. Math. Appl., 60(6):1620-1632, 2010. URL: https://doi.org/10. 1016/j.camwa.2010.06.044, doi:10.1016/j.camwa.2010.06.044.
- [55] P. Heider, D. Berebichez, R. V. Kohn, and M. I. Weinstein. Optimization of scattering resonances. *Struct. Multidiscip. Optim.*, 36(5):443-456, 2008. URL: https://doi.org/10.1007/s00158-007-0201-8, doi:10.1007/s00158-007-0201-8.
- [56] E. Heine. Handbuch der Kugelfunctionen. Theorie und Anwendungen. Band II. Zweite umgearbeitete und vermehrte Auflage. Thesaurus Mathematicae, No. 1. Reprint, Physica-Verlag, Würzburg, 1961, (1878 original).
- [57] A. Henrot. Extremum problems for eigenvalues of elliptic operators. Frontiers in Mathematics. Birkhäuser Verlag, Basel, 2006.
- [58] D. Henry. Perturbation of the boundary in boundary-value problems of partial differential equations, volume 318 of London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge, 2005. doi:10.1017/ CB09780511546730.
- [59] E. Hille. Analytic function theory. Vol. II. Introductions to Higher Mathematics. Ginn and Co., Boston, Mass.-New York-Toronto, Ont., 1962.
- [60] L. Hörmander. Linear partial differential operators. Springer Verlag, Berlin-New York, 1976.
- [61] S. Jahani and Z. Jacob. All-dielectric metamaterials. *Nature Nanotechnology*, 11, 2016. doi:10.1038/nnano.2015.304.

- [62] T. S. Kalmenov and D. Suragan. A boundary condition and spectral problems for the Newton potential. In *Modern aspects of the theory of partial differential equations*, volume 216 of *Oper. Theory Adv. Appl.*, pages 187– 210. Birkhäuser/Springer Basel AG, Basel, 2011. URL: https://doi.org/ 10.1007/978-3-0348-0069-3\_11, doi:10.1007/978-3-0348-0069-3\_11.
- [63] C.-Y. Kao and F. Santosa. Maximization of the quality factor of an optical resonator. Wave Motion, 45(4):412-427, 2008. URL: https://doi.org/10.1016/j.wavemoti.2007.07.012, doi:10.1016/j.wavemoti.2007.07.012.
- [64] S. Kim and J. E. Pasciak. The computation of resonances in open systems using a perfectly matched layer. *Math. Comp.*, 78(267):1375–1398, 2009. URL: https://doi.org/10.1090/S0025-5718-09-02227-3, doi:10.1090/S0025-5718-09-02227-3.
- [65] J. C. Knight, N. Dubreuil, V. Sandoghdar, J. Hare, V. Lefèvre-Seguin, J. M. Raimond, and S. Haroche. Mapping whispering-gallery modes in microspheres with a near-field probe. *Opt. Lett.*, 20(14):1515–1517, Jul 1995. URL: http://ol.osa.org/abstract.cfm?URI=ol-20-14-1515, doi:10.1364/OL. 20.001515.
- [66] J. C. Knight, N. Dubreuil, V. Sandoghdar, J. Hare, V. Lefèvre-Seguin, J. M. Raimond, and S. Haroche. Characterizing whispering-gallery modes in microspheres by direct observation of the optical standing-wave pattern in the near field. *Opt. Lett.*, 21(10):698–700, May 1996. URL: http://ol.osa.org/abstract.cfm?URI=ol-21-10-698, doi:10.1364/OL.21.000698.
- [67] A. I. Kuznetsov, A. E. Miroshnichenko, M. L. Brongersma, Y. S. Kivshar, and B. Luk'yanchuk. Optically resonant dielectric nanostructures. *Science*, 354(6314), 2016. URL: http://science.sciencemag.org/content/354/ 6314/aag2472, arXiv:http://science.sciencemag.org/content/354/ 6314/aag2472.full.pdf, doi:10.1126/science.aag2472.
- [68] C. C. Lam, P. T. Leung, and K. Young. Explicit asymptotic formulas for the positions, widths, and strengths of resonances in mie scattering. J. Opt. Soc. Am. B, 9(9):1585-1592, Sep 1992. URL: http://josab.osa.org/abstract. cfm?URI=josab-9-9-1585, doi:10.1364/JOSAB.9.001585.

- [69] A. Lemenant, E. Milakis, and L. V. Spinolo. Spectral stability estimates for the Dirichlet and Neumann Laplacian in rough domains. J. Funct. Anal., 264(9):2097–2135, 2013. doi:10.1016/j.jfa.2013.02.006.
- [70] J. Lin and F. Santosa. Resonances of a finite one-dimensional photonic crystal with a defect. SIAM J. Appl. Math., 73(2):1002–1019, 2013. URL: https://doi.org/10.1137/120897304, doi:10.1137/120897304.
- [71] J. Lin and F. Santosa. Scattering resonances for a two-dimensional potential well with a thick barrier. SIAM J. Math. Anal., 47(2):1458-1488, 2015. URL: https://doi.org/10.1137/140952053, doi:10.1137/140952053.
- [72] D. Ludwig. Geometrical theory for surface waves. SIAM Rev., 17:1-15, 1975.
  URL: https://doi.org/10.1137/1017001, doi:10.1137/1017001.
- [73] P. A. Martin. Multiple scattering, volume 107 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 2006. Interaction of time-harmonic waves with N obstacles. URL: https://doi.org/10. 1017/CB09780511735110, doi:10.1017/CB09780511735110.
- [74] P. A. Martin. Multiple scattering, volume 107 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 2006. Interaction of time-harmonic waves with N obstacles. URL: https://doi.org/10. 1017/CB09780511735110, doi:10.1017/CB09780511735110.
- [75] B. Matkowsky. A boundary layer approach to the whispering gallery phenomenon. Quart. Appl. Math., 2018. doi:10.1090/qam/1513.
- [76] V. Maz'ya and S. Nazarov. Singularities of solutions of the Neumann problem at a conic point. Sibirsk. Mat. Zh., 30(3):52-63, 218, 1989. English translation in Siberian Mathematical Journal, 30(3):387-396, 1989. doi: 10.1007/BF00971492.
- [77] V. Maz'ya, S. Nazarov, and B. Plamenevskij. Asymptotic theory of elliptic boundary value problems in singularly perturbed domains. Vol. II, volume 112 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 2000. Translated from the German by Plamenevskij. doi:10.1007/ 978-3-0348-8432-7.

- [78] T. Meklachi, S. Moskow, and J. C. Schotland. Asymptotic analysis of resonances of small volume high contrast linear and nonlinear scatterers. J. Math. Phys., 59(8):083502, 20, 2018. URL: https://doi.org/10.1063/1.5031032, doi:10.1063/1.5031032.
- [79] A. M. Micheletti. Perturbazione dello spettro dell'operatore di Laplace, in relazione ad una variazione del campo. Ann. Scuola Norm. Sup. Pisa, 26(3):151–169, 1972.
- [80] A. M. Micheletti. Perturbazione dello spettro di un operatore ellittico di tipo variazionale, in relazione ad una variazione del campo. Ann. Mat. Pura Appl., 97(4):267–281, 1973. doi:10.1007/BF02414915.
- [81] A. M. Micheletti. Perturbazione dello spettro di un operatore ellittico di tipo variazionale, in relazione ad una variazione del campo. II. Ricerche Mat., 25(2):187–200, 1976.
- [82] B. Min, E. Ostby, V. Sorger, E. Ulin-Avila, L. Yang, X. Zhang, and K. Vahala. High-q surface-plasmon-polariton whispering-gallery microcavity. *Nature*, 457, 2009. doi:10.1038/nature07627.
- [83] B.-T. Nguyen and D. S. Grebenkov. Localization of Laplacian eigenfunctions in circular, spherical, and elliptical domains. SIAM J. Appl. Math., 73(2):780-803, 2013. URL: https://doi.org/10.1137/120869857, doi: 10.1137/120869857.
- [84] A. N. Oraevsky. Whispering-gallery waves. Quantum Electronics, 32(5):377, 2002. URL: http://stacks.iop.org/1063-7818/32/i=5/a=R01.
- [85] B. Osting and M. I. Weinstein. Long-lived scattering resonances and Bragg structures. SIAM J. Appl. Math., 73(2):827–852, 2013. URL: https://doi. org/10.1137/110856228, doi:10.1137/110856228.
- [86] S. Ozawa. Singular variation of domains and eigenvalues of the Laplacian. Duke Math. J., 48(4):767-778, 1981. URL: http://dx.doi.org/10.1215/ S0012-7094-81-04842-0, doi:10.1215/S0012-7094-81-04842-0.

- [87] S. Ozawa. Spectra of domains with small spherical Neumann boundary. Proc. Japan Acad. Ser. A Math. Sci., 58(5):190-192, 1982. URL: http: //projecteuclid.org/euclid.pja/1195516025.
- [88] A. Pick, B. Zhen, O. D. Miller, C. W. Hsu, F. Hernandez, A. W. Rodriguez, M. Soljačić, and S. G. Johnson. General theory of spontaneous emission near exceptional points. *Opt. Express*, 25(11):12325-12348, May 2017. URL: http: //www.opticsexpress.org/abstract.cfm?URI=oe-25-11-12325, doi:10. 1364/OE.25.012325.
- [89] J. Rauch and M. Taylor. Potential and scattering theory on wildly perturbed domains. J. Funct. Anal., 18:27–59, 1975.
- [90] F. Rellich. Perturbation theory of eigenvalue problems. Gordon and Breach Science Publishers, New York-London-Paris, 1969.
- [91] G. F. Roach. *Green's functions*. Cambridge University Press, Cambridge-New York, second edition, 1982.
- [92] F. Ruesink, H. M. Doeleman, R. Hendrikx, A. F. Koenderink, and E. Verhagen. Perturbing open cavities: Anomalous resonance frequency shifts in a hybrid cavity-nanoantenna system. *Phys. Rev. Lett.*, 115:203904, Nov 2015. URL: https://link.aps.org/doi/10.1103/PhysRevLett.115. 203904, doi:10.1103/PhysRevLett.115.203904.
- [93] G. Savaré and G. Schimperna. Domain perturbations and estimates for the solutions of second order elliptic equations. J. Math. Pures Appl. (9), 81(11):1071–1112, 2002. doi:10.1016/S0021-7824(02)01256-4.
- [94] M. A. Schmidt, D. Y. Lei, L. Wondraczek, V. Nazabal, and S. A. Maier. Hybrid nanoparticle-microcavity-based plasmonic nanosensors with improved detection resolution and extended remote-sensing ability. *Nature Communications*, 3:1108 EP-, Oct 2012. Article. URL: http://dx.doi.org/10.1038/ ncomms2109.
- [95] M. Teytel. How rare are multiple eigenvalues? Comm. Pure Appl. Math., 52(8):917-934, 1999. doi:10.1002/(SICI)1097-0312(199908)52:8<917:: AID-CPA1>3.3.CO;2-J.

- [96] K. Uhlenbeck. Generic properties of eigenfunctions. Amer. J. Math., 98(4):1059–1078, 1976. doi:10.2307/2374041.
- [97] G. Verchota. Layer potentials and regularity for the Dirichlet problem for Laplace's equation in Lipschitz domains. J. Funct. Anal., 59(3):572-611, 1984.
   URL: http://dx.doi.org/10.1016/0022-1236(84)90066-1, doi:10.1016/ 0022-1236(84)90066-1.
- [98] F. Vollmer and S. Arnold. Whispering-gallery-mode biosensing: label-free detection down to single molecules. *Nature Methods*, 5, 2008. doi:10.1038/ nmeth.1221.
- [99] F. Vollmer, S. Arnold, and D. Keng. Single virus detection from the reactive shift of a whispering-gallery mode. *Proceedings of the National Academy of Sciences*, 105(52):20701-20704, 2008. URL: http://www. pnas.org/content/105/52/20701, arXiv:http://www.pnas.org/content/ 105/52/20701.full.pdf, doi:10.1073/pnas.0808988106.
- [100] M. Wei, G. Majda, and W. Strauss. Numerical computation of the scattering frequencies for acoustic wave equations. *Journal of Computational Physics*, 75(2):345-358, 1988. URL: http://www.sciencedirect.com/science/ article/pii/0021999188901179, doi:10.1016/0021-9991(88)90117-9.
- [101] W. Yan, R. Faggiani, and P. Lalanne. Rigorous modal analysis of plasmonic nanoresonators. *Phys. Rev. B*, 97:205422, May 2018. URL: https://link.aps.org/doi/10.1103/PhysRevB.97.205422, doi:10.1103/ PhysRevB.97.205422.
- [102] Z.-J. Yang, R. Jiang, X. Zhuo, Y.-M. Xie, J. Wang, and H.-Q. Lin. Dielectric nanoresonators for light manipulation. *Physics Reports*, 701:1– 50, 2017. Dielectric nanoresonators for light manipulation. URL: http:// www.sciencedirect.com/science/article/pii/S037015731730203X, doi: 10.1016/j.physrep.2017.07.006.
- [103] Q. Zhao, J. Zhou, F. Zhang, and D. Lippens. Mie resonance-based dielectric metamaterials. *Materials Today*, 12(12):60-69, 2009. URL: http:// www.sciencedirect.com/science/article/pii/S1369702109703189, doi: 10.1016/S1369-7021(09)70318-9.